Designing for STEM E-Mentoring for Young Women: the Saudi Context



Aseel Alhadlaq

School of Computing Science Newcastle University

This dissertation is submitted for the degree of

Doctor of Philosophy

June 2020

Abstract

The underrepresentation of women in the STEM workforce is a global issue. In Saudi Arabia, women constitute 48% of undergraduates in STEM. This is in large part a result of the unique cultural context, gender segregation in educational institutions and the perceived prestige associated with studying a STEM subject for both women and men. However, these high levels of educational participation have not translated into a significant increase in the number of women in the STEM workforce. The exposing of girls to female role models and mentors in STEM, and raising awareness of their achievements and successes, is an approach that has been developed by previous research to address perceptions that girls are less likely to succeed in a STEM field than boys. In Saudi Arabia, the lack of female role models and the absence of mentoring and support programmes are likely some of the factors that have contributed to the low levels of progression of female graduates into STEM professions.

In this research, I explore the current state of e-mentoring, seek to understand the requirements in designing STEM e-mentoring platforms for teenagers and young women in the Saudi context and understand how mentors and mentees interact within an e-mentoring relationship. I conducted four studies that aimed to understand, design and evaluate e-mentoring in the Saudi context. First, I deployed a structured STEM e-mentoring program, with the aims of understanding the applicability of e-mentoring in Saudi Arabia. The findings from this study led to carrying out two co-design studies to explore the perceived barriers to, and opportunities for, alternative and more flexible forms of e-mentoring. The findings from the three studies have led to the design of *Qudwa* based on four design requirements: 1) flexibility and control; 2) visibility; 3) integration of e-mentoring with daily internet activity; and 4) a trusted connection between mentors and mentees. Finally, I discuss the deployment and evaluation of *Qudwa*.

This research makes three contributions, first it is the first systematic investigation of e-mentoring in Saudi Arabia. While I aimed to understand the role of cultural norms, the studies show that the findings are affected less by cultural norms and more by the traits of the participants' age group. The second novel contribution of this research is the application of co-design methods to help address the gap in understanding the needs and opportunities in designing e-mentoring for young people. The results point to a need for a shift in the design of e-mentoring models for younger generations into less committed and more flexible relationships. Finally, it presents the design and evaluation of *Qudwa*, which facilitates e-mentoring through existing social media technologies and practices (un-platforming).

III

To my kids Layan, Dana and Abdulaziz

Acknowledgments

I would like to express my gratitude to my supervisor Dr. Ahmed Kharrufa for his patience, support, and guidance throughout the process, his guidance helped me in all the time of research and writing of this thesis. I wish to express my deepest gratitude to Professor Patrick Olivier for his support, motivation, and knowledge throughout my PhD journey. I would like to show my appreciation to Dr. Abeer Al Nuaim from King Saud University for her continuous support.

I would like to thank King Saud University for supporting and funding my study. The research would not have been possible without the staff and students of Open Lab, Newcastle University and the participants who volunteered their time. Special thanks and gratitude to Dr. Clement Lee for his help in creating the visualisations for participants' interactions in the first study, Dalya Al-Shahrabi for her technical help and support on *Qudwa* and Euijin Hwang for his drawings on *Qudwa*'s user journey.

I wish to acknowledge the support and great love of my parents, my brothers and sisters. Special thank you to my kids Layan, Dana and Abdaulaziz. They kept me going on and this work would not have been possible without their love, understanding and support.

Publication

Alhadlaq, A., Kharrufa, A., Olivier, P., 2019. Exploring e-mentoring: co-designing & unplatforming. Behaviour & Information Technology 38, 1122–1142. https://doi.org/10.1080/0144929X.2019.1571110

Table of Contents

Abstract		III
List of Fig	ures	XIII
List of Tab	bles	XVI
Chapter 1	Introduction	
1.1	Overview	
1.2	Motivation	
1.2.1	Global initiatives to advocate STEM for women	
1.3	Research Context	
1.3.1	E-mentoring	
1.3.2	The Saudi context	9
1.4	Research Questions	
1.5	Summary of Contributions	
1.6	Thesis Structure	
Chapter 2	Literature Review	
2.1	Introduction	
2.2	E-mentoring Models and Frameworks	
2.3	Geographical Distribution of E-Mentoring	
2.4	E-mentoring Relationships and Contexts	
2.5	Means of Communications Used in E-Mentoring	
2.6	STEM E-mentoring for Female Students	
2.7	Mentoring in Saudi Arabia	
2.8	Summary	
Chapter 3	Methodology	
3.1	Introduction	
3.2	Related Work: Co-design and Teenagers	
3.3	Methods	
3.3.1	Recruitment	
3.3.2	Ş, ,	
3.3.3	C	
3.3.4	Thematic analysis	

3.4	Summary	
Chapter 4	Pilot Study: Structured E-mentoring	42
4.1	Introduction	
4.2	Method	
4.2.1	Platform selection	43
4.2.2	Recruitment	
4.2.3	Matching process	
4.2.4	Study design	49
4.2.5	Data collection	
4.3	Results and Analysis	
4.3.1	Pre-study survey	
4.3.2	Level of engagement	51
4.3.3	Analysis of posts	55
4.3.4	Interviews with mentors and mentees	
4.3.5	Interviews with parents	63
4.3.6	Focus group	67
4.3.7	Post-study survey	70
4.4	Discussion	71
4.4.1	Language, culture and mentoring	71
4.4.2	Relationship trajectory	72
4.4.3	Overestimated effect of cultural context	74
4.4.4	Measurement of impact	75
4.5	Summary	
Chapter 5	Co-Design Workshops	78
5.1	Introduction	
5.2	Method: Study 2 – Co-design with Undergraduate Female Students	
5.2.1	Participants	79
5.2.2	Workshop 1: Problem statement stories	
5.2.3	Data from Workshop 1	
5.2.4	Online survey	
5.2.5	Workshop 2: Bodystorming	
5.2.6	Data from Workshop 2	
5.2.7	Workshop 3: Paper prototyping	
5.2.8	Data from Workshop 3	93
5.3	Method: Study 3 – Co-design with High School Female Students	
5.3.1	Participants	
5.3.2	Workshop: Design the box	
5.3.3	Data from the workshop	101

5.3.4	Online survey	
5.4	Findings and Analysis of Study 2 and Study 3	
5.4.1	Online survey	
5.4.2	Thematic analysis of discussions and presentations	
5.5	Discussion	
5.5.1	Cultural dimensions of the target age group	
5.5.2	Co-design e-mentoring	
5.5.3	Un-platforming E-mentoring	
5.6	Summary	116
Chapter 6	The Design of <i>Qudwa</i>	117
6.1	Introduction	117
6.2	Requirements for Design	
6.2.1	RQ1 – Flexibility and control	
6.2.2	RQ2 – Visibility	
6.2.3	RQ3 – Integration of e-mentoring with daily internet activity	
6.2.4	RQ4 – Trusted connection	
6.3	Qudwa	
6.3.1	What is Qudwa	
6.3.2	Meeting the requirements	
6.3.3	Design of Qudwa	
6.4	Summary	
Chapter 7	Deployment and Evaluation of <i>Qudwa</i>	
7.1	Introduction	
7.2	Study Design	
7.2.1	Recruitment	
7.2.2	Announcements	
7.2.3	Wizard of Oz	
7.2.4	Data collection	
7.3	Findings and Analysis	
7.3.1	Mentors	
7.3.2	Submissions	
7.3.3	Mentees	
7.3.4	Introduction through social media	
7.3.5	Join WhatsApp group	
7.3.6		
7.3.7	Survey	
7.3.8	Interviews	

7.3	.9 Social media announcements	171
7.4	Discussion	
7.4	.1 Design goals	
7.4	.2 Improvement of Qudwa	176
7.5	Summary	
Chapter	8 Discussion and Conclusion	
8.1	Introduction and Overview	
8.2	The Saudi Context	
8.2	.1 The demand for mentoring	179
8.2	.2 Lack of Arabic content on the internet	
8.2	.3 Technology and culture	
8.2	.4 Why Saudi female mentors	
8.3	The E-mentoring Design	
8.3		
8.3	3.2 Supporting social media technologies	
8.3		
8.3	C C	
8.4	Design Implications	
8.5	Limitations and Future Work	
8.6	Conclusion	
Referenc	ces	197
Reference		
Appendi	x A Pilot Study	
A.1	Mentors Profiles	
A.2	Pre-Study Survey for the Mentees	
A.3	Post-Study Survey for the Mentees	
<i>A.4</i>	Semi-structured Interview Questions for the Mentors	
A.5	Semi-structured Interview Questions for the Mentees	
A.6	Semi-structured Interview Questions for the Parents	
Appendi	x B Co-design Workshops	224
B.1	Survey	
Appendi	x C Qudwa	
<i>C.1</i>	System Architecture	

<i>C.2</i>	Post-Study Survey	226
<i>C.3</i>	Semi-Structured Interview Questions for the mentors and the mentees	230

List of Figures

Figure 2-1 Structured e-mentoring by Single and Muller (2001)15	5
Figure 4-1 Edmodo home page allows creating accounts for teachers, students and parents44	4
Figure 4-2 Student registration form shows field for Group Code40	5
Figure 4-3 Parents can add their children by submitting a Parent Code obtained from their	
children's Edmodo accounts	5
Figure 4-4 Group A had the largest number of mentees, but their activity level was low or	
passive	3
Figure 4-5 Group C shows a high-activity mentor who wrote more than one new post in a	
single day, even though some of these posts did not receive any response from the mentees in	
the group54	4
Figure 4-6 Group F shows no interaction between participants. Mentee ALFL initiated the first	t
post in her group, but mentor F _L disregarded her post55	5
Figure 5-1 Screenshots from the video presenting the main character, Sara	1
Figure 5-2 Screenshots expressing Sara's goal and challenges	1
Figure 5-3 Screenshots displaying the setting of the story (Sara's love of space and her family	r
background)	1
Figure 5-4 Printed icons that were given participants to help them with their posters	2
Figure 5-5 Poster by Group#182	3
Figure 5-6 Poster by Group#2	4
Figure 5-7 Poster by Group#3	4
Figure 5-8 Poster by Group#485	5
Figure 5-9 Poster by Group#580	5
Figure 5-10 Development of scenes in bodystorming workshop	7
Figure 5-11 Ideas discussed through the bodystorming workshop	8
Figure 5-12 Discussion points around scheduling with a mentor	9
Figure 5-13 Discussion around being connected to one mentor vs. more than one mentor90)
Figure 5-14 Live session suggestions	0
Figure 5-15 Mobile apps vs. web platforms	1
Figure 5-16 Encouraging professionals into mentoring92	2
Figure 5-17 Discussion points on sharing with others92	3
Figure 5-18 Annotated sketch of Group#194	4
Figure 5-19 Annotated sketch of Group#290	5
Figure 5-20 Annotated sketches of Group#3	8

Figure 5-21 Participants in session designing their products	100
Figure 5-22 Groups' representatives 'selling' their product	101
Figure 5-23 world.com design	102
Figure 5-24 Mentor Me design	103
Figure 5-25 Know Your Major design	103
Figure 5-26 Answers to the question: "Do you sometimes feel the need to talk to someone	
regarding your study or future career?"	105
Figure 5-27 Answers to the question: "Do you have a female role model in your life?"	106
Figure 5-28 Answers to the question: "Do you have a clear vision and knowledge about ca	reer
options available in your field of study?"	106
Figure 6-1 Home page of Qudwa	122
Figure 6-2 User journey of Qudwa presenting each step and its design requirements	123
Figure 6-3 Bottom of the home page	126
Figure 6-4 Profile page of a mentor	126
Figure 6-5 Dropdown menu shows how to connect with a mentor	127
Figure 6-6 Request form for the WhatsApp group connection method	127
Figure 6-7 Request form for the social network connection method	128
Figure 6-8 Request form for the online meeting connection method	128
Figure 6-9 Let us help form	129
Figure 6-10 Recommend a mentor form	129
Figure 6-11 Qudwa editor page	130
Figure 6-12 Creating a mentor by filling in information in English and Arabic	131
Figure 6-13 Submissions page	132
Figure 6-14 A request submitted by one of the mentees	132
Figure 6-15 Analytics page showing an overview of activity on Qudwa	133
Figure 6-16 Visitor activity shows all activities of users on Qudwa (top) and Visitor paths	
showing individual paths of users and whether they are a returning user or not (bottom)	134
Figure 7-1 The first scene (top left) presents the target audience, which is high school and	
university female students. The second (middle left) and third (bottom left) scenes present	the
problem faced by the students by asking questions: "Do you need guidance from an expert	t?"
"Do you feel confused between study fields and work fields?" The fourth scene (top right))
presents the voluntary nature of Qudwa. The fifth scene (middle right) explains how to use	•
Qudwa by choosing a mentor and how you want to connect with her. The final scene (bott	om
right) displays how to reach Qudwa.	137
Figure 7-2 Profiles of <i>Qudwa</i> on Instagram (left) and Twitter (right)	138

Figure 7-3 Introducing two mentees in a WhatsApp group	140
Figure 7-4 Examples of mentors with engineering, science or technology tags	143
Figure 7-5 Profiles of the three mentors who tagged themselves with technology and	
engineering	143
Figure 7-6 Twitter analytics	172
Figure 7-7 Number of views of the promoted tweet	172
Figure 7-8 Insights from Instagram on the number of visits and views	173
Figure 7-9 Insights about the audience on Instagram	173

List of Tables

Table 2-1 E-mentoring phases proposed by Risquez (2008)	17
Table 3-1 Studies conducted in this research	35
Table 4-1 Parents' demographics	51
Table 4-2 Numbers of posts and likes of members in all groups	53
Table 5-1 Summary of the two studies showing the numbers of participants, ages of	
participants and purpose of each study.	78
Table 5-2 Demographics of parents from Study 2 and Study 3	104
Table 6-1 Summary of <i>Qudwa</i> 's design requirements	118
Table 7-1 Tasks that are completed by the admin or the system	139
Table 7-2 Total number of submissions and number of each type of request	144
Table 7-3 Number of submissions received for each mentor	144
Table 7-4 Number and type of submissions sent by the same mentee	145
Table 7-5 Age group of mentees	147
Table 7-6 Why the mentees were requesting to contact the mentors	149
Table 7-7 Number of mentees in each WhatsApp group	151

Chapter 1 Introduction

1.1 Overview

This research aimed to study science, technology, engineering and mathematics (STEM) ementoring in Saudi Arabia from a human-computer interaction (HCI) perspective. The aims of the STEM e-mentoring were to expose young Saudi women and teenagers to female role models and mentors in STEM, and raise their awareness about achievements, successes and career opportunities. Saudi Arabia has a unique cultural norm that is different from other contexts. While e-mentoring is applied widely in the Western context, it is not in Saudi Arabia or similar contexts. Therefore, this research aimed to apply exploratory research to understand and study how e-mentoring can be applied in this context appropriately within the culture and how participants would interact within an e-mentoring relationship. Exploratory research tends to have a qualitative nature (Hanington, 2010; Stebbins, 2001). In this research, qualitative methods were applied to gain a full understanding of e-mentoring in the area of the study. In HCI, qualitative methods are essential methods for understanding the area and users, gathering requirements and evaluating outcomes (Adams et al., 2008; Blandford et al., 2016). In this research, several qualitative methods were used to elicit data and requirements including interviews, pre-study and post-study surveys, focus groups, co-design workshops and observation of interactions.

Four studies were conducted to fulfil this aim. The first step was to apply a STEM ementoring program for Saudi female teenagers to explore its potential in the context. The initial exploration reported that the cultural norms had little effect on the requirements and needs, and suggested that age group traits are what dictate the design requirements. Therefore, two co-design studies were conducted with the targeted population to gain a deeper understanding of the design requirements and needs. Based on these requirements, *Qudwa* was designed. *Qudwa* is based on the idea of un-platforming, which facilitates existing social media networks to develop less committed and more flexible e-mentoring relationships.

1.2 Motivation

It has been widely observed that the participation rate of girls in the STEM fields is lower than that of boys. In some countries, such as the USA, the proportion of women in the STEM workforce is significantly lower (i.e., less than 25%) than that of men, despite sustained national, regional and local policy initiatives (Beede et al., 2011; Khare et al., 2013). On the other hand, participation in STEM subjects in further and higher education in a number of

countries in the Middle East and North Africa region has approached gender equity, as over the past five decades there has been a significant and steady growth in the proportion of women and girls studying STEM subjects at school and in university (Ramirez and Kwak, 2015). For example, in Saudi Arabia 48% of undergraduates in STEM subjects are female (Ministry of Education, n.d.). This high rate is in large part a result of the unique cultural context, gender segregation in educational institutions and the perceived prestige associated with studying a STEM subject for both women and men. However, these high levels of educational participation have not translated into a significant increase in the number of women in the STEM workforce. The statistics of Saudis employed in 2015 showed that the number of newly employed Saudi women in STEM sectors was lower in comparison to other sectors. For example, only 3% of Saudis who were employed in the energy sectors were female and in the science and technology sector 17% were female, while Saudi women represented 55.7% of newly employed Saudis in the social services sector and 74.6% in the education sector (General Authority for Statistics, 2015). In 2016, women in Saudi Arabia accounted for 33% of the workforce, 39.5% of them in clerical jobs, 3.3% in the engineering workforce, 1.4% in the industrial and chemical sectors, and 4% in the information and communication technology (ICT) workforce (General Authority for Statistics, 2016).

In addition, Saudi women and girls continue to be discouraged by prevailing social attitudes and a climate of cultural self-restriction (Abalkhail and Allan, 2015; Al-Asfour et al., 2017). These factors are likely to have contributed to the low levels of progression for female STEM graduates into STEM professions (Abalkhail and Allan, 2015; Al-Asfour et al., 2017). The lack of female role models and the absence of mentoring and networking programs were identified as some of the main barriers for the professional advancement of women in Saudi Arabia in different fields such as medicine (Alwazzan and Rees, 2016), leadership in higher education (Alsubaie and Jones, 2017) and entrepreneurship (Danish and Smith, 2012). Fundamental issues arise around Saudi women in STEM. The lack of female role models and the lack of mentoring and networking support systems have been some of the reasons for the low participation of women in STEM jobs even though almost half of the Saudi undergraduates in STEM fields are female.

1.2.1 Global initiatives to advocate STEM for women

Recently, there have been many regional and global initiatives to encourage girls into STEM studies and careers; for example, initiatives instigated by the IEEE Women in Computing

Society¹, Arab Women in Computing² and Society of Women Engineers³. Moreover, a number of awards have been established to recognise women in STEM, including the Women in IT Awards⁴, and L'Oréal-UNESCO Women in Science Awards⁵. Outreach-targeted promotional programs have also been developed to promote girls' interest in STEM careers and shed light on the value of STEM professions. For example, NASA has developed SISTER, a one-week summer outreach program that aims to inspire young girls by exploring non-traditional career fields with women engineers, mathematicians and scientists (Garner, 2016). Many attempts have been made to address the misconception that girls are less likely to succeed in STEM fields than boys by exposing girls to female role models in STEM and raising awareness of their achievements, successes and everyday professional activities. These so-called *mentors* perform a considerably larger role than acting as simple role models: they encourage, advise, counsel and share their knowledge with the girls involved (Stoeger et al., 2013). Ensher et al. (2003) described a mentor's responsibility towards their mentees as falling into three categories: vocational support, psychological support and role modelling. Organisations such as Million Women Mentors⁶ promote traditional face-to-face mentoring for women and young girls to increase their interest in STEM and help them achieve their goals. However, the organisation of such face-to-face mentoring programs faces a number of significant challenges, including the availability of mentors, mentor training and geographical distance. These barriers can be reduced through the use of computer-mediated communications in e-mentoring (Bierema and Merriam, 2002; Khare et al., 2013). Accessibility, efficiency, flexibility and personalisation have all been identified as advantages of e-mentoring over traditional face-to-face mentoring (Khare et al., 2013).

My personal experience of being a Saudi woman in a STEM field has driven this research. During my undergraduate studies, I faced frustration and confusion in understanding different fields and career options. I needed someone to guide and support me, a female who had been in the same circumstances. I could not reach out to any professional Saudi women at that time and age. I studied computer science not because I have knowledge about it or was interested in it, but because my school grades were excellent. The cultural norm (for males and females) was to study medicine, computer science or engineering if you had high grades.

¹ https://www.computer.org/communities/women-in-computing

² http://arabwic.org/

³ http://societyofwomenengineers.swe.org/

⁴ http://womeninitawards.com/

⁵ https://www.forwomeninscience.com/en/awards

⁶ https://www.millionwomenmentors.com

Engineering colleges were not available for females at that time and medicine was not appealing to me, so I choose to study computer science. After graduation, my knowledge about career options was also limited. It was common for Saudi women to work in either the health sector or the education sector. I applied for a job as a teaching assistant at the university and still work there. I have two daughters, who will be applying for university in a couple of years, thinking about different fields and career options, and I wish for them a better experience and wider options than mine.

The specific objective of this research was to examine e-mentoring in the Saudi context. This examination has been conducted through four studies that applied e-mentoring, explored the needs and requirements, then designed, deployed and evaluated a new STEM e-mentoring system to encourage teenagers and young women in Saudi Arabia to pursue STEM academic degrees, advise them and raise their awareness about STEM career opportunities.

1.3 Research Context

1.3.1 E-mentoring

Definitions and functions

A mentor is defined by Zey (1984, p. 7) as a person "who oversees the career and development of another person, usually a junior, through teaching, counselling, providing psychological support, protecting, and at times promoting or sponsoring." Given the long history of mentoring, there is a lack of a unified definition and specified functions of mentoring (Crisp and Cruz, 2009; Haggard et al., 2011; Jacobi, 1991). Jacobi (1991) presented 15 different definitions of 'mentoring' and 'mentor' from the literature in the fields of education, management and psychology. Haggard et al., (2011) found over 40 definitions of a mentor. Crisp and Cruz (2009), through their review of the literature, identified over 50 definitions of mentoring depending on the field or the discipline. For example, Crisp and Cruz (2009) reported that Roberts (2000) defined mentoring as "a formalized process whereby a more knowledgeable and experienced person actuates a supportive role of overseeing and encouraging reflection and learning within a less experienced and knowledgeable person, so as to facilitate that person's career and personal development", while Blackwell (1989) said that mentoring "is a process by which persons of a superior rank, special achievements, and prestige instruct, counsel, guide and facilitate the intellectual and/or career development of persons identified as protégés."

In comparison to the wide range of English definitions of 'mentoring' and 'mentor', the Arabic language lacks words for mentoring, mentor and mentee. Since "language

expresses cultural reality", as noted by Kramsch and Widdowson (1998), the lack of the word 'mentoring' in the Arabic language emphasises the fact that mentoring is not a recognised concept in Saudi Arabia and the Arab region.

Despite the fact that there is inconsistency in the definitions of mentoring, Jacobi (1991) presented the aspects on which researchers have agreed about mentoring. First, while the forms of the mentoring relationship may vary, they all serve the purpose of helping and assisting the mentee. Second, the mentoring relationship helps in providing emotional and psychological support, career and professional development, and role modelling. Third, the benefits of mentoring are mutual for both the mentor and mentee. Fourth, mentoring relationships are personal and require direct contact between a mentor and a mentee. Finally, mentors have more expertise, power and accomplishment in comparison to their mentees.

The advancement of technology and computer-mediated communications (CMC) have provided enhancement and development of the mentoring process (Bierema and Merriam, 2002). E-mentoring is based on the application of technology for communication between mentors and mentees. It has been found that the earliest form of e-mentoring was developed in 1994 (Mason et al., 1994). The aim of the project was to mentor school students by university students to improve their writing skills via "telecommunications technology", which was not described. Interestingly the authors explained that they would use the term 'mentor' instead of 'tutor' as the term 'tutor' was commonly used at that time. It seems that back in the early 1990s mentoring was concerned with emotional and psychological support, and started expanding to career development and skill enhancement; as the authors said, "we believe that a mentor not only provides emotional and psychological support but also provides direction in career and professional development" (Mason et al., 1994, p. 124).

Other terms have been used to describe the use of CMC in mentoring such as telementoring, cybermentoring, virtual mentoring and online mentoring (Rowland, 2012; Single and Muller, 2001). The term 'e-mentoring' is better because it indicates the use of electronic communication such as email (Single and Single, 2005). E-mentoring has been defined as "a computer mediated, mutually beneficial relationship between a mentor and a protégé which provides learning, advising, encouraging, promoting, and modeling, that is often boundaryless, egalitarian, and qualitatively different than traditional face-to-face mentoring" (Bierema and Merriam, 2002, p. 214).

The terms 'mentee' and 'protégé' have been used in the literature to describe the person who is being mentored. Some works have used both terms interchangeably; for example, Crisp and Cruz (2009) in their review of the literature used both terms without

distinguishing them. Other examples of using both terms are found as well (Dawson, 2014;

DiRenzo et al., 2013; Risquez, 2008; Rowland, 2012).

In the dictionaries, a mentee is defined as:

a person who is advised and helped by a more experienced person over a period of time (Oxford Learner's Dictionaries, n.d.)

a person who is helped by a mentor (Cambridge Dictionary, n.d.)

Protégé has a French origin that means 'protected' (Oxford Learner's Dictionaries,

n.d.). It has been defined as:

a young person who is helped in their career and personal development by a more experienced person (Oxford Learner's Dictionaries, n.d.)

a young person who is helped and taught by an older and usually famous person (Cambridge Dictionary, n.d.)

Klasen and Clutterbuck (2012) favoured the term 'mentee' over 'protégé'. They explained that 'protégé' means "protected one", which indicates inequality of power in the mentoring relationship, while power should be excluded from the mentoring relationship. This also implies that the 'protégé' is naïve and has nothing to add to the relationship, which is not true; even though the mentee is the one seeking help or knowledge, the mentor learns and benefits from the mentee as well (Klasen and Clutterbuck, 2012). The term 'mentee' will be used in this work to describe the person who is being mentored.

It has been found that e-mentoring provides the same benefits and support to the mentees that are provided in other forms of mentoring (Single and Single, 2005). Bierema and Merriam (2002) proposed that a mentor's role or function is to provide the mentee with help and assistance needed in any aspect that ensures their development and to be a role model for them to learn from and observe. Risquez (2008) mentioned that a mentor facilitates the professional or personal development of a mentee during a transitional time in their lives. Ensher et al. (2003) defined three functions of a mentor: vocational or instrumental support, psychological support and role modelling. Vocational or instrumental support is related to the professional development of the mentee, psychological support is related more to counselling, encouragement and building self-esteem, and role modelling is related to presenting an example for the mentee to follow and be influenced by. Single and Single (2005) stated three roles for a mentor, two of them matching those of Ensher et al. (2003): psychosocial support and instrumental support. The third role presented by Single and Single (2005) is informational support, which means the process of sharing information and knowledge requested by the mentee on a particular matter or subject.

While e-mentoring is described as a "mutual beneficial relationship" (Bierema and Merriam, 2002), the benefits for mentors and mentees differ, and most studies have focused on mentees' benefits in a mentoring relationship (Ehrich et al., 2004). A review of the literature shows that networking, reflection, professional development and personal satisfaction are among the most cited benefits for the mentors (Ehrich et al., 2004).

Advantages of e-mentoring

Space and time limitations were the two most commonly faced obstacles to maintaining the traditional face-to-face mentoring relationship; these obstacles are overcome by e-mentoring (Bierema and Merriam, 2002; Rhodes et al., 2006; Risquez, 2008; Single and Muller, 2001). Using CMC is time efficient. With online communications, travelling to and from the physical mentoring location is not required, therefore it saves travel time (Ensher et al., 2003; Single and Single, 2005). Since e-mentoring has lower time investment, it provides opportunities for mentors and mentees who find the required time for traditional mentoring an obstacle to their participation in a mentoring relationship (Single and Muller, 2001). If asynchronous means of communications are used in e-mentoring, this allows participants to engage at their convenience and reduces the pressure to respond immediately (Bierema and Merriam, 2002; Ensher et al., 2003; Single and Muller, 2001). In addition, online communication facilitates the exchange of more knowledge and information in a shorter period of time (Bierema and Merriam, 2002).

The elimination of the geographical constraints in e-mentoring gives it another benefit over traditional mentoring. Because e-mentoring is not restricted by geography, mentees are not limited to the mentors around them, which provides them with greater access (Ensher et al., 2003). With the reduction of time and space constraints, the pool of mentors available increases in e-mentoring (Ensher et al., 2003; Rhodes et al., 2006). The connection between mentors and mentees across different locations and organisations has two advantages: it expands their professional networks and it promotes impartiality (Single and Single, 2005). In traditional mentoring, the dyads in the mentoring relationship are usually from the same organisation or network. Being from the same organisation makes the mentee susceptible to judgement and may affect their professional development due to the mentor's opinion of them, or the mentee can be judged for participating in a helping program (Single and Single, 2005). On the other hand, e-mentoring provides inter-organisational connections which expand professional networks, especially for those who find it hard to have an informal mentoring relationship in their own organisation (Single and Single, 2005). In addition, e-

mentoring supports impartiality because the mentor, who is from a different organisation, has no hidden motivation for being in this relationship (Risquez, 2008; Single and Single, 2005).

The nature of online communication methods also has advantages that support ementoring. One main advantage is that e-mentoring conceals physical characteristics and social cues (Bierema and Merriam, 2002; Ensher et al., 2003; Single and Muller, 2001). People usually build an initial impression of another person based on their physical appearance, but with online communication in e-mentoring, participating individuals have the opportunity to focus less on outer characteristics and more on personalities, thoughts and needs (Ensher et al., 2003; Rhodes et al., 2006). Women and minority groups have fewer chances of engaging in a mentoring relationship (Bierema and Merriam, 2002). The reduced focus on demographics creates an opportunity for women, minorities and others who are excluded from traditional mentoring relationships (Bierema and Merriam, 2002; Ensher et al., 2003; Single and Muller, 2001; Single and Single, 2005). It also provides opportunities for personalities who are shy or have reservations about reaching out for mentoring from their community (Rhodes et al., 2006). In traditional mentoring, the person with higher status (the mentor) has the power and control in the relationship; on the other hand, the levelling of status in e-mentoring offers a safe context for the relationship and eliminates the power dynamics (Bierema and Merriam, 2002; Ensher et al., 2003; Single and Single, 2005). Equalisation of status provides a comfortable environment for mentees who perceive themselves as lower than their mentor (Ensher et al., 2003; Single and Single, 2005). Honesty and self-disclosure also tend to be of higher levels in e-mentoring due to the nature of the online communication methods (Rhodes et al., 2006).

The use of CMC liberates e-mentoring from time and geographical constraints, which results in saved time and cost, and expands professional networks. However, e-mentoring is not cost-free, due to the need for websites, administration, matching protocols and evaluation techniques (Single and Single, 2005). While CMCs have the advantage of concealing social cues and status, which creates a safe environment and opportunities for more diverse mentees to be in an e-mentoring relationship, it may create an easier context for mentees to ignore or terminate the e-mentoring relationship (Single and Single, 2005) or an environment where harmful messages can be easily exchanged (Rhodes et al., 2006). Miscommunication or misinterpretation can be a challenge faced in CMC as well (Ensher et al., 2003; Rhodes et al., 2006).

1.3.2 The Saudi context

ICT in Saudi Arabia

In the late 1990s the internet was launched to the public in Saudi Arabia (Alshahrani, 2016). The delay compared to other countries came from the conservative nature of Saudi society. The government had concerns and reservations about introducing internet service to the society (Alshahrani, 2016). In 1997, an official decree was issued to give the King Abdulaziz City for Science and Technology (KACST), which is an organisation that reports to the Prime Minister of Saudi Arabia, the responsibility to provide and control the internet in Saudi Arabia (Alshahrani, 2016). KACST filters and supervises the internet gateways and blocks undesired websites to ensure that the content delivered is culturally acceptable. Despite the delay in the launching of the internet, Saudi Arabia is considered the fastest and largest growing Arab country in terms of the use of ICT and internet user population (Alshahrani, 2016; Bafakih et al., 2016; Stanger et al., 2017). According to Stanger et al. (2017), Saudi Arabia has been among the top ten highest performing countries on the International Telecommunication Union's ICT development index since 2010. Statistics show that Saudi Arabia ranked second (after Iran) in the number of internet users in the Middle East, with 18 million users (Bafakih et al., 2016). The number of internet users jumped from one million users in 2003 to 16.5 million users in 2013, which represents more than half (55.1%) of the Saudi population (Alshahrani, 2016).

Saudis are avid consumers of global media and widening access to the internet in Saudi Arabia (Bafakih et al., 2016; Winder, 2014) has led to rapid and intense adoption of social media by young Saudis in particular (Bafakih et al., 2016). Globally, Saudi Arabia has ranked number one in the highest annual growth rate of social media users and as the country with the highest level of Snapchat adoption (Radcliffe and Bruni, 2019). In 2017, the Arab Social Media Report showed that Saudi Arabia ranked number one in the number of Twitter users and the number of shared tweets in the Arab region (Salem, 2015). In 2013, Saudi Arabia was the country with the highest Twitter usage (Winder, 2014). With increasing interest in Twitter in 2013, Facebook lost its popularity and dramatically dropped in usage from 61.17% in 2012 to 35.68% in 2013 (Reyaee and Ahmed, 2015). The highest number of views of YouTube in the world was of Saudis, with 90 million videos per day (Winder, 2014).

Despite the interest in and rapid growth of the internet and social media penetration in Saudi Arabia, there is little research on understanding the reasons which facilitated this adoption in Saudi Arabia (Akram and Albalawi, 2016) and the Arab Gulf region (Reyaee and Ahmed, 2015). It is considered that the high level of social media use by young Saudis is due

to the space that these platforms have created for them to connect with the world around them and to express their opinions to the public without being monitored (Akram and Albalawi, 2016; Winder, 2014). The high usage of social media among Saudi youth has made them more involved and engaged with their society (Kuppuswamy and Rekha, 2015). Winder (2014) in his exploration of the reasons behind the adoption of Twitter among young Saudis concluded that young Saudis use Twitter as a medium for expressing their anger in relation to financial and social issues. Young Saudi females had their share in expressing their frustration on Twitter about the ban on women drivers. In Saudi Arabia interaction between males and females is very limited, and Twitter created a space for different Saudi users to interconnect with each other, especially males and females (Winder, 2014). The adoption of social media was tested among young Saudis and the results showed that their social media adoption was motivated by the perceived connectedness and enjoyment (Akram and Albalawi, 2016) and perceived social capital (Al-Ghaith, 2015).

Saudi females, culture and ICT

Religion and culture are two key factors that determine the norms of Saudi society (Al Lily, 2011; Al-Saggaf, 2011). It has been discussed in the previous section that Saudis are high consumers of the internet, but this does not change the fact that Saudi society is still highly influenced by religion and culture (Al-Saggaf, 2016). The main aspects that characterise Saudi Arabia as a conservative country are family honour and gender segregation (Al Alhareth, 2013; Al Lily, 2011; Al-Saggaf, 2011; Al-Saggaf, 2016; Guta and Karolak, 2015; Yamin and Aljehani, 2016); these factors have also affected Saudi women's participation in the online context.

The collectivist nature of Saudi Arabia has encouraged the family honour concept, where a female does not only represent herself but also represents her whole family and, in some cases, her extended family as well (Al Lily, 2011; Al-Saggaf, 2016; Guta and Karolak, 2015). This cultural norm has made Saudi women cautious regarding revealing their private information online. The loss of private information such as name, workplace, school, address, phone number and personal photos can create serious harm to a woman herself, her family and their reputation (Al-Saggaf, 2011; Al-Saggaf, 2016; Binsahl and Chang, 2012; Guta and Karolak, 2015). Studies of the attitudes of Saudi women using social networking sites (SNS) (Al-Saggaf, 2011; Al-Saggaf, 2016; Binsahl and Chang, 2012) showed that Saudi women do not share or post their personal photos even with privacy settings set to the highest level on these sites. Even though they might have only real-life friends on their SNS, they fear that their friends might show their personal photos to others (Binsahl and Chang, 2012). Sharing

personal photos online might put them into a blackmail situation and dishonour of their family reputation (Al-Saggaf, 2011; Al-Saggaf, 2016). The fear and caution about losing private information has raised Saudi women's awareness about privacy and security in online contexts (Al-Saggaf, 2016; Binsahl and Chang, 2012; Guta and Karolak, 2015). Studies have shown that they develop a sense of self-censorship (Guta and Karolak, 2015) and awareness about privacy features and how to protect themselves in an online context (Al-Saggaf, 2011).

Saudi society tends to protect females in all aspects, hence it is not customary for a Saudi woman to leave her family for education or work (Al Alhareth, 2013; Yamin and Aljehani, 2016). When it comes to marriage or education, families favour marriage over education for their daughters, which may result in them dropping out of school or not continuing their undergraduate studies, especially in rural areas of Saudi Arabia (Al Alhareth, 2013; Yamin and Aljehani, 2016). It is believed that marriage protects family honour (Al Alhareth, 2013). Limited mobility and early marriage create obstacles to study and work for Saudi females who are located in rural areas. The freedom of movement which is provided by the internet has created an environment where these women can work and study from their homes (Al Lily, 2011).

Another factor that distinguishes Saudi society from others is the segregation between men and women, publicly and privately (Al Alhareth, 2013; Al Lily, 2011). Social interaction between men and women is inappropriate, with the exception of professional interactions (Al-Saggaf, 2016). In the real world, friendship between males and females is not culturally accepted, and the same applies in online spaces (Al-Saggaf, 2016; Guta and Karolak, 2015). Although male–female online interactions are not acceptable, the use of online identities and concealing of private information have created a space for this kind of interaction and networking to occur and to cross the gender lines created by the culture (Al Lily, 2011; Guta and Karolak, 2015). In other words, the two genders are physically separated and virtually connected (Al Lily, 2011).

Gender segregation has weakened Saudi women's voices (Al Lily, 2011; Guta and Karolak, 2015). The culture requires women to be modest, shy and reserved, so women do not share their voices or opinions in public, especially on politics and social issues (Al-Saggaf, 2011). The online space has constructed a place for Saudi women to share their opinions and views in all aspects (Al Lily, 2011; Al-Saggaf, 2011; Guta and Karolak, 2015). In fact, Saudi women's online participation is of an equal share to their counterparts (Al Lily, 2011). Saudi women use SNS to express themselves and their political and social views (Al-Jabri et al., 2015; Al-Saggaf, 2011) and as a medium to influence and create change in society (Guta and Karolak, 2015).

These cultural factors have contributed to decreases in visibility and connectedness of Saudi female professionals in society, while the development of the internet and social media networks has created a medium for gradually increasing their visibility and voice.

1.4 Research Questions

This research aimed to explore the opportunities for STEM e-mentoring in Saudi Arabia to encourage teenagers and young women into STEM fields and raise their awareness about their career opportunities and about existing Saudi female STEM professionals, which would lead to a higher level of Saudi women's participation in the STEM workforce. The research questions of the research are the following:

- **RQ1.** What is the potential of applying a STEM e-mentoring program for teenagers and young women in the Saudi context?
- **RQ2.** What are the factors that contribute to designing a STEM e-mentoring system for teenagers and young women in the Saudi context?
- **RQ3.** How to design a STEM e-mentoring system for teenagers and young women in the Saudi context?

1.5 Summary of Contributions

This research contributes to this growing area of e-mentoring research in three aspects:

- 1. This is the first study to systematically investigate e-mentoring in Saudi Arabia. The lack of previous examples applied in the same context pointed to the need to understand the cultural norms that would affect the application of e-mentoring in Saudi Arabia. This research proves the applicability and success of e-mentoring in Saudi Arabia for teenagers and young women, and suggests the application of e-mentoring in other fields and settings in Saudi Arabia and the Arab region.
- 2. This research presents new insights into how co-design methods are beneficial in designing e-mentoring for young people. A novel contribution of this research is the application of co-design methods to address the gap in understanding the needs and opportunities in designing e-mentoring for young people. The application of co-design methods has contributed to a new form of e-mentoring based on the needs and requirements of the targeted mentees. Other forms of e-mentoring could emerge from applying co-design methods with different purposes and different populations.
- This research describes the design, deployment and evaluation of *Qudwa*. *Qudwa* is an ementoring design that is based on utilising existing technologies to accomplish less committed and more flexible e-mentoring relationships between mentors and mentees.

This contribution shows the potential that existing technologies and social media platform have to facilitate a new form of e-mentoring.

1.6 Thesis Structure

The research presented in this thesis is structured as follows, with a brief overview of each chapter.

Chapter 2 introduces e-mentoring models and frameworks, the different e-mentoring contexts and relationships, and the means of communications that have been used, followed by examples of e-mentoring programs that were applied to advocate for STEM for females. It also gives an overview of the Saudi context, addressing the existing literature around mentoring in Saudi Arabia, the cultural factors that identify such a context and the ICT adoption by Saudis. The final section discusses co-design methods with the targeted population. The chapter concludes with the research gaps identified: the gap in the literature about Saudi Arabia regarding the non-existence of e-mentoring programs and the lack of applying co-design methods for designing e-mentoring programs.

Chapter 3 presents the methodology of this research and describes the design methods in each study.

Chapter 4 presents the first study of this research (Study 1). This pilot study was a systematic investigation of e-mentoring in Saudi Arabia. An e-mentoring program was deployed aiming to encourage high school girls in Saudi Arabia to pursue STEM higher education studies and raise their awareness of STEM career opportunities. The design, analysis and findings of the study are documented.

Chapter 5 presents the second and third studies, which are both co-design workshops. Study 2 consisted of three co-design workshops over a period of three weeks with first-year female undergraduates to understand the participants' expectations of e-mentoring. Study 3 was a one-session workshop with schoolgirls (aged 17–18). The chapter discusses the findings of both studies (Studies 2 and 3).

Chapter 6 introduces the design requirements and system design of *Qudwa*, an unplatformed STEM e-mentoring system designed to facilitate communication between Saudi female professional mentors and young female mentees. The design requirements are derived from the previously conducted studies (Studies 1, 2 and 3).

Chapter 7 presents the fourth and final study of this research. The deployment of *Qudwa*, data collection, evaluation and discussion are presented in this chapter.

Chapter 8 discusses the findings of the work presented in this thesis and highlights potential future research.

Chapter 2 Literature Review

2.1 Introduction

The literature review in this chapter addresses the concept and applications of e-mentoring in general and in Saudi Arabia specifically. The chapter begins with a description of existing e-mentoring models and frameworks, the different e-mentoring contexts and relationships, and means of communications that have been used, followed by examples of e-mentoring programs that were applied to advocate STEM for females. The following section addresses the existing literature on mentoring in Saudi Arabia. The final section discusses co-design methods with young people. It concludes with discussion of the research gaps identified from the review of the literature.

2.2 E-mentoring Models and Frameworks

Review of the literature identified a few scholarly articles that presented e-mentoring models. Single and Muller (2001) presented the model of structured e-mentoring. First, they defined structured e-mentoring as follows (Single and Muller, 2001, p. 108):

E-mentoring that occurs within a formalized program environment, which provides training and coaching to increase the likelihood of engagement in the e-mentoring process, and relies on program evaluation to identify improvements for future programs and to determine the impact on the participants.

Their model consists of three phases: 1) planning; 2) structured implementation; and 3) assessment (Figure 2-1). In the planning phase, a holistic idea of the e-mentoring program is developed, from identifying the participating population and matching process to setting program goals. Recruitment informs mentors and mentees about the e-mentoring opportunity. Through the e-mentoring website, mentors and mentees can fill in the required applications. It is important to manage the expectations that mentors and mentees agree on, the frequency and method of contact, and the expected outcomes from their relationship. Single and Muller described three ways to match e-mentoring pairs. The first is *participant choice*, which entitles mentees to choose their own mentors based on their displayed biographies. The second is *uni-directional*, where the program administrator matches a mentee with a mentor based on the preferences of that mentee. The final matching method is called *bi-directional*, which collects the preferences of mentors and mentees to create pairs based on their preferences.



Figure 2-1 Structured e-mentoring by Single and Muller (2001)

The second phase is structured implementation, which is affected by program goals and is intended to aid the participants to make the most of their participation. It includes training, coaching and community building. At the beginning of the program, training is delivered to mentors. Training can be delivered electronically, focusing on relevant issues around mentoring and the targeted population. Coaching, on the other hand, is delivered during the program. Coaching is delivered through short messages sent to the mentor containing discussion topics and mentoring tips. Coaching helps to support the continuity of the relationship because it prompts topics for conversation. It also helps in creating a line of connection between the program administrator and the participants. Successful e-mentoring relationships occur when the program promotes a sense of community to its participants. To promote community building, the program creates a space where mentors and mentees can communicate with participants other than their one-to-one pair.

The final phase focuses on evaluation of the program. Single and Muller's (2001) assessment involves collecting three types of data for evaluation: involvement data; formative data; and summative data. According to Single and Muller (2001), the frequency of exchange is an indicator of success in e-mentoring programs. Involvement data investigates the interaction between mentors and mentees by examining their relationship, frequency and engagement. Formative data relates to the process of mentoring and help with the modification and improvement of the program – it is the data that helps to improve future iterations of the program. The final type of data is summative data, which addresses the effectiveness and outcomes of the program by evaluating the achievement of the program's goals and the value.

Bierema and Merriam (2002, p. 214) in their work provided a definition of ementoring:

> e-mentoring is a computer mediated, mutually beneficial relationship between a mentor and a protégé which provides learning, advising, encouraging, promoting, and modeling, that is often boundaryless, egalitarian, and qualitatively different than traditional face-to-face mentoring.

They proposed a conceptual framework in order for e-mentoring relationships to be successful. The first step is to have the right tool (i.e. a proper computer with internet access) and to have knowledge of how to use it. The second step is to identify and match the mentors and the mentees. After that, the relationships are formed by managing the expectations of both parties and the frequency of exchanges. They emphasised that regular communication between mentors and mentees is a key factor in the success of e-mentoring. The expected regularity of contact is set to twice a week in order to form a relationship. The next step is to foster familiarity between them by sharing personal experiences, lives and interests, providing support and facilitation to monitor, support and make sure that communications are going on between mentors and mentees. The final step is to provide a positive end or closure to the relationship.

O'Neill (2004) in his work used e-mentoring as a facilitator in a knowledge-building community. He presented the "mentoring in the open" model, which is different from previous models in the sense that mentees are grouped together with a mentor. Therefore, there is no private one-to-one communication between mentors and mentees. O'Neill (2004) reported that this "mentoring in the open" model had three advantages: 1) mentees learned from their peers' mentoring relationship experiences; 2) mentors enabled learning by scaffolding social comparisons between peers; and 3) mentors could be helpful in promoting peer support.

In her review of the literature on best practices, Risquez (2008) concluded that in conducting an e-mentoring program, the program should go through three phases: 1) design and planning; 2) implementation; and 3) evaluation. She stated that for e-mentoring systems to function effectively, it is necessary to apply the three phases properly. Table 2-1 shows a summary of each phase and its associated factors.

Phase	Factors
Design and planning	Statement of purpose and long-range plan Relevant populations and stakeholders Contextualisation Technology strategy Promotion and marketing policy Safety measures
Implementation	Recruitment plan Eligibility screening Induction and orientation Coaching Matching and re-matching Monitoring Support, recognition and retention Closure steps
Evaluation	Types of data collected Moment Dissemination

Table 2-1 E-mentoring phases proposed by Risquez (2008)

Another review of the literature by Salimi et al. (2017) concluded with a framework for an effective e-mentoring relationship. They applied a qualitative meta-synthesis method to extract the major factors that contribute to building a framework for effective e-mentoring. They extracted four major factors: 1) communication medium; 2) quantity of communication; 3) quality of communication; and 4) communication outcome. The first factor is the type of communication method, its accessibility and its familiarity to the mentor and the mentee. The second factor is related to the frequency of contact, the duration of each session and the duration of the whole relationship. Quality of communication describes the mutual trust, commitment and interest in developing the relationship. It also describes the harmonisation between the mentor and the mentee. The final factor represents the content that has been exchanged between the mentor and mentee, and the goals that have been accomplished through the e-mentoring relationship.

Through this literature review, it has been found that the studies which discussed ementoring applications had learning or social theoretical backgrounds. E-mentoring was used as a support mechanism to apply those theories. For example, Yalof and Chametzky (2016) and Rashid and Rahman (2014) used e-mentoring to explore the concept of the community of practice in a learning context, Li et al. (2010) used e-mentoring to support the inquiry-based learning concept, while Savoy's (2013) work on e-mentoring was based on social cognitive career theory. Lamb and Aldous (2014), on the other hand, presented an e-mentoring study based upon the theoretical concept of the pedagogical device by Bernstein (2003). This review of the literature shows that only a handful of articles presented an ementoring model or framework. It also shows the lack of development of models based upon existing e-mentoring models. Lenear (2007) and Williams et al. (2012) in their work identified a lack of developed e-mentoring models. Both studies indicated a rise in literature around the application of e-mentoring programs in comparison to the modest literature around e-mentoring models and frameworks. Lenear (2007) identified some limitations in the ementoring literature: the lack of theoretical e-mentoring models and the limited number of research on youth mentoring. While she overlooked Bierema and Merriam's (2002) model, she criticised the broadness of Single and Muller's (2001) model and that their model can be applied to any educational setting, not only mentoring. Similarly, Williams et al. (2012) indicated the need for developed e-mentoring models that cover all relevant factors and dimensions. Williams et al. (2012) did not discuss the models presented above (Bierema and Merriam, 2002; Single and Muller, 2001), but they concluded that due to the nature of applying e-mentoring in different contexts and the fast development of technology, it would be difficult to provide a completely developed model of e-mentoring.

The e-mentoring literature has agreed that e-mentoring relationships and programs must be structured and formatted in order to be successful. According to the presented models and frameworks, there are certain aspects that need to be addressed to establish a successful ementoring program and relationships. These aspects include the structure of the program, frequently scheduled communication, administrative support and evaluation. Group ementoring has proven to be beneficial in providing peer support and enhancing mentees' mentoring experience by learning from their peers. While the presented works agreed on how to structure e-mentoring, they failed to identify other aspects that may influence the structure of e-mentoring. The cultural attributes of a context might have an effect on the best ways to structure and apply e-mentoring.

2.3 Geographical Distribution of E-Mentoring

Headlam-Wells (2004) in her review of e-mentoring showed that the USA was dominant in the topic and development of e-mentoring, with some other work conducted in Australia, Europe and the UK. Similar to her findings, my review of the literature has shown that ementoring work mainly took place in the USA (DiRenzo et al., 2013; Hicks, 2012; Kalisch et al., 2005; Khan and Gogos, 2013; Khare et al., 2013; Lenear, 2007; Mason et al., 1994; Mollica and Mitchell, 2013; Muller and Barsion, 2003; Piliouras et al., 2013; Rhodes et al., 2006; Sanchez and Harris, 1996; Todd et al., 2016; Ware and Ramos, 2013; Wilburn et al., 2009; Wolfe and Gregg, 2015; Yalof and Chametzky, 2016). Modest activity on e-mentoring has been done in Canada (Letourneau et al., 2012; Li et al., 2010; Stewart and Carpenter, 2009; Wallis et al., 2015), UK (Headlam-Wells, 2004; Headlam-Wells et al., 2006; Hixenbaugh et al., 2006; Lamb and Aldous, 2014; Macafee, 2012; McCall, 2011), Australia (Redmond, 2015; Rickard, 2004; Sinclair, 2003; Woodley et al., 2015) and Malaysia (Dahalan et al., 2011; Fong et al., 2012; Ligadu and Anthony, 2015; Nor and Kasim, 2015; Rashid and Rahman, 2014).

Europe, Asia and South America have also had their share of participation. In Europe, studies have been conducted in Germany (Schimke, 2007; Stoeger et al., 2013), Greece (Petridou, 2009), Hungary (Dorner, 2012; Dorner and Kumar, 2016) and Slovakia (Bacikova, 2014). Other examples were applied around Asia and Africa, for example in China (Wu et al., 2012), Korea (Lee and Noh, 2003), Taiwan (Yang and Liu, 2002) and South Africa (Norodien-Fataar, 2012). Two studies were conducted in South America, specifically in Brazil (Mizukami et al., 2015) and Chile (Quintana and Zambrano, 2014).

In contrast to the growing body of literature on e-mentoring from countries around the world, published work on e-mentoring in the Arab region and specifically in Saudi Arabia is non-existent. One of the gaps that this research identifies is a lack of understanding of the cultural requirements for designing and applying e-mentoring programs in Saudi Arabia and similar contexts.

2.4 E-mentoring Relationships and Contexts

The traditional models for e-mentoring are adult to adult and adult to youth. E-mentoring has been applied in different contexts and served mentees across a wide spectrum of education and profession levels. In some cases of adult-to-adult relationships, undergraduate students were mentored by senior students (peer e-mentoring) mainly to provide support and increase their sense of social integration and student retention (Hixenbaugh et al., 2006; Mollica and Mitchell, 2013; Risquez and Sanchez-Garcia, 2012). Hixenbaugh et al. (2006) discussed student retention and reduction of student withdrawal in a department of psychology. Third year students mentored first-year students through email, which resulted in a greater sense of integration and satisfaction. Mollica and Mitchell (2013) also described an e-mentoring program between junior-level and senior-level nursing undergraduates to tackle the same problem of student retention in nursing. Results reported that e-mentoring helped in reducing students' anxiety, which led to increasing the nursing retention rate. In the study of Risquez and Sanchez-Garcia (2012), the program was open to first-year students from any department to help them with their transition to university. Each mentoring pair had a private asynchronous discussion forum where they communicated for 12 weeks. The results showed

psycho-emotional e-mentoring relationships developed and helped students to build a support network.

In other cases, undergraduate and postgraduate students were mentored by professionals to bridge academia and industry (Khan and Gogos, 2013; Rashid and Rahman, 2014; Wilburn et al., 2009). Wilburn et al. (2009) established an e-mentoring program to overcome the geographical challenge of their campus, which was located in a small town far from major recruiters. Mentors were accounting professionals from large accounting firms and businesses that were in larger cities, and mentees were undergraduate accounting students; they communicated through email. Mentees discussed curriculum choices and career opportunities with their mentors and were able to build their networks and learn about the profession in the real world. Similarly, Khan and Gogos's (2013) program connected biotechnology master's students with professionals in the field; participants communicated through email, online chat or video-conferencing. The program showed an increase in students' interest and satisfaction, and helped them identify their career options. Another ementoring program for undergraduates was developed to enhance interior design students' creativity (Rashid and Rahman, 2014). In the study of Rashid and Rahman (2014), students who had below-average grades in their first year were assigned to professional architects via a Facebook group. Comparison between pre-tests and post-tests showed enhancement of students' creativity in designing after e-mentoring.

In another context, some studies described e-mentoring programs for teachers, aiming to professionally develop pre-service, first-year or rural teachers. For example, Mizukami et al. (2015) established an online platform for novice teachers to communicate with experienced teachers. The study evidenced novice teachers' development and the distribution of knowledge between participating members. Redmond (2015) deployed an e-mentoring program that helped rural pre-service teachers in Australia with specific issues and challenges in different fields. For each field, one experienced teacher mentored a group of pre-service teachers in the same field; they interacted through Wikispaces. It appeared that mentees gained knowledge, collected resources and had opportunities to network with their peers. Likewise, another study was conducted in Chile to enhance the development of rural pre-service teachers through email (Quintana and Zambrano, 2014). Participating mentees agreed that the e-mentorship had a positive impact on their skills and practices.

Review of the literature found some e-mentoring relationships that were not common. Plummer and Nyang'au (2009) developed a reciprocal type of e-mentoring that was truly mutual. In this relationship, the two participating individuals took turns in acting as the mentor. For example, a US university professor and the director of a non-government
organisation in Kenya exchanged ideas, plans and experiences to conduct fundraising and research (Plummer and Nyang'au 2009). Another example is an e-mentoring program in Sri Lanka by Gunawardena et al. (2012). They reversed the model of mentoring to accommodate the needs of developing online learning in Sri Lanka. In mentoring, mentors are usually of a similar or older age but not in this context, where young adults mentored older professionals. Since Sri Lanka did not have enough experience in online learning, the project sought international graduates from the USA who were more familiar with online learning than Sri Lankan academics and professionals. The purpose of this project was to train academics and professionals in Sri Lanka to become e-mentors and online tutors through a communitybuilding model. In this project, age did not determine the mentors but their experience with online learning environments. A similar example was presented by Woodley et al. (2015), where university students in Australia mentored small business owners who had little or no ICT skills to support their businesses. Even though neither article (Gunawardena et al., 2012; Woodley et al., 2015) discussed the difference in age between mentors and mentees and whether this had an effect on their relationships, Woodley et al. (2015) showed that the mentors (university students) were encouraged and supported by experienced academic staff. They had to convince the mentors (university students) that they had the ability to perform the role of mentor to business owners who were older than them.

Review of the literature identified only one publication on youth-to-youth e-mentoring (Letourneau et al., 2012). The purpose of the program was to provide online support for young people with asthma from their peers through chat messages, email and bulletin boards. An increase in confidence levels and reduction of a sense of loneliness and isolation among the participating youth were key findings of the study.

E-mentorship programs differ in the connection between mentors and mentees; generally the ties are either one to one or one to many. In one-to-one e-mentoring, a mentor mentors one mentee individually, while in one-to-many e-mentoring, a mentor mentors more than one mentee as a group (Dawson, 2014). The literature shows cases of many-to-many or group e-mentoring, as presented by Kalisch et al. (2005) and Hicks (2012). The single case of youth-to-youth e-mentoring had a one-to-one relationship (Letourneau et al., 2012).

E-mentoring has been applied in different contexts and for different purposes. The previously discussed literature in this section, with their different relationships, contexts and connections, show evidence of the positive effect e-mentoring had on the mentees.

2.5 Means of Communications Used in E-Mentoring

E-mentoring is based on using technology to communicate between mentors and mentees.

Review of the literature shows a variety in communication methods, both synchronous and asynchronous. Email alone was dominant as the most frequently used method of communication in e-mentoring (Hixenbaugh et al., 2006; Janasz and Godshalk, 2013; Lamb and Aldous, 2014; Macafee, 2012; Quintana and Zambrano, 2014; Rhodes et al., 2006; Rickard, 2004; Wilburn et al., 2009). For example, Lamb and Aldous (2014) described an e-mentoring program to support gifted and talented high school students in physical education. For six months, undergraduate physical education students mentored gifted and talented high school students in a one-to-one relationship through email. At the beginning and the end of the program, the mentees visited their mentors on the university campus. Following the first visit, weekly emails were exchanged between mentors and mentees. The data was collected from emails, focus groups and questionnaires. The mentees demonstrated positive feelings towards using email to keep in contact with their mentor. They reported the positive impact of the two face-to-face interactions with mentors, which does not contradict the valuable contribution of e-mentoring in the context. They added that further research should address family support and coaching methods.

Headlam-Wells (2006) noted that the advancement of the Internet and CMC has provided advanced means of communication between mentors and mentees in comparison to earlier examples of e-mentoring which were limited to the use of e-mail between participants. Forums and discussion boards were also used as asynchronous communication methods. Adams and Hemingway (2014) presented an e-mentoring program where plant scientists mentored teams of school students to motivate and encourage their plant investigations. Each team and mentor interacted through a private discussion board. The post-study survey showed that mentors were motivated to inspire and increase students' knowledge of science. The study focused on analysing the content of the conversations to capture the mentors' mentoring techniques. The nature of the communication method used was not discussed in this study.

In publications describing e-mentoring, programs used chat (Letourneau et al., 2012) and video conferencing (Li et al., 2010) as means of synchronous communication. An ementoring program for youth with asthma and allergies demonstrated the use of chat sessions to provide social support by older peers with the same medical conditions (Letourneau et al., 2012). The program included the use of chat, email and discussion boards, and it was found that the weekly chat sessions were used more than the other methods. The study focused on the outcomes of the program and reported that follow-up interviews and questionnaires showed the mentees felt more normal and more confident, and had reduced their sense of social isolation.

Another study employed the use of video conferencing to connect mentors with rural secondary school students during their school hours (Li et al., 2010). The study aimed to increase the students' learning in math and science. The findings demonstrated that students were more interactive and more aware of the roles of scientists and mathematicians in real life. The students appreciated that the video-conferenced discussions with the mentors gave them the chance to be heard. The use of video conferencing gave the sense of having real contact with the mentors. Some mentees thought that the use of video conferencing was more exciting than face-to-face meetings. On the other hand, the poor internet connection created some technical difficulty and caused frustration among the students, and some sessions had to be rescheduled. Some students felt discouraged from discussing their comments with the mentors in front of the whole class. The study concluded that the use of technology was a successful implementation in the learning context.

Publications on e-mentoring demonstrated programs that employed one or a mixture of communication methods. As mentioned earlier, some programs used email as the only means of communication between mentors and mentees, while others combined more than one method. For instance, in the study of Lasater et al. (2014) the e-mentoring program for nurses encouraged the use of journaling and email; however, one mentor–mentee pair communicated through audiovisual technology (Skype) to create an electronic face-to-face meeting where they could be open in expressing their feelings and being more empathetic. In the study of Norodien-Fataar (2012), one group of mentors were trained to use existing platforms like learning management systems (LMS), while the second group were encouraged to use Facebook to contact their mentees. Results showed that mentors from all three groups used Facebook and MXit (the largest instant messaging and social network platform in Africa) more frequently than LMS. One mentor explained that the LMS was not an effective platform for mentoring because his mentee did not know how to use it, unlike Facebook and MXit, which mentees were already acquainted with.

Virtual worlds have become a platform for education and training (Todd et al., 2016); likewise, virtual worlds have been used as a platform to conduct e-mentoring, especially with youth with serious illnesses or people with disabilities (Cantrell et al., 2010; Todd et al., 2016). E-mentoring through virtual worlds was explored to test its role and efficacy for young people with serious illnesses; Zora (Bers and Cantrell, 2012; Cantrell et al., 2010) provided a virtual community for youth transplant recipients with their peers who were experiencing similar circumstances. Mentors helped the participants become more engaged in the virtual communities, provided support and helped them to find ways to cope with their illness. The

use of virtual world was to provide a safe space for the participating youth to develop and explore issues around identity (Bers and Cantrell, 2012). The results indicated that mentees created objects that related to hope, memories and friendship, but these objects did not have a significant effect on the mentees' hopefulness, for example. The findings reported that their parents favoured the use of Zora in comparison to other social media platforms due to its safe context (Bers and Cantrell, 2012). While the virtual world of Zora was a suitable and favourable context for youth (11–15 years old), this was not the case with an older population who used BreakThru. BreakThru (Todd et al., 2016) is a project based on the virtual world of Second Life. Mentoring islands were created in Second Life to advocate STEM to university students with disabilities. The virtual space was the main interaction space between mentors and mentees, but it was supplemented with SMS, Skype, email and Facebook. The results showed that participants mainly used communication methods that they were accustomed to (e.g. SMS and email) and recommended that the use of virtual worlds would be useful as a secondary or supplementary method.

Asynchronous communication methods have the benefit of flexibility in responding, where participants can respond and participate at their convenience (Single and Muller, 2001; Risquez, 2008), and provide a space for discussions to grow and last for longer periods of time (Redmond, 2015). On the other hand, the long time between responses may cause frustration and disconnection in the relationship (Smith-Jentsch et al., 2008). In contrast, synchronous communication requires scheduling and real-time feedback to conduct e-mentoring. CMC in general (synchronous and asynchronous) are perceived as mediums that increase the likelihood of misinterpretation and misunderstanding because of the lack of nonverbal communication cues (Ensher et al., 2003); this can be overcome by the use of audio and video conferencing to make a personal connection (Lasater et al., 2014). However, the lack of nonverbal cues in email and chat, for example, has the benefit of reducing biases in e-mentoring relationships (Ensher et al., 2003).

The advancement of technology presents different means of communications that have been applied in e-mentoring. It was noted that e-mentoring studies with older participants (i.e. undergraduates) were more flexible in choosing the communication method and some even used social media platforms to engage the mentees with the mentors (Lasater et al., 2014; Norodien-Fataar, 2012; Todd et al., 2016). These studies showed that they preferred communication methods that they were acquainted with and used frequently. On the other hand, e-mentoring studies of teenagers and school students used communication methods that ensured the privacy of their mentees; some designed a customised platform (Cantrell et al., 2010) and others used existing technologies or platforms that provided private spaces and

settings for their young participants (Adams and Hemingway, 2014; Li et al., 2010). Private communication methods were favoured by the parents of these participants (Bers and Cantrell, 2012). Therefore, the privacy of the mentees should be taken into consideration when designing e-mentoring programs for young participants.

2.6 STEM E-mentoring for Female Students

Despite a recent increase in interest in e-mentoring, the literature describing research on ementoring for young people is not extensive and is spread thinly across programs with different objectives that were applied in a range of contexts and mentee populations (Lenear, 2007; Wallis et al., 2015). While some programs aimed to provide support and guidance about life or career decisions, others focused on skill or academic development; some programs were open to all, while others had specific target mentees such as female students, young people or gifted students. For example, the studies of Rhodes et al. (2006) and Wallis et al. (2015) aimed to promote positive development, prevent behavioural, academic and psychosocial problems, and eliminate school dropout for all students, while in the study of Lamb and Aldous (2014) the aim was to support gifted students in physical education. Ementoring was utilised in advocating STEM for school students and providing support for STEM post-secondary students. Where all the STEM e-mentoring studies had a general audience for advocating STEM, some studies addressed specific issues and populations. Review of the literature showed that some STEM e-mentoring studies addressed the underrepresentation of women (Lee and Noh, 2003; Reid et al., 2017; Savoy, 2013; Single and Muller, 2001; Stoeger et al., 2013) or individuals with disabilities (Todd et al., 2016; Wolfe and Gregg, 2015) in STEM, while another study addressed the issue of geographical distance for rural students (Li et al., 2010). In this section, the focus will be on studies that were targeting female students.

MentorNet⁷ was established in 1997 and is an ongoing large-scale, nationwide ementoring community in the USA (Muller and Barsion, 2003; Single and Muller, 2001). It matches female undergraduate and postgraduate students with industry professionals (female or male) for a year-long email-based one-to-one relationship. MentorNet is based on the model of structured e-mentoring (Single and Muller, 2001). MentorNet applies a bidirectional matching algorithm. Based on the choices of the mentors and mentees, the algorithm searches the database and finds the best matches. Regular committed communications between mentors and mentees are expected and communications are set as a

⁷ www.mentornet.net

weekly exchange of emails (Single and Muller, 2001). Online training is mandatory in MentorNet and participants must complete the training materials. Weekly emails are sent to mentees and mentors to remind them to keep communicating and to help them develop their e-mentoring relationship; these emails are considered coaching materials. In an assessment of MentorNet in 2002 (Muller and Barsion, 2003), the results showed that mentees appreciated the support, hope and encouragement received from their mentors. They emphasised the benefits they gained from discussing the challenges of women in STEM with their mentors. Almost 90% of the mentees reported their intention to continue in their STEM fields. On the other hand, mentors indicated their appreciation of the personal satisfaction gained from their participation in MentorNet to share their knowledge with the younger generation (Muller and Barsion, 2003).

At the time MentorNet was established, in the late 1990s and assessed in early 2000, other technologies were not yet founded; this could be the reason why the study did not address the effectiveness of the communication method that was used (i.e. email) as there were no other options, in comparison to later studies mentioned in this section which made use of Skype, Facebook and MOOCs, for example. Recently, it was found on the MentorNet website that the length of the relationship and the communication methods have changed. The length of the relationship has shortened from one year to four months. It is also noted that MentorNet has kept pace with the emerging new communication methods. A YouTube video posted by MentorNet (MentorNet Program Account, 2016) that explains how MentorNet works mentions that mentors and mentees can communicate through on-platform chat, Skype, Google Hangouts, email or other methods of their choice.

In Korea 2001, an e-mentoring pilot study was conducted with a government program called Women into Science and Engineering (WISE) (Lee and Noh, 2003). Bulletin board systems were utilised to encourage female students of all school levels (from middle school to postgraduate) to pursue careers in science and engineering. The key behind using bulletin boards was that the main aim of the pilot study was to gather data about e-mentoring in the context, so bulletin boards were thought of as a good medium for measuring interaction and usage. After matching was done based on participants' area of interests, a bulletin board was set up for each mentor-mentee pair. The mentors and mentees showed some enthusiasm before the pilot study started, yet the results were surprisingly low. One third of both mentors and mentees (33%) wrote zero posts and only 5% of the mentees met the expectation of the study in writing 10 posts. In some cases, mentees or mentors initiated a conversation but did not receive any reply. A mentoring training workshop was held before the study for the mentors, yet 48 out of 111 mentors did not post a single message, which indicates that the

training was insufficient. The post-study survey (answered by 91 out of 143 mentees) showed that mentees had benefited from the program. Based on their study, Lee and Noh (2003) made a number of design recommendations for such approaches to e-mentoring, including: asking mentors to initiate the first post and introduce themselves; suggesting weekly discussion topics to mentors; and active administrative support. Lee and Noh (2003) discussed that the low participation could be due to: technical difficulties; the need for more training; or the fact that the study did not provide any administrative support to participants. In this study, the authors overlooked the wide age range of mentees; while 53.1% of mentees were graduate and postgraduate students, 7.7% were from middle school and 39.2% were from high school (a total of 46.9%). The study did not indicate which age level had no or low interaction. The age range is worth considering when exploring the reasons behind low participation because younger students might not believe in the effectiveness of e-mentoring and so will not interact with the program.

Women in Technology Sharing Online (WitsOn) is an e-mentoring study that utilised a MOOC as a medium for communication in the USA (Savoy, 2013). The choice of a MOOC was based on the high usage of MOOCs among higher education students. Savoy (2013) described WitsOn as different from other e-mentoring studies in that it was not structured and was self-directed. MOOCs are usually used for providing curriculum, but in this study the MOOC was used for making connections; WitsOn connected female undergraduates with professional women. Mentors shared their biographies, stories of their career paths and how they overcame challenges. Mentees and mentors could post questions and initiate discussions. In addition, each week a mentor created a video answering mentees' questions and posted it online for all participants. The findings of an online survey showed that mentees were mostly reading posts rather than interacting or contributing to the content, which is common with MOOCs users. Even with low participation rates, the mentees acknowledged the positive influence of the study on them; the component that had the most effect on them was mentors' career stories, struggles and biographies (Savoy, 2013).

CyberMentor is another example; it was used to advocate STEM to girls in Germany (Stoeger et al., 2013). A total of 800 female students (aged 11–18) participated in a one-year program where each mentee was matched with a female mentor who was either a college student studying a STEM subject or a STEM professional. Mentees and mentors exchanged weekly emails discussing STEM-related subjects. They could also interact through the members-only community platform, which included forum and chat features. The idea of the platform was to promote new role models to the mentees, showing them that their mentor was just one of many successful women in STEM. The program also introduced mentees to other

girls of their age who were interested in STEM. Among the mentees, there was an initial increase in interest in STEM activities and academic elective intentions in regards to STEM subjects. However, while the effectiveness of the program was equated with that of offline youth mentoring, the initially high level of participation by the mentees was not maintained through to the end of the program. It was observed that as the participants became familiar with the mentors and the platform, the number of interactions decreased. This has been identified in other domains; Stoeger et al. (2013) noted that a drop in participation rates after familiarisation is a common issue among online communities. In this study, the length of the relationship (one year) was not explored as a reason for the low participation; since the study showed positive outcomes, it is reasonable to consider that the girls stopped engaging after they had reached a level of satisfaction with the benefits of the program. The same problem of low participation rates was addressed by Savoy (2013) and both studies showed positive outcomes.

Reid et al. (2017) designed a structured program to help first-year female undergraduates who came from low socioeconomic backgrounds in Australia to choose and develop STEM career choices. Each mentee was matched with two mentors, one from academia and one from industry, based on an expression-of-interest survey completed by mentees prior to the program. Nine out of 47 of the mentors were male and the mentees could place a specific request to be mentored by female mentors. In addition, the program uploaded videos of interviews with successful women in STEM talking about their journeys and challenges. A Facebook group was created to facilitate communication between mentors and mentees in an informal setting. Participants also had the freedom to choose their communication method, including face-to-face meetings. Weekly informational emails about mentoring were sent by the program. At the end of the program, mentors and mentees were surveyed and interviewed. The survey results showed that a major increase was found in confidence about choosing STEM career options. The main struggle that was faced by mentors and mentees was to find enough time for the relationship. Results indicated that online communications were the most utilised methods, even among participants who were in the same city. While face-to-face communication comprised 30% of all contacts (137 contacts), Skype was used in 11% of cases and email in 30%. Both the Facebook group and the information emails were found to be very popular among mentees, while only 33% of the mentors found the Facebook group useful. The point of view on usage of Facebook in ementoring was not discussed in the study. The generational differences between the mentors and mentees could have contributed to the differences in their perceptions of the usefulness of Facebook as an e-mentoring channel or it may have been that the mentors did not think of Facebook as a professional medium for communicating with the mentees.

The discussed studies did not present details on the basis of the designed programs or what models they were based on, except for MentorNet (Single and Muller, 2001), which was designed based upon the model of structured mentoring. Even though the presented programs shared the aim of advocating STEM for females, they demonstrate a diversity in communication methods and e-mentoring relationships. Three out of the five programs faced problems in maintaining e-mentoring relationships and suffered from low levels of interaction, but all these studies showed positive outcomes and appreciation from the mentees. Several design implications can be derived from these studies. Most of the presented studies were structured and required frequent communication between a mentor and a mentee (Lee and Noh, 2003; Muller and Barsion, 2003; Reid et al., 2017; Stoeger et al., 2013). Some studies recommended sending informational emails and weekly discussion topics (Lee and Noh, 2003; Reid et al., 2017) and providing administrative support (Lee and Noh, 2003). Same-gender e-mentoring and the feature of sharing the biographies and experiences of STEM female mentors had positive impacts on the female mentees (Savoy, 2013; Stoeger et al., 2013). Group e-mentoring was found to be beneficial in advocating STEM e-mentoring for females because it creates a sense of community and shows the mentees that other girls their age are interested in STEM and are facing the same challenges (Stoeger et al., 2013).

2.7 Mentoring in Saudi Arabia

Traditional face-to-face mentoring programs in Saudi Arabia are uncommon and the existing research literature concerning mentoring in Saudi Arabia is limited and relatively recent (Abalkhail and Allan, 2015; Shukri and Leil, 2017; Sulphey and Allam, 2017). Prior work introduced in this section represents the only examples found of mentoring in Saudi Arabia, indicating that mentoring is not a well-explored topic in this context.

Abalkhail and Allan's (2015) comparative study of female managers' perceptions of mentoring and networking in Saudi Arabia and the UK shed light on their very different concepts, experiences and attitudes towards mentoring. Abalkhail and Allan (2015) noted the lack of mentoring literature in the Arab world, especially in Saudi Arabia, in comparison to the large amount of work conducted on mentoring in the Western context. Their study focused on understanding the differences and similarities between women managers in the UK and Saudi Arabia regarding their views and experiences of mentoring. The results showed that women managers in the two cultures had different experiences and conceptualisations of mentoring. Most of the women managers in Saudi Arabia reported that they had never been in

a mentoring relationship and that they lacked the chance to be connected to a male role model or mentor. On the other hand, most of the female UK managers in the study considered mentoring as an activity that took place in the workplace and had experienced it as such, whereas the female managers in Saudi Arabia considered mentoring as an activity in which they would engage with male family members. Abalkhail and Allan (2015) considered that the reasons for Saudi women managers to depend on their male family members for advancement in their careers could be related to a number of factors: 1) there are no mentoring programs; 2) men and women are physically segregated in the workplace; 3) women in Saudi Arabia have only recently reached higher management positions, which makes finding a female role model or mentor difficult; and 4) Saudi society has a collectivistic nature which provides support and protection for its members.

Health and medical schools are the dominant fields where mentoring in Saudi Arabia has been studied, through understanding students' perceptions about mentoring (Al Qahtani, 2015; Sattar et al., 2017) and presenting the perspective of mentees on their formal mentoring programs (Fallatah et al., 2018; Ghawji et al., 2017). Due to the non-existence of formal mentoring programs in universities, and across Saudi Arabia, Al Qahtani (2015) and Sattar et al. (2017) sought to understand students' perceptions of mentoring with the aim of developing formal mentoring programs. Al Qhatani (2015) studied the knowledge and perceptions of master's students towards developing a mentoring program, and Sattar et al. (2017) studied the views of medical students regarding mentoring and mentors. Al Qahtani (2015) showed that 92% felt the need for a mentoring program and 60% had positive attitudes towards mentoring. Similarly, Sattar et al. (2017) showed that almost 87% of the students stressed the need for having a mentor during their medical studies. Regarding their knowledge of mentoring, 70% ranked their knowledge from moderate to very good (Al Qahtani, 2015). In the study of Sattar et al. (2017) participants described the mentor's role as that of a "guide". Al Qahtani (2015) showed that there were no gender differences in the perceptions and attitudes towards mentoring and developing a mentoring program. Sattar et al. (2017) documented that the majority of students did not think that the gender of the mentor was important, while 30% reported that it was important to have a mentor of the same gender. Overall, both studies highlighted the students' need for mentoring during their studies and documented positive attitudes from the participants towards mentoring.

Ghawji et al. (2017) reported on a Saudi university mentoring program in which undergraduate medical students (of both genders) were assigned to a mentor (faculty member) at the beginning of their first year for academic and career-planning guidance and support. They highlighted the lack of studies on structured mentoring in Saudi Arabia, which

generated a gap in understanding and evaluating the effectiveness of such programs. Some resistance was found at the beginning of the program from the mentors. The program director and the college dean discussed with the mentors the importance of mentoring, their roles and the mandatory meetings with the mentees. An email was sent to the mentees that specified the program's goals and their responsibilities. The results showed no differences in responses between male and female students. However, it was found that many of the mentees never met their mentors. Third-year students had the most positive attitudes towards mentorship in comparison to students from younger years. It was clear that first-year and second-year students did not value the importance of mentoring, nor knew what they would gain from their relationship with their mentor. Ghawji et al. (2017) believed that third-year students during their study went through passive training that developed them into good mentees who had better relationships with their mentors. They concluded that communication and motivation were the main aspects of a successful mentoring program.

Likewise, Fallatah et al. (2018) conducted a mentoring program for fourth-year female undergraduate medical students to understand the characteristics of mentors that contribute to the development of mentees' academic performance. The study was initially designed for females because at the university the male and female students had different cohorts with different curricular structures. Each faculty mentor was assigned 10 mentees, where they had group and one-to-one meetings. One day of training was held for mentors. A survey was sent to mentors to measure their views on the importance of mentoring and their interest in participating. Based on their answers, mentors were classified into motivated mentors (52%) and unmotivated mentors. Mentees' attendance (group and one-to-one meetings) was higher with motivated mentors than with unmotivated mentors. Results showed advancement in the academic performance of mentees who attended the meetings regularly. The selection of the mentors was not an option by the mentees, which may have contributed to the low attendance rate in mentees with unmotivated mentors. The seniority of the mentor was a favourable factor for the mentees. In addition, the level of motivation and interest of the mentor had an impact on the effectiveness of the mentoring relationship. Overall, most of the mentees (83%) had positive thoughts about mentoring and believed in its values and benefits. Older students tended to value mentoring more than younger students (Fallatah et al., 2018; Ghawji et al., 2017).

On the other hand, Shukri and Leil (2017) reported the results of a peer mentoring program in an English language institute in Jeddah that aimed to help new teachers with their professional development and ease their transition to a new work environment and a new culture. The mentees were new female teachers from different Western nationalities who

came to work at an English language institute on a female university campus. The study questioned eight mentees and most (75%) agreed that mentoring eased their process of adjusting to a new culture and a new profession. All the mentees agreed that mentoring was a significant factor in overcoming the challenges of their new environment, especially cultural misunderstandings. They also agreed that observing their mentors helped them improve their teaching skills.

In all the studies reviewed, there was an emphasis on the non-existence of mentoring programs or literature in their fields prior to their attempts. They also emphasised the need for mentoring programs because previous literature in other contexts had proven the effectiveness of mentoring programs, so they stressed that these attempts were worth replicating and conducting in the Saudi context. From a cultural perspective, Fallatah et al. (2018) explained that their study was limited to females mentors to avoid the cultural challenges that might occur from having male mentors mentoring female mentees. On the other hand, Sattar et al. (2017) reported that only 30% of their participants were concerned with the gender of the mentor and preferred same-gender mentoring. While the literature around mentoring in Saudi Arabia is limited, most of these published studies have not addressed the cultural aspects of the context of Saudi Arabia and how culture might affect the application of mentoring. This presents a gap in research for further exploration about the role of Saudi in applying mentoring and e-mentoring.

2.8 Summary

This chapter has reviewed the limited number of published works that presented e-mentoring models or frameworks. The presented work agreed that for successful e-mentoring relationships, e-mentoring programs should be structured and require frequent scheduled interactions between mentors and mentees. Cultural requirements for designing and applying e-mentoring programs in Saudi Arabia have not been discussed in the literature due to the lack of published work on e-mentoring in Saudi Arabia or similar contexts.

While this review of the literature has found only a handful of articles that applied ementoring to advocate STEM for females, several recommendations for designing STEM ementoring have been identified. These recommendations are: (1) committed relationships, managed expectations, and regularity of contact between a mentor and a mentee (<u>Bierema and</u> <u>Merriam, 2002; Single and Muller, 2001</u>), (2) sending informational emails and weekly discussion topics (<u>Lee and Noh, 2003; Reid et al., 2017; Single and Muller, 2001</u>), (3) providing administrative support (<u>Lee and Noh, 2003</u>), (4) same-gender e-mentoring and group e-mentoring (<u>O'Neill, 2004; Savoy, 2013; Stoeger et al., 2013</u>), and (5) the use of

communication methods that created a private space for the mentor and mentee relationship (Adams and Hemingway, 2014; Bers and Cantrell, 2012; Li et al., 2010).

In conclusion, three factors will be addressed in this research: the context of Saudi Arabia, targeting females and the young age groups of the participants. This chapter has reported a gap in the literature about Saudi Arabia regarding the non-existence of e-mentoring programs and the lack of understanding of the role of culture in e-mentoring. Therefore, this research aims to identify which factors are important in the design of STEM e-mentoring for teenagers and young women in Saudi Arabia.

Chapter 3 Methodology

3.1 Introduction

Due to the lack of prior research on e-mentoring in Saudi Arabia, this research was exploratory and interpretive in nature. Stebbins (2001) mentioned that researchers carry out exploratory research when "they have little or no scientific knowledge about the group, process, activity, or situation they want to examine but nevertheless have reason to believe it contains elements worth discovering." Exploratory research in design gives a good understanding of the people and the issues that are being examined and typically ends with implications for design (Hanington, 2010). Because of the lack of previous work on e-mentoring in Saudi Arabia, user involvement in the process was essential in understanding the needs and requirements of e-mentoring in the Saudi context, therefore user-centred design (UCD) approach was adopted in this work. User-centred design is an approach for incorporating end-users in the design process and giving them a sense of ownership of the final product (Abras et al., 2004). It involves a variety of methods, including field studies, observing and analysing tasks and requirements, prototyping, user interviews, usability evaluation, surveys, and co-design (Vredenburg et al., 2002).

Exploratory research tends to be qualitative in nature, supplemented with quantitative data (Hanington, 2010; Stebbins, 2001). The qualitative approach was mainly applied in this research, supported with quantitative data to gain an understanding of e-mentoring in the area of the study. HCI research has turned from using measuring instruments and generating numbers to understanding users' motivation, needs and social norms, and how they think and feel (Adams et al., 2008). In HCI, qualitative methods are essential for understanding the area and users, gathering requirements and evaluating (Adams et al., 2008; Blandford et al., 2016). In this research, several UCD methods were used to elicit data and requirements including interviews, pre-study and post-study surveys, focus groups, co-design workshops and observation of interactions and posts written by participants. This research was conducted through exploring the context with a pilot study, gaining insights and requirements for design through co-design workshops, and design and evaluation of the new design (Table 3-1). Design methods for each study are described in more detail in the following chapters.

This chapter begins by presenting related work on co-designing and teenagers showing the significant difference in research design when working with teenagers compared to other ages. Later, it describes the UCD methods used in this research.

Study	Purpose	Research question	Data collection		
Study 1 Pilot study	1 Pilot Explore structured e- mentoring in the Saudi context		phone interviews, online pre- and post-study surveys, online focus groups, numbers and content of posts, written by participants		
Studies 2 & 3 Co-design workshops	Collecting design requirements	RQ2	problem statement stories, bodystorming, paper prototyping, designing the box, online surveys		
Study 4 Qudwa	Build a design based on the collected requirements, deploy it and evaluate it	RQ3	phone and WhatsApp interviews, online survey, observation of interactions, submissions by participants, analytics from social media promotions		

Table 3-1 Studies conducted in this research

3.2 Related Work: Co-design and Teenagers

Designing with adults and children has been the focus of HCI research and between these two populations lies the understudied population of teenagers (Fitton and Bell, 2014; Poole and Peyton, 2013; Read and Horton, 2016; Yarosh et al., 2011). Teenagers go through fundamental cognitive, biological, emotional and social changes in their transition from childhood to adulthood, and these changes have attracted the attention of few researchers in addressing their special needs and understanding the challenges in HCI and co-design related to this population (Fitton et al., 2013; Poole and Peyton, 2013; Yarosh et al., 2011). Their semi-tribal behaviour of moving in groups makes them a distinct and interesting user group and culture (Read and Horton, 2016). With their creativity and early adoption and use of technology, teenagers can provide different and innovative perspectives in developing digital solutions as they embrace technology in their own way to meet their own needs (Read and Horton, 2016).

In recent years, HCI communities have started to publish work on understanding teenagers (Fitton and Bell, 2014; Read and Horton, 2016), with data collection and methodologies suited to teenagers (Poole and Peyton 2013), as well as working with teenagers as designers (Ashktorab and Vitak, 2016; Mazzone et al., 2008; Vacca, 2017). Fitton and Bell (2014) reported that to understand teen development, three aspects should be taken into

consideration: 1) the cognitive, biological and social changes that happen in teenage years; 2) the social environment the teenager exists in; and 3) the psychosocial changes, which are both social and psychological (Fitton and Bell, 2014). Teens occupy a cultural space which is distant from that which adults occupy and this distance from the adult culture increases the autonomy of teenagers, but on the other hand creates problems in communication between adults and teenagers (Read and Horton, 2016).

Poole and Peyton (2013) discussed how traditional qualitative methods when designing for adults and children help in capturing their needs and requirements, while these exact methods can be a problem when working with teenagers due to the differences between adults, teenagers and children. In interviews, answers in some cases are limited to 'yes' or 'no' because the teenager sees the researcher as an authority who has power over them or feels discomfort with the researcher. Some suggestions to overcome this problem include using computer-assisted self-interview methods or giving the teenager the power to choose the interview location. On the other hand, focus groups were found to be helpful with teenagers (Poole and Peyton 2013). The benefits of focus groups for teenagers originate from the presence of peer support, social interaction and the advantages teenagers have over adults in their knowledge about their peers.

Literature on co-design and participatory design methods have been applied with different groups of teenagers. In their work around preventing cyberbullying in teenagers, Ashktorab and Vitak (2016) discussed the challenges and solutions for cyberbullying prevention in working with teenagers. They conducted five design sessions over a period of five weeks with high school students. The sessions consisted of focus groups, scenario centres, bags of 'stuff', mixing ideas and evaluating prototypes. Participants in the final design session were asked to reflect on their experience with co-designing. Participants appreciated the opportunity to be involved in creating solutions for a problem that affected them or people around them. The novelty of working with adults as equal design partners was an exciting aspect of the experience for them.

Vacca (2017) worked with Latina teenagers to explore the role of technology in providing emotional support. In this work, participatory design was used to understand the cultural identity and Latina emotional support system. A total of eight sessions were conducted with 13 Latina teenagers recruited through social media and afterschool activities. Participants explored emotional health, discussed scenarios, engaged in brainstorming and prototyping sessions, and reflected on the process. The findings emphasised that their nonsingular cultural identification (biculturalism) and living in two cultures played an important role in participants' technological designs for Latina emotional support.

Another study with disaffected teenage students presented the challenges faced in working with this specific user group (Mazzone et al., 2008). The purpose of the study was to develop a computer game with the involvement of disaffected teenagers to teach them how control and understand their emotions. Due to the lack of experience and lack of previous work on designing with a user group characterised by disruptive behaviour, Mazzone et al. (2008) had to conduct a couple of sessions before learning the best way to approach the participants and conduct the design sessions. Certain aspects of the participants, such as short attention spans, low motivation, critical behaviour and unpredictable attendance, forced the researchers to adapt and shape the sessions around the participants. Similar to the study of Ashktorab and Vitak (2016), it was found that the participants were excited to be given the opportunity to be part of designing a video game and they felt responsible. The researchers concluded that the design sessions contributed not only to the final design, but also to improvement in the behaviour of the participants and their progressive engagement in the sessions.

Co-design was also employed to understand the feelings of teenagers with diabetes (Glasemann and Kanstrup, 2011). Glasemann and Kanstrup (2011) aimed to gain insight into how these young people felt about their illness. The participants were supplied with a pack filled with different materials including glue, pens, a polaroid camera, a paper prototype of a personal digital assistant and modelling clay to help them mock up mobile diabetes supporters. The participants worked on their mock-ups for 45 minutes, followed by a presentation session. There were 12 mock-ups as a result of the design session. Glasemann and Kanstrup examined how the participants presented themselves through the designs as young and diabetic. Their examination resulted in addressing three themes: identity; the burden of dealing with their illness on a daily basis; and connectivity. The authors sought to understand the emotions and feelings behind the designs. They stated that it was as if the participants used a different language to express their emotions that could be described as a "particular sort of partly unconscious communication" (Glasemann and Kanstrup, 2011, p. 129). The study concluded that teenagers through co-design could express themselves and their feelings, and that it was a valuable method for understanding them.

Of particular relevance to this research, my review of e-mentoring literature identified only one article that included mentees aged 15–21 (middle and high school students) in a codesign activity and this only focused on the "visual" design of a website (Kalisch et al., 2005). Kalisch et al. (2005), in their program to advocate nursing as a career choice, connected students with volunteer nurse mentors. The initial design of the of the mentoring website was a joint effort between the research team of nurses working on the project and the web

developers. One key aim was that the website was attractive to the target age group. Therefore, the web developers visited schools to give students the opportunity to improve the look of the initial design (i.e. the colours, adding pictures and rearrangement of components). In this regard, through this research it was found that co-design activities and participatory design methods have not been addressed in designing e-mentoring with teenagers, particularly in the Saudi context.

3.3 Methods

3.3.1 Recruitment

The principal participants in this research were the mentees and the mentors. The recruitment purposes for mentors and mentees are different, therefore different approaches were used to recruit mentors and mentees. To recruit mentees, announcements were made through three channels: 1) a public scientific event (Genuino Day 2016); 2) schools and university; and 3) social media channels. One approach to recruit mentees and mentors for Study 1 (the pilot study) was to call for participation at Genuino Day 2016. This event was dedicated to females only with attendees of different ages. I gave a brief talk about e-mentoring and attendees who wanted to participate as mentors or mentees provided their information. School visits helped in recruiting mentees for Study 1. Participants in Study 2 (the co-design study) were recruited from the university mail list and the co-design workshops were conducted there. Study 3 (the other co-design study) was conducted on a school campus with school students. Single and Muller (2005) suggested the use of electronic communications in recruitment since ementoring is based on electronic communications. The final channel was to call for participation through social media networks such as Twitter, WhatsApp and Instagram, as they have been found to be tools to overcome the challenges that researchers face in research recruitment (Gelinas et al., 2017). Social media platforms with their advertisement settings provide specific targeting options such as age, location, gender and interests, and are a popular method to target hard-to-reach populations (Guillory et al., 2018). In Study 1 and Study 4, I broadcast an announcement to WhatsApp groups and individuals, and asked them to forward the announcement to their own contacts. Promotions on Twitter and Instagram were used for the recruitment and announcement of Study 4.

LinkedIn was mainly used for recruitment of mentors to overcome the challenges that were faced in finding Saudi STEM female professionals; not because of a lack of such professionals, but because of the absence of their visibility and communities, and networks of Saudi female professionals. I searched for Saudi female STEM professionals and approached them with the projects' aims and objectives. I also utilised my personal contacts to find

potential mentors. Professionals who agreed to be mentors recommended others through snowball recruitment.

3.3.2 Surveys, interviews and focus groups

Surveys, interviews and focus groups are commonly used methods in HCI (Lazar et al., 2010). Since the early years, surveys have been a main method for data collection in HCI (Sears et al., 2007). The popularity of surveys in HCI comes from their characteristic of being an easy-to-use tool that helps in gathering an overview of users' satisfaction, opinions, ideas and evaluations regarding a system (Lazar et al., 2010; Sears et al., 2007). In Study 1 (the pilot study) pre-study surveys were filled out by the mentees and their parents which included demographic data and their past experiences and expectations of mentoring and e-mentoring. The post-study survey was conducted to evaluate the pilot study. In Study 2 and Study 3 (the co-design workshops), the surveys collected demographic data about the mentees and their access to female role models. In Study 4 (*Qudwa*), the post-study survey was used as an evaluation method to evaluate the developed design from three perspectives: the nature of the relationship, communication channels and the impact of *Qudwa* on them.

Direct discussions with the participants or users provide deeper insights and understanding. Interviews and focus groups are useful for gathering requirements and evaluations of systems (Lazar et al., 2010). Semi-structured interviews are useful when little is known about the topic of investigation and to obtain deeper insights and design requirements (Lazar et al., 2010). In Study 1, semi-structured interviews were conducted by phone with mentors, mentees and parents. Interview questions concerned the respondents' participation in the e-mentoring program in general and also focused on the barriers and benefits of their experience with e-mentoring. Study 1 was designed to keep parents involved, yet none of the parents participated. It was essential to understand the parents' views, opinions and reservations about e-mentoring. An online focus group with some of the mentees supplemented the interviews in helping to initially understand their needs and requirements. At the end of Study 4 (*Qudwa*), semi-structured interviews were also conducted with mentors and mentees to discuss their experience with *Qudwa* and e-mentoring.

3.3.3 Co-design methods

Design practice has been influenced by changes in the perspective on user-centred design research. User-centred methods are most beneficial when designing products for users, but nowadays designers are designing for experiences, cultures and communities of people (Sanders and Stappers, 2008). Therefore, co-design has shifted from designing products for

users to designing for people's experiences and purposes. This shift has changed the role of the user from a passive role in user-centred design to an active role in the co-design approach (Sanders and Stappers, 2008). In co-design, the people who will benefit from the design (the users) play a large role in gathering knowledge and generating ideas; they play the role of the 'expert on their own experience'. User participation in technology co-design aims to elicit knowledge and ideas from expert users who know the problem and have been in the relevant situation (Vines et al., 2013). In addition, co-design gives power and agency to those who are usually ignored in the process (Vines et al., 2013). Giving power and agency to people to participate in the design of systems they will use results in successful participation in design by incorporating their understanding of the challenges and barriers. In Study 2 and Study 3, the employment of co-design workshops was intended to empower the target population in addition to gaining a deeper understanding of the perceived barriers to and opportunities for ementoring from their point of view. Four co-design methods were conducted: problem

In Study 2, the first workshop was intended to introduce the problem space to the participants. I used an animated video to tell a story, which in the research phase of the design process is known as "problem statement stories" (Gruen et al., 2002). Stories are tools used to encourage others to consider relevant or similar situations from their own experience. Most people understand and empathise with stories that have detailed characters, settings, goals and challenges. Empathising with stories prompts innovation and creativity (Gruen et al., 2002). The second workshop used bodystorming as a method to make the concept of mentoring more tangible to the participants. The role of acting in bodystorming helps to support empathy towards users and helps in them becoming familiar with unfamiliar situations, which results in better understanding of the problem (Oulasvirta et al., 2003). Rather than using familiar alternatives such as personas, participants were asked to play the roles of the mentor and the mentee to support their understanding of e-mentoring and the production of ideas for realising it (Schleicher et al., 2010). In the final workshop, the participants engaged in a process of low-fidelity prototyping using paper. The aim was not to produce detailed user interface designs, but to complete their understanding and validate the ideas generated in the previous workshops. Paper prototyping with young users is a preferred method for capturing, understanding and reflecting on the requirements and needs of the young participants (Glasemann et al., 2010).

In Study 3, the goal of the co-design workshop was to design an e-mentoring product that expressed the e-mentoring experience that the participants desired. The purpose of this activity was to translate participants' ideas and needs onto a physical object, 'the box'. This

box is a medium where information is physically presented (Gray et al., 2010). The outcome of this workshop was a box which captured and reflected the needs of the participants (Wienhofen et al., 2014). Wienhofen et al. (2014) described the outcome of this activity as a product that gives prospective users the "I want this and I want it now" sensation.

3.3.4 Thematic analysis

Thematic analysis (Braun and Clarke, 2006) was applied to all qualitative data: posts, interviews (with mentees, mentors and parents), focus group discussions and presentations. Thematic analysis is useful in participatory or UCD research where users have an active role in the process (Braun and Clarke, 2006). Since this research applied UCD methodology, thematic analysis was applied to help better understand the intended users.

In Study 1, Study 2 and Study 4, I applied the inductive approach to thematic analysis (Braun and Clarke, 2006), a bottom-up method where data coding is conducted regardless of researchers' assumptions. It is considered a data-driven methodology because themes originate from the content of the data. I studied the data, generated themes and validated them with my supervisory team.

For the data from Study 3, I applied deductive thematic analysis (Braun and Clarke, 2006), which is analyst-driven. Study 3 was conducted to validate the findings from Study 2, so the themes emerging from Study 2 were the basis of the deductive thematic analysis. All Arabic text was translated and is marked with an asterisk (*) at the end of the quotes.

3.4 Summary

Due to the exploratory nature of this research, user-centred design methodology was applied to explore and understand the users, their requirements and the context. Previous work showed how research design differed when working with teenagers compared to other age groups. It has also addressed the lack of applying co-design methods in designing e-mentoring programs for teenagers. Therefore, this research aims to apply several UCD methods in order to identify which factors are important in the design of STEM e-mentoring for teenagers and young women in Saudi Arabia.

Chapter 4 Pilot Study: Structured E-mentoring

4.1 Introduction

Due to the lack of research about e-mentoring in the Arab region and the Saudi context specifically, I have carried out a pilot study to explore the attitudes and behaviours of young Saudi women (16–18 years old) towards a structured e-mentoring program. The aim was to understand the feasibility of, and barriers to, the use of existing online platforms as a vehicle for female Saudi STEM role models to mentor young women in Saudi Arabia with the goal of increasing their interest in STEM careers. The research question addressed in this chapter is exploring the potential for applying a STEM e-mentoring program for teenagers and young women in the Saudi context.

The review of the literature in Chapter 2 concluded with several recommendations for designing STEM e-mentoring programs. In previous e-mentoring literature, there was a consensus that e-mentoring relationships and programs have to be structured and formatted to be successful. Bierema and Merriam (2002) stressed that certain factors must be in place in order for e-mentoring relationships to be successful. These factors are committed relationships, managed expectations and regularity of contact between a mentor and a mentee. In her review of the literature on best practices, Risquez (2008) divided e-mentoring programs into three phases: 1) design and planning; 2) implementation; and 3) evaluation. She stated that in order for e-mentoring systems to function effectively, it is necessary to apply the three phases properly. In a more detailed format, Single and Muller (2001) presented a model of structured e-mentoring. The model consisted of three phases: 1) planning; 2) structured implementation; and 3) assessment. In the planning phase, the overall idea of the e-mentoring program is developed, from identifying the participating population and matching process to setting program goals. The second phase is structured implementation, which is influenced by the program goals and is intended to aid the participants to make the most of their participation. The final phase focuses on evaluation of the program. Single and Muller's (2001) assessment involves collecting three types of data for evaluation: involvement data, formative data and summative data.

E-mentoring studies for young participants ensured the use of communication methods that created a private space for the mentor and mentee relationship (Adams and Hemingway, 2014; Bers and Cantrell, 2012; Li et al., 2010). STEM e-mentoring studies for females were structured and required frequent communication between mentors and mentees (Lee and Noh, 2003; Muller and Barsion, 2003; Reid et al., 2017; Stoeger et al., 2013). Sending informational emails and weekly discussion topics, and providing administrative support were

recommended for the design (Lee and Noh, 2003; Reid et al., 2017). Same-gender ementoring and group e-mentoring were favourable in advocating STEM e-mentoring for females (Savoy, 2013; Stoeger et al., 2013).

Given the lack of e-mentoring research and practice in Saudi Arabia, the initial design of this study was based on the lessons learned from the literature, the recommendations on ementoring programs for young people in other contexts and the following cultural assumptions: the use of textual communication for privacy and parents' involvement. The selection of textual communication as the mean of communication was based on maintaining the privacy of the female participants (Al-Saggaf, 2011). Parents' involvement has been considered an external factor that contributes to the effectiveness of face-to-face mentoring programs (DuBois et al., 2002; McDaniel and Yarbrough, 2016). DuBois et al. (2002) in their evaluation of face-to-face mentoring programs for youth reported that programs which supported parents' involvement showed a larger effect size than other programs. Therefore, the design choice to involve parents was because mentoring is not a familiar concept in Saudi Arabia and therefore parents might have concerns about their daughters talking to professionals they do not know, as well as increasing the effectiveness of the e-mentoring program.

In this study, I have followed Single and Muller's (2001) model of structured ementoring in conjunction with recommendations from previous studies on structured ementoring (Bierema and Merriam, 2002; Risquez, 2008) and lessons learned from previous ementoring studies.

4.2 Method

4.2.1 Platform selection

In this study, the search was for a platform that creates a free, safe online environment that supports textual communication (including Arabic) between participants and allows the involvement of parents. When it comes to mainstream social platforms, Facebook is the most popular social network, reaching one billion active users (Greenhow and Askari, 2017), and has been used as an e-mentoring platform (Rashid and Rahman, 2014; Ware and Ramos, 2013). Even though Facebook is popular worldwide, this is not the case in Saudi Arabia (Alwagait et al., 2015). Facebook has been criticised for its vulnerabilities in terms of privacy. Therefore, for educational purposes, other social networks such as Edmodo and Ning have been suggested as safer alternatives (Thongmak, 2013).

Ning is a commercial platform that allows the creation of your own social network. Ning provides a 14-day free trial. Hicks (2012) used Ning to create La Cuna, an e-mentoring

platform for the professional development of librarians. At the beginning of the study, Ning was a free platform. Over a period of six months, 38 librarians joined the forum and nine discussions were written within the first month, with an average of two comments per discussion. Hicks described participation rates as low in comparison to the number of participants. Results showed that seven participants were passively participating in La Cuna and four did not interact with La Cuna in any way after signing up. The timing of the study (the middle of the academic year), participants' perception of expertise (whether they considered themselves mentors or mentees), the manual setup of email notifications for new discussions and replies, and the surprising discontinuation of free access to the platform were the reasons that the e-mentoring program failed.

Edmodo is a free and secure platform available both as a web-based tool and as a mobile application that facilitates an educational social network aiming to connect teachers, students and parents (Figure 4-1). It looks like Facebook, but for educational purposes it is more secure and private, as it enables teachers to create and manage accounts, and only their students who are given a *Group Code* can access and join the group. Parents can also sign up and use a unique *Parent Code* that is accessed in their child's Edmodo account. Similar to Ning, Edmodo has been used as a platform for e-mentoring (Scott, 2013); that project was designed to provide fifth and sixth grade girls interested in science with the opportunity to communicate online with successful STEM female professionals. Edmodo was successfully used as an e-mentoring platform, resulting in a positive impact on girls and helping improve their attitudes towards science and interests in STEM. It was shown that the participating girls enjoyed Edmodo's format, ease of use, safety and security.



Figure 4-1 Edmodo home page allows creating accounts for teachers, students and parents

Edmodo has been employed in other educational settings in Saudi Arabia (Al-Kathiri, 2015; Al-Said, 2015). Al-Kathiri (2015) examined the prospect of integrating Edmodo into the instruction in a Saudi EFL (English as a Foreign Language) female high school. It focused on the perceptions and challenges of students concerning the use of Edmodo and its impact on their attitudes towards EFL learning. Study findings showed that using Edmodo improved the attitudes of female Saudi students towards EFL learning in comparison to learning the traditional way. Al-Said (2015) measured undergraduate students' perceptions of Edmodo and m-learning (mobile learning) in a Saudi university. Findings also showed that the perceptions of students towards Edmodo and m-learning were generally high and most students believed that learning with Edmodo facilitated and increased the effectiveness of learning communication and they appreciated Edmodo because it saved time.

Even though Edmodo is not generally used for e-mentoring, it fulfilled the contextspecific requirements of this e-mentoring structured program in two respects. First, it supports textual communication (including Arabic) between participants while maintaining the privacy of participants. Second, it provides a safe and secure online environment that allows the involvement of parents. The familiarity of Edmodo in the Saudi context and its safe environment, involvement of parents and emphasis on collaborative content and social networking appeared to make it the best fit for this study.

Edmodo is designed to be a safe forum-like platform that allows the creation of private groups, where each group has a Group Code that is required for access (Figure 4-2). Edmodo can be accessed through both a web-based interface and a mobile application. Teachers create a group and share the Group Code with their students. Teachers can also assign another teacher as a co-owner of the group. Edmodo allows teachers the ability to manage members of their groups and monitor group membership. When a student requests to join the group, the teacher will receive a notification. Once all required students have joined a teacher's Edmodo group, the teacher can lock the Group Code so that no other students can join.

Edmodo for Students			
Join your classroom in less than 2 minutes	Or you can		
Last Name	Sign Up with Office 365		
Group Code	G Sign Up with Google		
Username			
Email (Optional)			
Password			
Sign Up For Free			

Figure 4-2 Student registration form shows field for Group Code

Parents/carers/guardians (PCGs) can sign up and use a unique Parent Code (which is different to the Group Code) to access their child's Edmodo account (Figure 4-3). Teachers have access to the Parent Codes associated with the members of their group. Students and teachers can create, reply to and 'like' posts. Posts are either visible to all members of the group or private between a teacher and a student. Edmodo does not allow student–student private messaging. All discussions that take place in a group are public and supervised by the teacher who created the group; PCGs have read-only access to all posts written in their children's groups and they can also view all private messages that their child writes or receives, but cannot send or receive messages.



Figure 4-3 Parents can add their children by submitting a Parent Code obtained from their children's Edmodo accounts

4.2.2 Recruitment

The principal participants were the mentees and the mentors. The recruitment purposes for mentors and mentees were different, so over two months different approaches were used to recruit mentors and mentees. To recruit mentees, announcements of the program were made through three channels: 1) school visits; 2) Genuino Day; and 3) social media channels. I started by visiting girls' schools in Riyadh and talking to the administration. One school

which believed in the importance of the project agreed to let me talk to their students. The school set a date and time for me to visit and talk to the students. The school administration thought that grade 12 (high school) students were the target group that was in most need of the help provided by the project. I introduced myself and the project's purpose, and talked with the students about their plans. At the end of the meeting, I distributed information sheets about the project and consent forms to be collected on a later visit. Another approach was to call for participation at Genuino Day 2016. Genuino Day is an event held worldwide for the Genuino and Arduino communities. The Women's Program at King Abdulaziz City for Science and Technology held the event in 2016 and it was dedicated to females only, with attendees of different ages. At that time, I was assigned as the Mentoring Coordinator of the Saudi Arabian Chapter of Arab Women in Computing (ArabWIC) from January 2016 until January 2017. The Chapter was participating by presenting a talk by its coordinators and running a standalone booth. I gave a brief talk about mentoring and the Chapter, and had the chance to present the e-mentoring project. At the booth, attendees who had attended my talk and others asked about the e-mentoring program. Attendees who wanted to participate as mentors or mentees provided their information. Information sheets and consent forms were sent to them by email.

Single and Muller (2005) suggested the use of electronic communication in recruitment, since e-mentoring is based on electronic communication. The final channel was to call for participation through social media networks such as Twitter and WhatsApp. I broadcasted the announcement to WhatsApp groups and individuals, and asked them to forward the announcement to their own contacts, including King Saud University staff and student email lists. People who were interested contacted me by email or phone. I explained the project to them, answered their questions and sent them information sheets and consent forms by email.

In this phase, the challenge was finding female Saudi STEM professionals, not because of a lack of such professionals but because of their lack of visibility and the lack of communities and networks of female Saudi professionals. I used LinkedIn as a professional network to look for members with appropriate profiles. I searched for female Saudi STEM professionals and approached them with the project's aims and objectives. I also utilised my personal contacts to find potential mentors. Professionals who agreed to be mentors recommended others through snowball recruitment. Information sheets and consent forms were sent to all potential mentors.

4.2.3 Matching process

By the start of the study, the recruitment resulted in 22 Saudi girls aged 14–17 from different provinces in Saudi Arabia and 17 Saudi women mentors from a range of STEM professions. Although the parents of the mentees were offered the opportunity to join the program, none chose to do so. The participants (mentors, mentees and parents) took part in the study voluntarily, with the option to withdraw at any time during the study. Information sheets about the study, data collection and data privacy were distributed to all participants. I encouraged the participants to ask questions before and after joining the study. Consent forms were collected from all participants.

I applied the participant-choice matching method (Single and Muller, 2001). In doing so, I sought to strike a balance between privacy and appropriateness by involving a trusted intermediary (myself) to mediate the connection between mentors and mentees. The aim was to expose the mentees to the variety of STEM career options available to them by displaying the group of participating mentors. To maintain the privacy of the female mentors, I applied Single and Muller's (2001) recommendation not to reveal any contact information of the mentors. That is, all the choices made by the mentees were sent to the intermediary (myself).

First, I asked the 17 mentors to write a short descriptive profile about themselves and their work, using accessible language, that would attract mentees. It is not customary for Saudi females to post their own photos publicly on the internet (Guta and Karolak, 2015); therefore, I asked the mentors to pick an image of a cat that would represent their work. The use of cat images was also intended to lessen the formality between mentors and mentees. Later, I sent the collected 17 profiles, with the associated cat images, to the mentees via email and asked them to choose their mentors. As the goal was to have no more than five mentees in each mentor's group, I asked each of the mentees to choose three mentors and rank them in descending order based on their personal preferences. Mentees were then organised into groups based on their selection of mentors. This resulted in choosing eight out of 17 mentors from a range of STEM professions (mechanical engineering, space technology, microbiology and game development); five were based in Saudi Arabia and three in the USA (Appendix A.1).

Mentees were grouped to create a sense of community, to maximise the benefits of information shared collectively by mentors and mentees, and to show the mentees that other girls their age were interested in STEM and faced the same challenges (Stoeger et al., 2013). The choice of groups was also derived from the lack of understanding of the mentoring (e-mentoring) concept, so grouping was intended to encourage participation and help

participants learn and understand how others interact. Group e-mentoring facilitates learning from peers' mentoring experience (O'Neill, 2004). While group e-mentoring was the main structure of this study, Edmodo also provides the feature of one-to-one messaging between a mentor and a mentee, which enabled one-to-one e-mentoring.

4.2.4 Study design

Mentors and mentees communicated through the Edmodo platform. For each mentor, I created an individual group, gave the mentor the Group Code and assigned her as a co-owner of the group. I sent each mentee her mentor's Group Code to access their group via email. Mentees shared their thoughts and ideas with their mentor. Mentors and mentees could post to the group, comment on other posts and 'like' posts. All discussions were supervised by me and the mentor in that group. PCGs could sign up and use the unique Parent Code obtained from their daughter's Edmodo account. I monitored all activities between mentors and their mentees. Mentors and mentees were made aware that I and their PCGs had access to all their private messages. The text-based messages were in Arabic and/or English.

It was important to manage expectations by identifying the goal and purpose of the program, and the expected contact frequency (Bierema and Merriam, 2002; Single and Muller, 2005) to increase the likeliness of success. Participants were informed that the study would last for two months and that the minimum expectation of engagement was one hour per week through Edmodo. The goal of the program was to advocate STEM, connect young females to STEM female professionals and help mentees find answers to their questions about a future in a STEM career in Saudi Arabia. The role of the mentor was to share personal experiences, to ask questions and engage in conversations to build their relationship with the mentee, and to guide discussions.

Training and coaching are two aspects of the planning phase (Single and Muller, 2001). Due to the small scale of this study and to keep materials on hand for the mentors in their personal emails (Single and Muller, 2001), I used email as a method of training. Prior to the study, instructions on how to use Edmodo were sent to both mentors and mentees. Resources gathered from the web about e-mentoring and building mentoring relationships with mentees were sent to mentors through email. Weekly emails were sent to mentors, with some sent as reminders to engage with their mentees. Others were informational descriptions of mentoring or topic suggestions to prompt discussion with their mentees. Reminder emails were also sent to mentees. Some mentors were worried about their mentees not responding and they contacted me to get in touch with the mentee. Use of these kinds of email has been previously established as an effective coaching mechanism to increase engagement (Single

and Muller, 2005) and has been suggested as a way to reduce dropout (Lee and Noh, 2003; Stoeger et al., 2013).

4.2.5 Data collection

Single and Muller (2001) in their work on the model of structured e-mentoring recommended collecting three types of data: involvement data, formative data and summative data. Involvement data is concerned with the nature and frequency of mentors' and mentees' interactions in the e-mentoring program. In mentoring programs, the frequency of face-to-face interactions is considered a measurement of success; therefore, according to Single and Muller (2001), the frequency of email exchanges is an indicator of success in e-mentoring programs. Involvement data looks into the interaction between mentors and mentees by examining their relationship, frequency and engagement, and is considered an important indicator of success (Bierema and Merriam, 2002; Single and Muller, 2001). Formative data relates to the process of mentoring and helps with the modification and improvement of the program – it is the data that helps improve future iterations of the program. Program administrators collect formative data from participants' evaluations of the program and identify which elements of the program were beneficial. The data collected helps in reducing barriers and problems faced in this iteration of the program and also in recognising its points of strength. The final type of data is summative data, which addresses the effectiveness and outcomes of the program by evaluating the achievement of the program's goals and the value. Each program focuses on one or more goals, such as increasing mentees' knowledge, attitudes or behaviours within a certain context.

Involvement data in this study was collected from posts written by participants. Formative data was collected from nine individual interviews with mentors (n = 3), mentees (n = 3) and parents (n = 3), and a focus group with six mentees. Summative data was collected from pre/post-study surveys and mentor and mentee interviews. Interviews and the focus group were recorded and transcribed. Arabic data was translated into English and is marked with an asterisk.

4.3 Results and Analysis

4.3.1 Pre-study survey

A pre-study online survey was completed by the mentees and their parents, and it was in two parts: 1) demographic data; and 2) mentoring experience and expectations. The demographic data, such as parental age, education level and profession, was completed by the PCGs. The second part was about the mentees' experiences and expectations of mentoring and ementoring (Appendix A.2).

Out of 22 mentees, 10 answered the pre-study survey. The results, shown in Table 4-1 below, show that four of the parents (two mothers and two fathers) were illiterate and four other parents (two mothers and two fathers) had only primary school education. Two fathers worked in STEM fields (engineering and computer science); however, none of the mothers did. Seven mentees knew adults who worked in STEM fields and five mentees knew females who worked in STEM fields. No participants or their family members had ever participated in a mentoring or e-mentoring program of any form. Participants joined the program to "engage in a new experience"* and "to gain more knowledge in subjects I'm interested in and make more decisive choices on my college major"*. One girl emphasised the lack of family members who could guide her: "My family do not have a scientific background, and I have big dreams and I need those who guide me"*. Four mentees joined the program via social media announcements, three mentees joined the program from a friend's recommendation and one mentee joined from my school visit. Two mentees specified that they knew about the program from other sources; one knew about it from her teacher and the other from her older sister.

		Fathers	Mothers
Age			
	30–45	1	3
	46–60	5	7
	61 or older	4	0
	Deceased	-	-
Level of education			
eaueunon	Illiterate	2	2
	High school	2	2
	Bachelor's degree	6	2
	Master's degree	_	2
	Doctoral degree	—	2
Major			
	Science	_	-
	Technology	1	_
	Engineering	1	-
	Math	_	_
	Other	8	10

4.3.2 Level of engagement

While participants were asked to spend a minimum of one hour per week interacting in their groups, which would be equivalent to at least one interaction per week (8 interactions per

participant for 8 weeks), the results showed significantly lower levels of participation than this, even for the most active participants. The numbers of posts, replies and likes were used as an indicator of the level of engagement of the mentors and the mentees (Adams and Hemingway, 2014). In their study, Adams and Hemingway (2014) characterised their mentors based on the average number of comments, with a total of 1086 comments written by 170 mentors with an average of 6 comments per mentor. As a result, they characterised mentors with 1–3 comments as "low", mentors with 4–7 comments as "medium" and mentors with more than 8 comments as "high". In this study, I followed Adam and Hemingway's (2014) approach to characterising participants. The total number of posts and likes was 121 (110 posts and 11 likes). While 8 was the expected number of interactions, 4 was the average figure recorded over the 8 weeks. Therefore *Low (L)* engagement was assigned to participants with between 1 and 4 interactions, *Medium (M)* for 5 to 8 interactions and *High (H)* for 9 interactions and above (13 was the highest number of interactions recorded).

Table 4-2 summarises the numbers of posts, replies and likes for all groups. To identify the mentors, I refer to them by their group identifier and activity level. For example, the mentor of group A, with a medium activity level, is identified as A_M. Mentees are identified by their activity level and their mentor's identifier. For example, mentee B in group A, with a low activity level, is identified as B_LA_M. Mentees who signed up and joined the group but did not participate are described as passive participants (P) and mentees who chose a mentor but did not join the group are described as not registered (NR). Owing to the limitation of not having access to log files, I was not able to determine whether passive mentees had followed discussions (but not contributed) or were completely disengaged.

	Posts	Replies	Likes	Total		Posts	Replies	Likes	Total
Group A					Group E				
A _M	1	6	0	7	E _H	6	5	0	11
ALAM	0	2	0	2	ALEH	0	2	2	4
B _L A _M	0	2	0	2	B _L E _H	0	2	2	4
C _L A _M	0	2	0	2	C _P E _H	0	0	0	0
D _P A _M	0	0	0	0	Group F				
E _P A _M	0	0	0	0	F _L	2	0	0	2
Group B					$A_{\rm L}F_{\rm L}$	1	0	0	1
B _M	1	5	1	7	B_PF_L	0	0	0	0
$A_M B_M$	0	5	0	5	Group G				
B _{NR} B _M	N/R	N/R	N/R	N/R	G _H	4	9	0	13
Group C					$A_{\rm H}G_{\rm H}$	0	6	2	8
C _H	8	2	0	10	$B_L G_H$	0	2	1	3
$A_M C_H$	0	5	0	5	C _M G _H	0	4	1	5
B _L C _H	0	3	1	4	Group H		-	-	-
Group D					$H_{\rm H}$	4	5	0	9
DL	2	1	0	3	$A_L H_M$	0	3	0	3
$A_L D_L$	0	2	1	3	$B_{\rm L}H_{\rm M}$	1	0	0	1
$B_P D_L$	0	0	0	0	C _L H _M	0	4	0	4

Table 4-2 Numbers of posts and likes of members in all groups

While participants were asked to spend a minimum of one hour per week interacting in their groups, the results showed significantly lower levels of participation than this, even for the most active participants. Group A had the largest number of mentees, but their activity level was low or passive (Figure 4-4). Mentor A_M in her last reply said: "I'm busy at this period of time, I will be back soon to answer your questions"*.



Figure 4-4 Group A had the largest number of mentees, but their activity level was low or passive.

Even the participants identified as H participants had lower levels than expected. H participants in some groups did not necessarily have many interactions with members of the group. Mentors E_H and C_H (Figure 4-5) had more than once posted more than one new post in a single day, which increased their level of participation to H, but some of these posts did not receive any response from the mentees in the group. The first month of the study witnessed most of the interactions. At that initial stage, the mentors were introducing themselves and the mentees were curious to know more.



Figure 4-5 Group C shows a high-activity mentor who wrote more than one new post in a single day, even though some of these posts did not receive any response from the mentees in the group.

Later, all the groups had gone inactive except for group G, which was engaged in discussions about the video game they were designing. Mentees did not respond or address each other, although all posts and replies were public within each group. All communications were between a mentor and a mentee. In group F, three new posts were written but without any interaction between their members. Mentee A_LF_L initiated the first post in her group: "Hello. When can we begin the discussion about architecture and pursuing it as a major and career?". She did not get any response. After two days, the mentor of that group Mentor F_L wrote a new introductory post, disregarding the first post by Mentee A_LF_L (Figure 4-6).



Figure 4-6 Group F shows no interaction between participants. Mentee A_LF_L initiated the first post in her group, but mentor F_L disregarded her post.

In group B, regular posts were exchanged between Mentor B_M and Mentee $A_M B_M$. When the mentee did not respond to the mentor within ten days, the mentor wrote another post. Then Mentor B_M sent an email to me asking about her mentee: "I wonder if there is a way to check on my mentee. I don't know if she still receives my messages or I somehow lost her." I contacted the mentee and informed her that her mentor was concerned about her. She replied to her mentor and apologised, saying she had not received notifications about the posts.

4.3.3 Analysis of posts

Thematic analysis (Braun and Clarke, 2006) was applied to the qualitative data: posts, and interviews (mentees, mentors and parents). I applied the inductive approach to thematic analysis (Braun and Clarke, 2006), a bottom-up method where data coding is conducted regardless of researchers' assumptions. It is considered a data-driven methodology because themes originate from the content of the data. I studied the data, generated themes and validated them with my supervisory team. A total of 110 posts were coded and there was significant variation in the content. I briefly describe the themes extracted from the posts to give an overview of the topics discussed. Three themes were identified within conversations: 1) request; 2) inform; and 3) engage & motivate.

Request

Different types of requests were noted by mentors and mentees. The majority of requests were

informational requests about study and careers from the mentees:

But I do not know exactly what it is like to study in universities and what possible jobs I could get if a choose physics as a major? This also applies to mechanical engineering* – Mentee A_LA_M

Mentees also requested resources in the form of books, websites or applications. One mentee asked her mentor a networking request. She asked her mentor to connect her with researchers at the centre she worked at:

I'm interested in working at ... so I thought this is the ideal place if I want to get involved in research ... I want to ask you if you know any researchers working in biotechnology if you could link me to them directly. – Mentee $C_M H_M$

On the other hand, mentors had different types of requests; they requested background information about their mentees. They asked their mentees to talk about their future plans. Mentors also requested feedback on previous suggestions:

Who you are \ldots talk about yourself? How do you see yourself in 3 to 7 years from now? – Mentor A_M

Inform

Mentors provided information about their studies and careers. They also explained the differences between study majors within a field (e.g. between studying electrical and mechanical engineering). Mentors discussed their job responsibilities and research areas. Mentor H_H talked about being among the first group of females working on space technology and the difficulties she faced:

It is a new field for all of us. We were the first group of females who worked in this field. Beginnings are always special. We needed to learn a lot to be able to continue with our work. We had online courses. Every now and then we decide on a course book and we study it together.* – Mentor H_H

Mentors provided their groups with suggestions and resources with or without a request from their mentees. Mentees gave feedback about suggestions or resources made by their mentors:

Thank you for the reply :) I might delve into the history of mathematics, and might as well take a few physics online courses because I think I need more practice in that subject. – Mentee B_LA_M

On several occasions, it was noted that mentors were justifying their absence and not responding very often:
I apologise for not replying sooner, but from now on I'll be posting every weekend.* – Mentor G_H

Engage & motivate

Mentors used a range of approaches to engage mentees, including asking mentees both personal and general questions, and encouraging mentees to research their answers and discuss the questions within their group. They aimed to give the mentees a chance to look up answers and discuss them within their group:

Have you heard about a lost satellite? How does a satellite get lost? What do you think they mean by lost?* – Mentor $H_{\rm H}$

Some mentors tried to establish common interests and personal connections. Mentee $A_M B_M$ mentioned that cooking was one of her hobbies and her Mentor B_M told her that they shared the same hobby and that she planned to cook two new recipes every month. Mentors shared personal pictures of their labs or workplaces as a way to engage the girls and make a personal connection. Tasks and projects were another way mentors attempted to gain engagement. Mentor G_H started a small project with her mentees to design a video game. The girls got excited and discussed the characters and story of the video game:

Mentor G_H : Let's start developing a horror game. Before developing, we need to design it. Without a design and plan, the project could take more time than expected and cost more. There is no one right way to design a game. It differs from one studio to another ... Let's start by choosing the world, characters, plot and way to play.*

Mentee A_HG_H : It would be nice to use space as the world for the game, Mars for example, since we have more information about it. The main character is a human. The story, for example an astronaut who is lost in space and trying to survive. We can add scary things like black holes, aliens or starbursts.*

It was noted that some mentors motivated the girls by addressing them as "future engineers" or "future scientists" and encouraging them to pursue their dreams. Mentor C_H shared a link that listed the top 20 engineering universities in the USA and described how she had been accepted in one of them after applying three times. In six groups, mentors communicated using an informal language and style with their mentees. Mentees and mentors used emoticons, such as smiley faces, flowers and hearts. Two other mentors had a different style of writing and the mentees in these groups were identified as low or passive. Mentor E_H , who is a lecturer at a public university, addressed her mentees as "students" and the group as a "study group" and she asked if they had understood the "lesson". Mentor F_L wrote two posts using a letter format:

Dear Young Ladies,

I hope this finds you well.

First, let me introduce myself ...

Hope to hear from you soon!

4.3.4 Interviews with mentors and mentees

For the post-study interviews, I selected participants who exhibited different levels of participation: one mentor with medium activity levels and two mentors with high activity levels, and one mentee with high activity levels and two mentees with medium activity levels. Mentees and mentors with low activity levels did not respond to my requests for interviews. The semi-structured interviews were conducted by phone. Interview questions concerned the respondents' participation in the e-mentoring program in general and also focused on the barriers and benefits.

As the interviews were semi-structured, I combined deductive and inductive techniques of thematic analysis (Braun and Clarke, 2006). In contrast to the inductive approach, the deductive approach is top-down and is driven by the researcher's interests. The combination of inductive and deductive approaches resulted in data-driven and interviewdriven themes. Some of the interview questions are listed below:

- What expectations did you have about the e-mentoring system?
- Why did you join the program?
- Why did you choose your mentor?
- What barriers or problems did you face during your participation?
- Did you share your experience or what you have learned with others?
- Would you join an e-mentoring program again?

The main themes that emerged were misunderstanding of e-mentoring, awareness of the low level of interaction, communication method as a barrier and benefits of e-mentoring.

Misunderstanding of e-mentoring

Two of the mentees had different expectations about e-mentoring. Mentee A_HG_H thought that the goal of the e-mentoring program was to introduce them to university majors, which was "what everyone is doing" and she thought the idea was "cliché". Mentee A_MC_H expressed that her expectations had been lower than what she experienced in the program. Her assumption was that it would be a Q&A session, but it turned out to be "something better". On the other hand, Mentee A_MB_M thought that the program was as she expected it to be. Mentor B_M and Mentor C_H expected to be guiding the mentees on a specific project with certain tasks and that they would be "the facilitator that will help them through some sort of electronic communication platform". Mentor H_H loved the idea of mentoring, but she was not prepared. She explained: "I started to ask people from work about interesting topics to share with those girls."

Awareness of the low level of interaction

Of the mentors and mentees who were interviewed, all agreed that they had expected more interaction. Mentors expected the girls to have more questions and requests, while mentees assumed mentors would actively initiate discussions and activities. Some said that having more mentees in a group might increase overall engagement. However, despite all communications between a mentor and a mentee being visible to all members of a group, mentees did not interact with each other. Mentors described their attempts to engage their mentees, but suggested that the mentees' low levels of engagement might also be due to girls of this age having reservations about being able to express themselves clearly:

I tried to use different methods and topics and see which way would attract the girls the most. – Mentor $H_{\rm H}$

As mentioned earlier, there was no interaction between mentees. The following comments extracted from mentees' interviews exemplify their thoughts about interacting with their group members:

I think we should get to know each other but I don't know how \dots like an introduction post about the girls. – Mentee $A_{\rm H}G_{\rm H}$

None of the other mentees wrote something that we could respond to or discuss with each other.* – Mentee $A_M C_H$

Communication method as a barrier

Both mentors and mentees agreed on the difficulty they experienced using text as the only means of interacting with each other and reported finding it challenging to express themselves as a result. They raised a number of issues with the asynchronous text communication of Edmodo, including the long response times, the time required to write posts and the mentees' struggles to express themselves:

I didn't like the long response time.* - Mentee A_MC_H

Mentor C_H on different occasions mentioned that using text was a barrier for her and her mentees: "Interaction using text is very slow ... Using text was an obstacle for me." She also commented about how using text could have affected the level of participation of the girls. She thought that using text from the girls' point of view would mean leaving a "trail" or a "mark". Mentor B_M talked about her situation being based in the USA and how the time difference that accompanied asynchronous communication was a problem:

The biggest limitation to me was the time difference. There is more than 10-hour difference than time in Saudi. I couldn't reply to her at the same moment she wrote to me. If she sent me something at 12 pm Saudi time, I could be going to bed and I might answer her after 7 hours. There is this big disconnect. Through this disconnect maybe she sent me something she was excited about and then she went on with her day and whatever she wanted to know or she was excited about might disappear. – Mentor B_M

She also mentioned that using the written platform to communicate required time to actually sit down and think about something to write after a very long day of work. Mentors and mentees found it challenging to express themselves or to understand others:

It takes a long time to explain our points to each other.* - Mentee A_MC_H

It would help me to know her way of expressing herself, I could know her demeanour, is she a shy person for example. We can connect in a different way. This is what I really wanted. I don't think that text would provide that missing part, but I thought that it would be really good if we were talking instead of writing ... There are lots or things missing and usually, you need to write less to write faster. – Mentor B_M

It was evident from interviews that participants wanted another method of

communication to build a stronger relationship. Suggestions were to use audio or video channels for communicating:

It will be more exciting if it was a video call especially with a group of girls, we could brainstorm and exchange ideas \dots Teenagers will be more excited if it was a video call meeting and we could know each other faster.* – Mentee A_HG_H

As a main way of communication, it should be something that we can talk in like Hangout or Skype ... There are verbal and nonverbal cues that you could get from an audio/video communication, you can know if she is thinking and you help her think or is she shy and silent. She can learn from me the way I talk and discuss things. Mutual benefits. – Mentor B_M

When I asked Mentor B_M if she had any concerns regarding scheduling an audio/video call with the time difference situation, she said that if there is a will, there is a way. She stated that sessions could happen during work breaks or early in the morning before going to work. On the other hand, Mentee $A_M B_M$ thought that the main problem was using a platform that nobody used and suggested platforms that were more popular. She explained: "Using a social communication network that was faster and easier to use, and everyone is using it already, so I won't forget to log in and check Edmodo."*

Benefits of e-mentoring

Another theme that emerged from the analysis of the interviews with mentors and mentees was the benefits of e-mentoring; since this is related to the outcome of the program, it is considered summative data. Although the levels of interaction were lower than anticipated, mentees showed a considerable increase in awareness of STEM careers and opportunities for females in Saudi Arabia. Girls were impressed by the achievements, studies and careers of their mentors. Prior to the study, two mentees reported not being aware that such careers existed for Saudi females. The study also had a positive impact on some girls who had decided to study a STEM major beforehand. One mentee, who liked to play video games, said that she had considered studying computer science but did not have a detailed understanding of the subject. However, thanks to her interactions with her mentor, she realised that game development was a career option for a computer science major. Another girl was relieved to know such a career existed for Saudi females:

Before joining the program, I was upset because I did not know that such career and jobs were available for females in Saudi and I thought that if I pursued this kind of study I would have to get a job abroad which is impractical and unlikely to happen. When I knew about the opportunities for females regarding science & technology, I was relieved, and I know now that I can pursue my dreams.* – Mentee A_MC_H

Some mentees showed how they shared the knowledge they gained from or about their mentors. One mentee shared a picture of robotics with her school team. They were in a robotics competition. The picture was posted by her mentor, showing a robot designed in the lab she was working in. Other mentees found mentors' career so interesting that they talked about them with their parents and friends. Mentors were encouraging and motivating by giving them examples of what they had accomplished:

One of the girls was interested in robotics when I saw this post on Facebook I remembered her and I wanted to share with her this post about top universities. I'm in one of these universities so I wanted also to show them that as a Saudi female you can achieve and get into any school you want if you have put your mind into it. I got into [university name] which is one of the top universities in its field after applying 3 times and got accepted the third time. – Mentor C_H

Mentor B_M and Mentor H_H shared their experiences with their colleagues at work. Mentor H_H said: "When I talked to people at my work about the program, they were surprised that there are girls interested in their future at this young age."* Mentor C_H and Mentor B_M talked about a different kind of sharing; they talked about sharing the e-mentoring experience with a wider audience: There is a part of the whole experience that can be shared with other participants. Sharing impressions, tips. – Mentor B_M

During the duration of the program I attended a conference and I wanted to share my experience with the others. If there was something like a twitter account, we follow and it could share or retweet my experience that would be nice. Or maybe a hashtag that we all use to post about our experience with e-mentoring. It will bring attention to e-mentoring. – Mentor C_H

Mentor B_M shared her previous experience with school students. She previously worked at an interactive science centre in Riyadh, designing science shows and programs for school students. From her experience, she found out that young Saudis had a conception that accomplishments and successes in STEM cannot be achieved in Saudi Arabia or by Saudis. To raise their awareness and hopes, she added male and female Saudi scientist to the shows, but the students felt that those people were exceptions and what they had done was extraordinary and unusual:

It is important in our society ... There are a lot of things happening that are not shared in our own culture, and if it is shared it will be shared in a context that these people are outliers, or this thing happened out of the extraordinary. And it's not, it's more common than we think. – Mentor B_M

Mentor H_H described her mentee asking for an internship opportunity at her workplace; she said: "it felt like networking." Even though the concept of mentoring is not common in Saudi Arabia, mentors agreed on the importance and empowering potential of mentoring. None of the mentors reported having been in a mentoring relationship before, as either a mentee or a mentor. They explained their personal motivation to mentor in relation to their desire to, or need to, have been mentored themselves earlier in their careers. Mentors believed that their lives and professional experiences should not be limited to themselves and that they should give back and share these with younger generations to give them hope and encouragement. The concept of mentoring was seen to be important and empowering, especially in Saudi Arabia:

We were a generation that needed support and we wanted to communicate with experts to guide us, especially when no one in your family or people around you have the same interest.* – Mentor H_H

It's empowering in so many ways, it has guidance and your experience is not limited to yourself it's part of giving back ... Having this platform that we can actually share the experience that led us to be where we are now with someone younger or someone at a crossroads. That would be very very rewarding at so many levels and at the same time empowering not only to me as a mentor but to the mentee it will leave an impression and I think that this is really important. – Mentor B_M

4.3.5 Interviews with parents

I sent requests for semi-structured interviews to all parents of mentees and three mothers agreed to participate. Mother#1's age range was 45–60 and Mother#2's and Mother#3's age range was 30–45. Mother#1 was an assistant professor in the College of Arts, Department of Arabic Language, Mother#2 was an English teacher and held a bachelor's degree in Languages and Translation (English) and Mother#3 held a bachelor's degree in Special Education and worked in the Ministry of Education.

Despite the platform's configuration, none of the parents actively participated in the program; therefore, it was essential to understand the reasons for this. I was keen to understand parents' views, opinions and reservations about the idea of e-mentoring. The themes that emerged from the mixed (inductive and deductive) thematic analysis of the parent interviews were appreciation and need for mentoring, own struggle and experience, concerns and mentoring and society. Some of the questions that were asked included:

- Have you ever been mentored? By whom?
- What do you think mentoring is?
- What do you think the benefits of mentoring are to your daughter?
- If your daughter asked you to join an e-mentoring program that uses video or audio communications, would you agree? Why or why not?

Appreciation and need for mentoring

All three mothers were appreciative of the program and stressed the need for and importance of it. Mother#1 informed me that she had attempted on her own to educate students at her daughter's school (which was a government school) about her major, but the school had told her that the Ministry of Education had refused the idea of having an external individual talking to the students. She also told me that she had been the one who told her daughter about this program, as she saw it as a good opportunity:

I really admire the effort you put into the mentoring program and I liked the pilot study that my daughter joined, she really benefited from it. It is something that serves the whole society, we have a lack of awareness about careers and studies. You can turn it into something big later on, like a social institution and others can sponsor it.* – Mother#1

The other mothers also expressed their appreciation and the need for mentoring:

I wish I had had someone to guide me or answer my questions. In my day the choices were limited, you became either a teacher or a doctor.* – Mother#2

If there is a chance that my daughter can meet some professional woman who is working is such a field, that would be a wonderful opportunity!* – Mother#3

Mother#2 talked about her younger daughter (12 years old), who was fascinated with space and astronomy, but they were concerned because they did not know about any female Saudi professionals in this field. They came to a decision: "The last thing she decided is that in the next two years, if she couldn't find a Saudi woman working in this field, she will study law and become a lawyer. There is no awareness in schools or any other place about women and their careers in Saudi"*.

Own struggle and experience

Through the interviews, the mothers shared their stories of the challenges they faced when their daughters were applying to universities and how they overcame these struggles. They also described the significant effort their daughters put into finding answers to their questions about course options.

Mother#2 explained: "I wish there were programs or events that introduce the girls to universities and careers ... I struggled personally with my daughter. I took days off from my work to take my daughter around universities in Riyadh. We wanted to know more about the opportunities available"*. She continued talking about their visits to universities, which were not as beneficial as they had expected. Later on, her daughter had attempted to make a detailed comparison between three majors she wanted to study using all the resources she could find, including websites, the university-provided course information and even by contacting current students who were studying these majors:

Luckily my daughter found a girl who is older than her and used to study at the same school. She got in touch and started asking more questions about the major and the university.* – Mother#2

Mother#3 told me that they had lived for a couple of years in the UK and recently come back to Saudi Arabia. She elaborated: "They asked the students to write a plan for their future and how they would imagine it. When we came back, my eldest became frustrated because what she dreamt of might not be applicable for a girl in Saudi"*. She explained that her daughter had "no clue" about how to find out more or who to ask about the field that she wanted to study, so she (her daughter) attempted to contact girls and women who were studying and working in the field and asked them about the things that concerned her.

Concerns

Two mothers (Mother#1 and Mother#2) had some concerns about the mentors' political,

religious and social views causing conflict between the mentee and her family, and the mode of communication. Regarding political and religious views, the mothers felt that these topics should not be discussed. Mother#2 showed concern that the mentor might take advantage of the young girls: "She [the mentor] might also tend to have political or religious views that are against our beliefs. It worries me a mentor might take advantage of this channel and use it to convince these young minds with their thoughts. If there are censoring and supervision of these calls, I might feel more comfortable"*.

Another concern of Mother#1 and Mother#2 was that they worried that the mentor would encourage the mentees to disobey or create conflict with their families. Even though she knew that her daughter would not be easily convinced by anyone, Mother#1 was worried that some girls might be. She talked about her relative's daughter, who was encouraged by her teacher (who was supposed to be a role model) to disobey her parents and the mother eventually had to transfer her daughter to another school:

A relative of mine has a very smart and gifted daughter, she always gets As in school ... That teacher supposed to be a mentor was forcing and convincing the girls to choose majors such as marketing and she told them to disobey and disagree with their parents if they advised them to choose majors like science or computer science. The mother did not like the way she [the teacher] was convincing the students and that she created a conflict that did not exist between the parents and their daughter.*

Mother#1 explained that the parents did not have a problem with any major their daughter wanted, but with the negative way that the teacher was encouraging them which created a conflict between the student and her parents. Mother#2 emphasised the importance of mentors not interfering in family decisions. She explained:

She should not interfere in family matters either. For example, if the girl wanted to study a major that is not available in Saudi universities and she has to study abroad but her parents are against the idea, the mentor should not encourage the girl to insist and rebel against her family, which might create problems between the girl and her family. If she tried to convince the girl peacefully and give her arguments that she could discuss with her parents or the mentor suggested that she talk to the parents and explain her point of view, I think that is fine.*

She also suggested that mentees should have the choice of changing their mentor: "It is also important that the girl can decide if she wants to continue with the same mentor or not"*.

Mother#3 did not share the same concerns and explained that every opposite opinion to theirs is available on the internet. She explained:

To be honest with you, the world has changed. The girls are part of the open world now; we cannot control what they see or read or know anymore. The internet has opened everything for them. I don't have issues with trust because I believe I raised my kids well enough to know that if there is something wrong, they will talk to me about it and I don't think the mentors would raise a topic that is not already there on the web.*

With respect to mode of communication, the three mothers did not encourage the use of video channels but accepted that their daughters were using these to communicate with their friends, and ultimately agreed that these were acceptable where the purpose would be beneficial for them. Mother#1 explained that the differences between generations meant she did not seek to exercise excessive control over her daughters' behaviours as she knew that they could do whatever they want behind her back:

In my opinion, I wouldn't encourage it, but I know that they use it. I would prefer audio conversations between them. You know, the differences between generations make me not strict in controlling my daughters because I know that they can do whatever they want behind my back. I would prefer to allow them and be in the picture.*

All the mothers did not feel the need to actively participate in the program themselves and explained this in various ways: they wanted their daughters to talk freely, they trusted their daughters and they believed that their daughters lived in an open world now where everything was available to them and they cannot control what they watched or read or who they talked to. Even though they did not participate, knowing that they could participate was important to them. Parents' participation might occur in cases where they were not convinced that what their daughter was learning was appropriate or where they wanted to participate in discussions on certain topics.

Mentoring and society

From the three mothers' point of view, the importance of mentoring meant that the benefits to society were greater than the benefits to the individuals. It would enhance their sense of belonging to society and awareness of others in the same position and of opportunities. It was a chance to network among others who shared their interests:

It is something that serves the whole society.* – Mother#1

Mentoring would give them safety and confidence that they are not alone in this, career options or other opportunities. To get introduced and network with females in their field. Mentoring is voluntary, which would make the girls appreciate their mentors and their society because they would feel cared about and that society is supporting them.* – Mother#2

I want the mentor to make the girls believe in themselves and their society.* – Mother#3

4.3.6 Focus group

Six mentees who exhibited different levels of engagement joined an online focus group to talk about their views, participation in the program and how the experience could be improved. At the beginning of the session, I asked the participants about their understanding of mentoring and their fears towards their future career wise, and then the participants shifted the discussion towards e-mentoring without guidance from my side. The participants felt more comfortable conducting the focus group in Arabic. The session was recorded, transcribed and translated.

Regarding their understanding of what a mentor was, one mentee said that she honestly did not understand what a mentor was, while two mentees described a mentor as an experienced person who guided you through a new phase:

A mentor helps you when you go into a field that you know nothing about or you have questions about something career wise, then they could guide you to the best way for you.* – Mentee B_LA_M

When you start something new, they are other people who went through similar experiences who you can learn from. – Mentee $A_{\rm H}G_{\rm H}$

Five of the participants shared fear of the unknown about their major and career. Because they did not have enough knowledge and information about majors and work opportunities, they worried that they would make the wrong decision and regret it:

I worry that when I finish my studies and have a career, I will not be satisfied with my job.* – Mentee $C_L A_M$

I really fear that I might study a field that I will not like and I would feel that this is not what I want to do in life. I worry that I might regret choosing a major or career that will not fit me.* – Mentee A_HG_H

While one mentee was not worried, she believed that if she felt that she had chosen a major that she did not like, she could gain a master's degree in another field. Mentee A_HG_H responded to this thought:

I think choosing a bachelor's degree is very important even if you choose to complete your master's in another field. A bachelor's degree gives you knowledge and information to build on and it would help you in your master's even if you chose something relatively different to your first major.*

They were worried, but they talked about their own process of seeking information about a certain field or career (e.g. talking to people, visiting labs and reading about the field): I talked to undergraduates who are in the field that I want to study. I asked them about the positives and negatives of the field. The most important question that I asked was who the perfect candidate is to study this field. From their answers, I could decide if it applied to me or not. I visited labs, universities and hospitals to know the environment of my future career.* – Mentee B_LA_M

I asked a lot of people around me in different fields, even in fields that I'm not interested in. The most important question that I asked was what the negatives of your career or field are, to see if these negatives will affect me or I will be fine with them. I also read about fields online. I might know what I'm not interested in, but I don't know what I want.* – Mentee $A_M C_H$

Three of the participants described the challenges they faced in finding professionals

to talk to, especially women, and the difficulties they experienced in identifying the right

questions to ask professionals who they would have no chance of meeting again:

Every person has their own experience, no two people will have the same experience. I either talk to people who graduated a long time ago or are still studying. I fear that if I choose a field, it will be different from theirs. Most of the people I talked to were male, not female. If I found a female and I could talk to her, I could relate to her more than if I ask a male about his experience.* – Mentee C_LA_M

I think that talking to people would not help me in choosing my major, but I believe it would help me in learning about their life experiences, their networks or connecting me with other people.* – Mentee $A_M B_M$

After that, the discussion moved to their e-mentoring experiences and they discussed how they had wanted it to be. All six mentees demonstrated a preference for not being required to have an exclusive relationship with one mentor, but rather having the freedom to choose who to talk to and for how long:

I think it would be much better to have and know more mentors. I'm clueless about all the fields and opportunities around me. That's why I would like to learn more from the mentors. Even if I'm not choosing her field as my major, I would be interested in knowing more about different fields and I might talk more than once with more than one mentor.* – Mentee C_MG_H

Mentee A_MB_M suggested adding some personal information about the mentor in addition to her career and professional profile; she explained: "Not just about her career, but also something personal. It would help me choose"*. Mentee C_MG_H showed her support of the idea of adding personal information about the mentor: "I liked what she said about adding personal information to the mentor's profile"*.

The mentees indicated that before committing to the mentor-mentee relationship, they would prefer to schedule a short (virtual) meeting with the mentor. Such a short meeting would help them decide whether they wanted to commit to a longer term relationship:

I could talk to more than one mentor for 15 or 30 minutes and then I would pick the one who I felt would help me the most and continue my sessions with her.* – Mentee $A_M B_M$

If I found one mentor who I wanted to build a mentorship relationship with, we could communicate with text, for example if I have a quick question, rather than scheduling a meeting. – Mentee $C_M G_H$

Participants' views differed on the nature of the e-mentoring session. Four mentees preferred being in group e-mentoring with a limited number of mentees (maximum 4 mentees); they explained that other participants might discuss or ask questions that they had not thought of. One mentee suggested that a one-to-one session could be scheduled afterwards when she chose to be committed to a specific mentor:

Sometimes people ask questions that never occurred to you, but the answer could really be helpful to you. Exchange of knowledge! I would suggest limiting the number of participants to 3 or 4 girls in group sessions to reduce interruptions.* – Mentee $C_M G_H$

I might start with group sessions and when I choose a specific mentor, then I would prefer one-to-one sessions. – Mentee $A_M C_H$

One mentee disagreed and preferred to have a one-to-one session with her mentor, but

she did not mind joining a group session. She explained:

Group mentoring could be helpful sometimes if I didn't know what I want. Usually when I choose to talk to someone, it means I know what I want to ask, and I would like it to be private.* – Mentee $A_M B_M$

One mentee was neutral between one-to-one or group sessions. She said that her

choice would depend on her purposes with e-mentoring. If she was just exploring, then she

preferred group mentoring, while if she had a specific domain, then she preferred one-to-one:

It depends if my questions are very general and I would like to explore the field, then I don't mind having a group session with others. But if I'm in depth with my questions and I need specific answers to my questions, I'd rather have a one-to-one session to get direct and clear answers without interruptions from others.* – Mentee B_LA_M

The issue of the seriousness of the program was also raised and they expressed a concern that some girls might waste the mentor's time or not show up. To this end, they suggested the application of a charge for not showing up, banning non-attendees from scheduling further meetings for a period of time or rating mentees by mentors so other mentors had the option to not accept meeting requests from a low-rated mentee:

I would recommend a fee to so the girls will know it is something serious and won't waste the mentors' time.* – Mentee A_MB_M

Sessions could be for free but if you missed a session without excuse, you should pay for it and the money would go for a good cause.* – Mentee B_LA_M

At the end of the session, the participants discussed the importance of mentoring and the ways to attract girls who were not interested in participating:

The idea of mentoring should be explained in a clear way for girls to understand its value. When I joined the mentoring program earlier, I didn't understand what it was exactly, but I joined because I wanted to learn about it, and I found it a very interesting experience and useful and not what I imagined.* – Mentee $A_{\rm H}G_{\rm H}$

I would really like to know about all the fields available and opportunities. A wider knowledge. I don't want to miss good opportunities just because I didn't know about them.* – Mentee $A_M B_M$

There was an argument between three participants on whether or not e-mentoring

could change girls' views and make them more interested and concerned about their future:

Mentee $A_M B_M$: I don't think that a 30-minute talk with a mentor would change the perspective of a girl who is not interested in thinking about her future.*

Mentee B_LA_M : I disagree, I think it could happen that a girl who is not interested in her future could be influenced by successful people by talking to them and learning about their achievements.*

Mentee $A_M B_M$: But examples of successful people are around us everywhere! You don't need to talk to them to know about them.*

Mentee A_HG_H : I agree with [Mentee B_LA_M]. One of my friends is really smart, but she does not have any plans for her future. No one mentors, guides, inspires or encourages her. When I try to advise her, she laughs because I'm her peer and I don't have enough experience and knowledge. That's why I think if e-mentoring exists, it would help her to shape her future even though she is not interested.*

4.3.7 *Post-study survey*

Mentees were asked to complete a post-study survey about their experiences with the ementoring program. It consisted of two parts: 1) open-ended questions; and 2) Likert-scale questions. It was answered by 8 out of 22 mentees.

Results showed that 7 of the 8 respondents described their e-mentoring experience as a positive one. 7 of the 8 respondents were satisfied with their mentoring relationship and described it as: "professional and nice"*, "respectful"* and an "amazing student/teacher relationship, she enthuses me"*. One mentee was not as satisfied: "Not as developed as I would have liked it to be, I feel like we only know each other on a name basis". Respondents said that the best thing about their participation was: "exchanging experiences"*, "discussion

and dialogue"* and "My mentor's personality was extraordinary, she changed the way I think"*. On the other hand, two respondents had a different experience; they responded: "nothing special"* and "If enough effort had been put into it [from the mentees and mentor], it would've been a wonderful experience"*.

Regarding the problems faced during their participation, five respondents agreed that lack of persistent communication and long response times between posts were the main struggles. Three mentees had different responses. One said that the timing of the program was a problem as it clashed with final exams. Another mentee thought that the mentor was not initiative in communicating with the group. One respondent expressed that she was not comfortable with the platform (Edmodo). Seven respondents said that they gained knowledge about the fields from experts; one explained: "I learned more about the facilities and resources in our country related to the fields I want to pursue later on." Another responded: "My love for science has grown"*.

Four respondents' suggestions for improving the program were mainly to set fixed times for communicating with the mentors. One suggested working on related projects, while two suggested using another platform, "a more approachable platform for communication. Ex: WhatsApp". The second part of the survey showed that 5 girls indicated that their awareness of STEM careers had increased. All 8 respondents agreed that the e-mentoring program did not consume a lot of their time and that they were confident to write and express their opinions with their mentors. The mentees were asked if being in a group with other girls was distracting; 7 disagreed. Additionally, 5 mentees agreed that they would participate in other e-mentoring programs in the future.

4.4 Discussion

4.4.1 Language, culture and mentoring

Language is the main means by which social lives are conducted. Language as a means of communication is strongly connected with culture in several complex ways (Kramsch and Widdowson, 1998). It is important to highlight the absence of words for 'mentoring', 'mentor' and 'mentee' in the Arabic language. Kramsch and Widdowson (1998) described the relationship between language and culture by saying "language expresses cultural reality". People use words, phrases and language to refer and reflect on their experiences, ideas and common practices in their community and culture (Kramsch and Widdowson, 1998). The absence of the word 'mentoring' from the Arabic language reflects the fact that mentoring is not an established concept in Saudi Arabia and the Arab region, and helps explain the

participants' lack of understanding of what mentoring is.

This lack of understanding of mentoring is exemplified in the findings. The interviews showed that one mentee thought e-mentoring would be a Q&A session and another thought the program would introduce them to different majors. Both mentees expressed that what they experienced in the program positively exceeded their expectations. Some mentors misunderstood their role at the beginning. Two mentors described mentoring as guiding through a project or task. One mentor highly appreciated the idea of mentoring, but she was not "prepared" and did not know what to do.

Even though the study was on a small scale, it is important to highlight the diversity of demographics of the parents of the mentees. The study showed that the need for mentoring was not limited to girls with parents who had low levels of education, but also extended to those with parents who specialised in STEM or were highly educated. Through the interviews, the mothers (all three of them educated) showed the need for mentoring; all three mothers told stories about how they and their daughters were looking for someone to talk to, to help their daughters. All three mothers took some time in the interview to genuinely thank me for creating this opportunity for their daughters; they believed that mentoring was beneficial not just to individuals but to the whole of society. In their opinions, mentoring would create a more connected society; when girls at this age received voluntary help from people they did not know, they would have a stronger sense of belonging and would want to give back to their community in the future. The findings of the interviews show that both mentees and mentors shared their experience and knowledge gained from the program with people around them at school, home and work.

These findings help in understanding that the absence of mentoring in the context created a greater need for mentoring. This need was recognised by the mentors, mentees and mothers. The findings of this pilot study have implications for e-mentoring to be a base on which to develop communities with mutual benefits.

4.4.2 Relationship trajectory

The results showed low levels of interaction between participants and the discontinuity of the relationships. Different conditions may have led to the unsustainability of relationships. For example, groups with mentors who used an informal conversational style had high or medium activity levels (e.g. groups C and G), while groups with mentors who used a formal and distant style had passive mentees or mentees with a low level of activity. While I cannot exclude the fact that this could be due to chance and the fact that some mentees had mentors who were more conversationally active by nature, the interaction style likely played an

important role here. Mentor E_H gained a high activity level indicator because she wrote a couple of posts in a single day without getting any response from her mentees; she also addressed them as "students" and used words like "study group" and "lesson", which might explain why even though her activity level was high, her mentees' activity level was low. On the other hand, in Group F (Figure 4-6), Mentor F_L disregarded her mentee's first post in the group and created a new post which resulted in three unrelated posts (1 by the mentee and 2 by the mentor); this action of disregard from the mentor might have caused the mentee to discontinue her participation in the group.

However, some mentors made significant efforts to connect and build a relationship with the girls they were mentoring. Mentor B_M tried to make a personal connection and asked about her mentee's interests and hobbies, and they found out that they shared the same hobby. Mentor B_M contacted me when her mentee suddenly stopped responding to posts; her concern and interest in her mentee on different levels helped in increasing the interaction level of her mentee ($A_M B_M$). The mentee also spoke positively about her mentor in the interview: "I really liked my mentor and the idea of the program. She was very helpful and kind"*.

Creativity was a significant factor in the sustainability of the relationship in group G. In a single case, Mentor G_H sensed that the girls were losing interest and communicating less frequently, and she responded by initiating a small project with them to design a video game. The number of posts and interaction level increased among her mentees (A_HG_H , C_MG_H and A_LG_H). In the first 4 weeks of the study, weekly materials about mentoring, topic suggestions and reminders were sent to mentors, yet mentors struggled to attract and engage their mentees. One mentor mentioned the benefits of such emails and suggested getting more tips during the study about mentoring and topic suggestions.

There was no interaction between mentees in any group, even though all posts were public and visible to all group members. This also may have weakened relationships in the groups. The results showed that mentees did not see the necessity of responding to each other; in addition, one mentee said that an introductory post about the members of the group would be helpful and support interaction among mentees in one group. The inability to interview passive mentees and low-activity mentors made it difficult to understand the reasons for their low engagement. Out of 22 mentored girls, 5 were identified as passive participants and one did not join her group. Access to log files was not possible and therefore I could not establish whether passive and low-activity participants were logging in and reading or were completely disengaged. The target group of mentees' commitment issues could be another reason for the unsustainability of the relationships. Girls at this age might feel burdened to commit to a long relationship.

Through the interviews, mentors gave their own explanations of mentees' low levels of interaction. Mentors had doubts about mentees' ability to talk confidently or felt textual communication created hesitation in terms of leaving a record. The results of the post-study survey contradict these thoughts. It shows that all 8 respondents felt confident to write and express their opinions with their mentors.

During the focus group session, an issue around the nature of the e-mentoring relationship was raised. It was readily apparent that the participating mentees were seeking flexible and less committed e-mentoring relationships than those provided by existing approaches to e-mentoring. Mentees explained that being in contact with more than one mentor would give them more information and knowledge about opportunities and work fields. Again, this different understanding of mentoring from participants could be a result of the absence of mentoring in the culture, as explained earlier.

Overall, two issues around participants' relationship were identified in this study. First, mentorship is not common in Saudi Arabia, so scaffolding in the platform could be designed to guide mentors and mentees in their mentoring relationships and help maintain these. Second, the findings raise questions about how to change the format of e-mentoring in terms of length, modality and commitment of e-mentoring programs.

4.4.3 Overestimated effect of cultural context

As discussed earlier, the concept of a mentor is new (for mentors, mentees and parents) in Saudi Arabia and this research has identified only a small number of articles addressing mentoring in this context (Abalkhail and Allan, 2015; Al-Mutairi et al., 2015; Ghawji et al., 2017).

From the outset, I was aware of the potentially significant cultural differences, particularly concerning the privacy requirements of the female participants (Al-Saggaf, 2011). This was the basis of my selection of text as the only means of communication between participants in the study. The findings showed that text communication was perceived as a major barrier. In all interviews, mentors and mentees reported that they felt the need to communicate in a more expressive manner, which would help them to understand each other and allow them to more readily express their personalities and characters. The majority of participants who responded to the post-study survey (6 out of 8 respondents) considered the communication method and platform as a barrier to their participation. Their complaints were around the long response times that the asynchronous method required and the unfamiliarity of the platform. The negative comments about the communication method were that the response times between participants were long, they were unable to express themselves easily

and that the act of typing consumed more time than other communication methods. One mentee of those who responded to the post-study survey suggested the use of a more commonly accessible platform.

All three mothers through the interviews indicated their acceptance of the use of different channels where they believed their daughters would genuinely benefit from this. This difference in cultural norms from widely reported expectations is likely to be related to the high rate of internet and social media usage in Saudi Arabia, which has led to the rapid adoption of such technologies (Bafakih et al., 2016; Winder, 2014).

The incorporation of parental involvement in the structured e-mentoring deployment in this study was a design choice that stemmed from the novelty of the mentoring concept in Saudi Arabia and the conservative nature of the Saudi community. I aimed to reassure parents about the process of mentoring (which was new to them) and to address anticipated parental concerns about their daughters talking to professionals who were strangers to them. Although parents had the opportunity to participate, the results show that none of the parents felt it necessary to do so. They believed that the rapid changes in the online world make it hard to control what their children watch, learn or use. They trust their children and trust that they raised them well to differentiate between right and wrong.

As mentioned earlier, my personal experience of being a Saudi woman in a STEM field has driven this research. As a Saudi my knowledge of the Saudi culture has also influenced my assumptions and expectations. The findings showed unanticipated insights regarding the cultural assumptions deployed in this study. The reasons behind the differences between what I expected and what was found, could be related to the differences in the way of thinking between me as an adult and between the teenagers (the mentees) and to the differences in the experiences that I had when I was their age. Another reason could be that the high use of the internet and social media in Saudi Arabia made Saudis more flexible to new technologies and online experiences.

4.4.4 Measurement of impact

The quantitative data on interaction frequency and duration did not meet expectations. Some mentors posted many posts that did not result in any replies or even likes. Others posted more than one different post in a single day without any response from their mentees. By examining interaction sequences (e.g. responses to others vs. isolated posts), I noted that high participation did not imply interaction between members and so could not be a measurement of success.

Conversely, qualitative data showed the impact of the program on the girls despite the low levels of interaction. The program raised their awareness about Saudi females in STEM and available opportunities for them in the future. Mentors and mentees shared their e-mentoring experiences with colleagues, friends, school and family. The study made a positive impact on the participants that resulted in them sharing their experiences with others.

Three years after the study, I followed up with the participants to gain insight into whether the e-mentoring program had influenced them. Only three responded. Mentee A_MB_M stressed that her love for science grew even more after her mentoring relationship with her mentor. She also mentioned that she still remembered her mentor and how she had a positive influence on her. She asked if there were any upcoming e-mentoring programs. Mentee B_LA_M explained that she had joined the project as a new experience. The major she was interested in was not among those of the participating mentors, which might explain why she had a low activity indicator. She suggested for wider options of majors to be added so that more girls would benefit. Mentee A_HG_H told me that her participation in the e-mentoring program had made her more determined about her choice. She added that the program had changed her perspective on the future and widened her ambitions, because she had learned about the different mentors, their majors and their careers.

The results show that the level of participation was not a measure of the level of interaction between participants and a low level of interaction did not necessarily imply the ineffectiveness of the program. Therefore, any evaluation criteria must include the impact on mentees and mentors. Measurement of impact needs to consider various aspects of the envisaged benefits of e-mentoring on mentors and mentees, depending on the motivation behind the program.

4.5 Summary

This study was the first systematic investigation of e-mentoring in Saudi Arabia. The program aimed to encourage high school girls in Saudi Arabia to pursue STEM higher education studies and raise their awareness of STEM career opportunities. The objective was to deploy an e-mentoring program in a structured configuration, where mentors and mentees were matched and regularly contacted each other. The findings from this study show low levels of engagement from both mentors and mentees. The results of this pilot study indicate that the structured e-mentoring configuration had a degree of effectiveness, but was not sufficiently engaging. In particular, the qualitative data from the interviews and the focus group deepened my understanding of the expectations of participants about e-mentoring. Insights from the findings, especially the focus group with mentees, imply the need to further investigate the

requirements of e-mentoring in terms of the relationship, commitment and opportunities for alternative and more flexible forms of e-mentoring.

Chapter 5 Co-Design Workshops

5.1 Introduction

In Chapter 4, the results from Study 1 (the pilot study) indicated that the structured Edmodo e-mentoring configuration was effective to some degree, but not engaging enough. Insights from interviews and a focus group with participants in the Saudi context gave direction to further investigation of the requirements of e-mentoring in this context. To be specific, the focus group with the participants in Study 1 showed interesting findings about how the participants described their e-mentoring relationships. They wanted more freedom and flexibility in their e-mentoring relationships, which contradicts the approaches of previous ementoring literature (Bierema and Merriam, 2002; Single and Muller 2001) where they encouraged frequent and committed meetings between a mentor and a mentee and long-term relationships. These findings made me take a step back and conduct two co-design studies (Studies 2 & 3) with two different age groups of girls to develop a better understanding of the requirements of e-mentoring (Table 5-1).

Table 5-1 Summary of the two studies showing the numbers of participants, ages of
participants and purpose of each study.

Study	Number of participants	Age of participants	Purpose of the study
Study 2	30	18–22	Understanding the requirements for e- mentoring through workshops (brainstorming, bodystorming and prototyping)
Study 3	15	17–18	Validating through designing an e-mentoring product by participants

Design practice has been influenced by changes in the perspective of user-centred design research. In co-design, the people who will benefit from the design (the users) play a large role in gathering knowledge and generating ideas; they play the role of the 'expert on their experience'. User participation in technology co-design aims to elicit knowledge and ideas from expert users who know the problem and have been in the relevant situation (Vines et al., 2013). In addition, co-design gives power and agency to those who are usually ignored in the process (Vines et al., 2013). Giving people power and agency to participate in the design of systems they will use results in successful participation in design by incorporating their understanding of the challenges and barriers. As discussed earlier (in Chapter 2), the target group of this research is generally considered to be an understudied population; based

on the literature review conducted in this research, no studies have been identified, especially in the Saudi context, on employing co-design or user participation in the e-mentoring context. Therefore, the employment of co-design workshops was intended to empower the target population in addition to gaining a deeper understanding of the perceived barriers to and opportunities for e-mentoring from their point of view. This chapter addresses the research question of identifying the factors that contribute to the design of a STEM e-mentoring system for teenagers and young women in the Saudi context.

This chapter describes the design of Study 2, which consisted of three co-design workshops over a period of 3 weeks with first-year female undergraduates to understand the participants' expectations of e-mentoring. Each workshop is presented in detail, then followed by description of the data collected from each workshop. Then Study 3 is described, which was a one-session workshop with schoolgirls (aged 17–18), followed by description of the data generated from that workshop. The following section reports the thematic analysis of the discussions and presentations from both studies (Study 2 and Study 3). Finally, a reflection on the findings is discussed which indicates the need to move from structured e-mentoring models towards other approaches.

5.2 Method: Study 2 – Co-design with Undergraduate Female Students

The lack of understanding of mentoring, the impact of the program on the mentees despite the low levels of interaction and the overestimated effect of the cultural context were the findings of the structured Edmodo e-mentoring deployment (Study 1) and the subsequent interviews and focus group. To build on these findings, three two-hour workshops were conducted once a week over a period of three weeks with female first-year undergraduate students, with the aim of gaining a deeper understanding of the perceived barriers to, and opportunities for, e-mentoring.

The three workshops were conducted with the same group of participants. Different design methods were used in each of the three workshops: problem statement stories, brainstorming, bodystorming and prototyping. The second and third workshops responded to the outcomes of the preceding ones. All workshops were conducted in Arabic, audio-recorded and photographed. The audio recordings were transcribed, then translated from Arabic into English. Full ethical review was conducted and approved by the Faculty of Science, Agriculture and Engineering at Newcastle University.

5.2.1 Participants

In this study, 30 female first-year undergraduate students (aged 18–20 years) were recruited

from a community college. In Saudi Arabia, a community college is an educational institute that grants diploma certificates (2 years) and bachelor certificates (4 years) in applied studies such as databases, programming and computer networks. Most students studying at community colleges come from low socioeconomic backgrounds.

At the end of the first workshop, participants were asked to complete an online survey about themselves and their families, which included demographic data (their ages, plus the ages, education levels and professions of their parents) and their access to female role models. A key consideration in the choice of first-year undergraduates was that they had had the experience of choosing a STEM field without access to mentoring; in this respect, they were identified as experts through experience.

Participants are identified in the following manner: the number of the study combined with the number of the participant. For example, participant number 3 from Study 2 is identified as S2P3. The participants are numbered based on the order of their quotes.

5.2.2 Workshop 1: Problem statement stories

In the first workshop, the problem space was introduced to the participants using a oneminute story (animated video) that I produced specifically for the workshop. Gruen et al. (2002) named stories in the research phase of a design process "problem statement stories". Stories are tools used to encourage others to share relevant or similar situations from their own experience. Most people understand and empathise with stories that have detailed characters, settings, goals and challenges. Empathising with stories prompts innovation and creativity (Gruen et al., 2002).

The non-narrative video had brief English captions in some scenes (Figure 5-1, Figure 5-2, Figure 5-3). It described a story of a 16-year-old Saudi girl who was passionate about space science and wanted to study and have a career related to space (Figure 5-1). Her father was an accountant and her mother was a teacher; they did not have the background needed to help or guide her (Figure 5-2), so she did not know who to ask or where to get advice about how to further her ambition (Figure 5-3). In order not to overly influence the participants, the term 'mentoring' was not mentioned in the video or at any time during the workshop. After the video ended, they were asked if they would like to watch the video again or if there were any English words that they did not understand.



Figure 5-1 Screenshots from the video presenting the main character, Sara



Figure 5-2 Screenshots expressing Sara's goal and challenges



Figure 5-3 Screenshots displaying the setting of the story (Sara's love of space and her family background)

The participants were asked to define the challenge and the factors that contributed to the problem from the story. Participants were divided into five groups and discussions around the story and the challenges faced by the female character, Sara, were held within each group. Each group presented to the rest of the participants what they understood from the story and what they felt were the main problems. Then each group brainstormed ideas to solve the problem and organised their ideas on paper boards using markers, printed icons and sticky notes. The printed icons were a group of random icons that represent mostly technology and connectivity icons (Figure 5-4). Each group received a random set of different icons. Although using the icons was optional, the aim was to help them visualise and present their ideas. All five posters were hung on the wall. Each group presented their idea to the other groups.





5.2.3 Data from Workshop 1

After viewing the animation video, one girl from each group presented their own view of the problem presented:

S2P1 from Group#1: She wants to be an astronaut, she looked and searched, but she didn't know how.*

S2P2 from Group#2: She wants to be an astronaut, but our society does not support such a profession. She might also have financial problems. It was clear that she tried to reach out to others, but no one helped her.*

S2P3 from Group#3: She needs someone to guide her to fulfil her dream.*

S2P4 from Group#4: Her parents are busy and therefore she did not receive the support she needed. She needs support from others outside her family members.*

S2P5 from Group#5: She has trouble managing, managing how to apply, managing her time, managing the pressure. She needs help.*

Each group brainstormed solutions to resolve the problem presented in the workshop.

They showed their ideas on posters; all groups hung up their posters and presented their ideas to the other groups.

Group#1

The application is called *Wasil* (which means 'connection' in English) (Figure 5-5). The application has two users, *Khebra* (an expert) and *Mustafeed* (a beneficiary), and it aims to connect experts and beneficiaries. The expert signs up and provides her information, expertise and field. The beneficiary signs up and looks for an expert in a certain field. The app provides communication between the expert and beneficiary via video/audio call, chat or messages. The app shows job advertisements and courses related to the field. A group of beneficiaries can connect with one expert. The group of beneficiaries support each other and share knowledge about the field they are interested in.



Figure 5-5 Poster by Group#1

Group#2

This group presented a flow chart of events about how the girl will solve the problem (Figure 5-6). First, she will look into books and YouTube to find some information. She will use Twitter to create a hashtag to be connected with other girls who are facing the same problem. This group of girls will create campaigns and leaflets to educate and raise awareness about the field. Then they will come up with an app idea. The app will help anyone who wants to learn more about the field. They can sign in and connect directly with professionals available. One of the features of the app is that it supports video/audio calls. Professionals provide their information and information about their field and workplace, and how to become involved in this field.



Figure 5-6 Poster by Group#2

Group#3

Their solution was an app that is supported by a network of experts (Figure 5-7). The app focuses on creating a connection. Members will use social media to increase the visibility of their app and people will become aware of it. The app will support girls from all around Saudi Arabia. Different fields will be presented and, for each field, it will show what opportunities you have if you study this field. Questions and problems will be sent through the app's email and will be answered and solved by experts. Answers will be provided in a short period of time.



Figure 5-7 Poster by Group#3

Group#4

This group's app is called *Help me* and a web version is available as well (Figure 5-8). There are sections in the app; one section is for books where you find books related to fields you want to know about. Another section is for trusted people, who provide verified information about themselves. A girl can connect with them through chat. She can create a group with experts and others to exchange ideas and knowledge. A user can share an image or a video with other users of the app and share a link about the app or content of the app on Twitter via a share icon, so people know about the app and share knowledge. Another section is gaming; users choose a field, for example 'biology', and they will be asked questions that test their level of knowledge about biology. It's fun and they will learn new things. If an expert posts a video or an article, the users will be notified. You can connect with an expert via the contact information they provide (email, phone ...) or you can use the chat. There will be a section for job announcements.





Group#5

Their idea was an application called *Mehan* ('professions') (Figure 5-9). At the start, a new user will play a game to test their preferences and knowledge about the field and learn new information. The game has levels and the user will become more knowledgeable at each level. A user can connect with professionals. The app provides financial support to help girls study the field of their choice. Users can advertise themselves to obtain job offers. The app is easy to use, supports all age groups and communications between a girl and an expert are via audio or video call.



Figure 5-9 Poster by Group#5

5.2.4 Online survey

At the end of the first workshop, a link to the online survey was shared with the participants and they were asked to use their devices (laptops or phones) to access the survey. All 30 participants agreed to answer the survey. The first part of the survey contained demographic questions about their families and the second part was about their access to female role models. The online survey was in Arabic.

5.2.5 Workshop 2: Bodystorming

In the second workshop, participants were introduced to the notion of mentoring and how their ideas, generated in the first workshop, related to the concept of mentoring and connecting with professional women.

As the concept of mentoring is not a familiar one in Saudi Arabia, bodystorming was used as a method to make the idea of mentoring more tangible. The role of acting in bodystorming supports empathy towards users and helps in becoming familiar with unfamiliar situations, which results in better understanding of the problem (Oulasvirta et al., 2003). Rather than using familiar alternatives such as personas, participants were asked to play the roles of the mentor and the mentee, to support their understanding of e-mentoring and the production of ideas for realising it (Schleicher et al., 2010).

A number of different situations were acted out by two of the participants, S2P6 as a mentor and S2P7 as a girl (mentee); each had a label on her back stating that she was the mentor or the mentee (Figure 5-10). All the scenarios were acted out by the same pair while the rest of the participants were observing, identifying problems and suggesting solutions/actions. The first scene was the starting point based on the results of the previous

workshop, which indicated contacting a professional as the solution to the problem. The following scenes were based on the suggestions and discussions of the observing group members. The first scene was that of a busy professional mentor trying to be contacted by a mentee. Each time the mentee tried to contact the mentor, she could not. Group members initially suggested scheduling meetings where the mentor displayed her availability and the mentee chose a date and time slot. In the simulated meeting (the second scene), the mentor and the mentee sat silent for extended periods of time, leading the group to conclude that simply agreeing on a day and time was not enough. Agendas for such meetings should be established beforehand to include the topics to be discussed. Another scene that was acted out was that, after 15 minutes of the scheduled meeting, the mentee had nothing to ask or discuss; this led the group to discuss that the time slot length should not be fixed and the length of the session should be an option to determine beforehand (Figure 5-10). The final scene was suggested by the girl who acted as a mentee. She suggested that the mentee skipped the scheduled meeting and the mentor waited for her.





5.2.6 Data from Workshop 2

The acting out of the scenarios resulted in the emergence of topics regarding mentoring that were raised by the group of participants and documented on a board in the same room (Figure 5-11). Some of the topics were raised by the group members and some were directed by me based on my previous knowledge about the nature of mentoring.



Figure 5-11 Ideas discussed through the bodystorming workshop

Scheduling with a mentor

Participants discussed that because mentors are usually busy, they should be the ones who display their schedules and availability. In meeting invitations, the mentee would specify the topics that they wanted to discuss, choose the mode of communication based on options the mentor specified and set the length of the session (suggestions ranged over 15–60 minutes). The participants emphasised that the first meeting should not be long, as the mentee might not be comfortable with the mentor. The participants thought that there was a chance that any two individuals (strangers) might not be comfortable talking to each other; therefore, to reduce awkwardness or discomfort, they suggested that the first meeting should not be long.

The final scene of the mentee not showing up to the meeting triggered a discussion on how to manage this situation. One problem was that scheduling and not attending the meeting would lower the chances for other mentees who were more committed to benefit from the mentor. The other problem was that it would be a waste of the mentor's time. One suggestion was to have a waitlist for each mentor; if the mentee with the original appointment did not show up, the mentee with the next waitlist appointment would have her turn. The participants discussed penalties for not showing up to prevent it from being a habit for mentees. If a mentee did not show up for three meetings, the penalties could be: a) she would be banned from scheduling a meeting for some time (for example, 1 month); b) she would be allowed to schedule meetings on the waitlist; or c) she would be charged a fee. The mentor was entitled to excuse a mentee in case the mentee apologised for her absence and it would not be credited as a "no show" on the mentee's record. For each meeting, a mentor would mark the attendance of the mentee with present, absent or apologised for not showing up (excused). Another suggestion was to charge a small fee for scheduling a meeting (Figure 5-12).

Schedulin with ment

Figure 5-12 Discussion points around scheduling with a mentor

One mentor vs. many mentors

It was not clear how the participants were visualising the mentoring relationship, so I asked whether a mentee would be connected to one mentor or many mentors. There were different opinions, but they agreed that it would be optional for the mentee to be connected with one or many mentors. The group of participants who were favouring being connected with one mentor said that they would want to keep their mentor updated about their progress and that they would contact her if they needed help because she (the mentor) had previous knowledge about her (the mentee). They stressed that this one person (the mentor) would give them the moral support they needed. Discussion around being comfortable with the mentor (who was a stranger) was raised again. They explained that sometimes you do not feel connected with someone and they did not want to be committed to a person they did not feel comfortable with. The participants said that if they liked the mentor and would like to keep in contact with her, they would schedule another meeting with her or subscribe to weekly or monthly meetings with her (Figure 5-13).



Figure 5-13 Discussion around being connected to one mentor vs. more than one mentor

The other group, who favoured being connected to many mentors, explained that they would benefit more from connecting with more people. They would be aware of more career options, have different opinions and answers to their questions and learn new ideas and perspectives.

Live sessions

During the discussion around the duration of each mentoring meeting, participant S2P8 suggested "live sessions" for longer periods of time and shorter sessions for one-to-one meetings; she explained: "live sessions like the ones we view on Instagram" (Figure 5-14). The group liked the idea and started stating the benefits of live sessions. One girl said that if she did not like what the mentor was saying or the way she thought, she could easily disconnect from the session without disturbing the mentor. Another girl said that it would help shy girls who found it hard to ask questions. It would also benefit the group of girls attending because one mentee might ask a question that others had not thought about.



Figure 5-14 Live session suggestions

Mobile app vs. web platform

Another confusion that needed to be cleared up was that there was a sense that participants were referring implicitly to the use of mobile apps (Figure 5-15). I asked them directly if this was what they preferred. The participants were divided into two groups, one that preferred mobile apps and the other that preferred web platforms. The first group described the benefits of using mobile apps: getting notifications and reminders about meetings or live sessions on their phone and ease of access as they carried their phones everywhere with them. The other group started by stating the disadvantages of developing the idea as a mobile app. They said that mobile apps consume phone memory space and cause it to lag. They preferred websites or web platforms because PCs and laptops had wider screens in comparison to mobile phone screens. They also mentioned that a web version would be accessible from both a PC/laptop and a mobile phone.



Figure 5-15 Mobile apps vs. web platforms

Appreciating mentors

The discussions were mainly about the mentees and the process. To get a better understanding of how the participants were thinking, I asked them to think about the mentors and why they would volunteer and continue mentoring. Some participants suggested that feedback was a good way to encourage mentors. This feedback could be in the form of ratings of multiple questions after each session; for example, one question could be: On a scale from 1–10 how helpful was the mentor? Some participants also suggested optional comment boxes. Participant S2P2 said that girls her age usually do not write comments. In the case of negative or bullying comments, the mentor could delete the comments and they would not be visible on her page. The administrators would have access to all comments, even deleted ones,

because if one mentor was getting a lot of negative comments it might be an indicator that the mentor was not good with the mentees. The mentor with the highest evaluation would be "mentor of the month". Some participants suggested that mentors should get financial support or be honoured by the government or any other organisation.

Another suggestion was to hold an annual meeting with the mentors and mentees to honour the mentors and network with their mentees. Participant S2P9 suggested that a mentee should share her achievements that were accomplished with the help of her mentors on her profile or social media; for example, "I applied for computing school with the help of mentorA and mentorB". This would be a way to appreciate her mentors. Another way was to allow mentees to invite potential mentors to join the program (Figure 5-16).



Figure 5-16 Encouraging professionals into mentoring

Sharing with others

The idea of sharing their achievements in appreciation of the help of the mentor inspired the participants with the idea of sharing with others to benefit them and share knowledge. Mentees would share their mentor's profile via WhatsApp, Twitter or any social media platform. If a mentor was asked a written question that she found frequent among mentees, the mentor could reply and publish the answer so that all the girls would be able to read her answer. The most frequent questions would be publicly displayed on the mentor's profile page, which might answer some questions without needing to schedule a session. Another suggestion to maximise the benefit was the idea of groups, where a group of mentees joined one mentor (Figure 5-17).


Figure 5-17 Discussion points on sharing with others

5.2.7 Workshop 3: Paper prototyping

In the final workshop, the participants engaged in a process of low-fidelity prototyping using papers that had a rectangle printed on them. They were told that theses rectangles could represent a mobile phone screen or a computer screen. In this session, the aim was not to produce detailed user interface designs, but to complete their understanding and validate ideas generated in the previous workshops. Paper prototyping with young users is a preferred method for capturing, understanding and reflecting on the requirements and needs of the young participants (Glasemann et al., 2010).

Half of the participants did not show up to this session for unknown reasons; therefore, the 15 present participants organised themselves into three groups. They were asked to sketch and represent their ideas on the screen templates provided, based on what they learned in the previous workshops.

5.2.8 Data from Workshop 3

Each group presented and displayed their sketches to the other groups. All groups referred to their sketches as mobile apps.

Group#1

These participants sketched the home page of a user. The sketch showed a space for the time line, where they got updates and notifications. It displayed a profile image, the number of following and followers, and icons for editing the profile and logging out (Figure 5-18, top). The sketch showed that fields such as computer science, science, chemistry and law were in different sections (Figure 5-18, bottom). In each section, a user could schedule a meeting by selecting a date and time. The bottom navigation bar showed a profile icon, a message icon for private message between a mentor and mentee, a calendar that showed scheduled meetings and a search icon to search for mentors.





Figure 5-18 Annotated sketch of Group#1

Group#2

Their page was divided into three sections. The section on the left contained (starting from the top): 'my account', 'fields' and 'appointments'. The participants used orange and yellow sticky notes to demonstrate that selecting the meeting icon displayed the available times for the mentor who the user was following and the live session notifications (Figure 5-19, bottom). At the top of the middle section, the round icons represented the live sessions in action for the mentors. Each round icon had the mentor's name, for example mentor Sara. The mentees who attended a live session could post questions which would be shown at the bottom of the middle section. The right section showed a 'message' icon with a notification for a new message, a 'pioneers and volunteers' icon where a user could search for a professional, an 'advertisements' icon for courses and jobs, and a 'logout' icon.



Figure 5-19 Annotated sketch of Group#2

Group#3

In Figure 5-20, the top sketch shows the main page. The top navigation bar consisted of (from the left): 'settings', 'notifications' and a search bar. In the middle, it showed a profile image and the number of following and followers. At the bottom of the page, there was a 'messages' icon for private messages. Under 'settings' there was a 'time' icon; when selected it displayed the bottom sketch (Figure 5-20, bottom). The 'time' page showed several fields (starting from the left): name of the mentor, day, time, date, presentation and attendance. The 'presentation' field specified whether the presentation by the mentor was a 'lecture' or a 'course'. A 'live' icon was at the bottom left corner for live sessions.





Figure 5-20 Annotated sketches of Group#3

5.3 Method: Study 3 – Co-design with High School Female Students

This study aimed to reflect on and extend the understanding of e-mentoring arising from both the structured Edmodo e-mentoring deployment (Study 1) and the co-design workshops with undergraduates (Study 2) and to this end, a co-design activity with female high school students was conducted. As participants from the previous workshops had demonstrated their understanding of and ability to design and set guidelines for e-mentoring, I aimed to explore the possibility and the capacity of the younger targeted population (female high school students) to design their own approach to e-mentoring. The workshop also aimed to validate the findings of the co-design workshops with undergraduates with the targeted population of this research, who were approaching the point of having to choose a STEM field (i.e. the same age group as in the structured Edmodo e-mentoring study). Participants were asked to complete the same online survey given in Study 2 at home (with the help and knowledge of their parents). The session was audio-recorded and photographed. Translation from Arabic into English was applied when needed. Full ethical review was conducted and approved by the Faculty of Science, Agriculture and Engineering at Newcastle University.

5.3.1 Participants

Fifteen schoolgirls (17–18 years old) were recruited from a private school. Private schools in Saudi Arabia are non-government, have high standards of education and require tuition fees for enrolment. Participants from this school came from higher socioeconomic backgrounds than the participants in Study 2. After conducting Study 2, I recognized that all the participants were from low socioeconomic backgrounds. In Study 1, the need for mentoring was not limited to mentees with parents who had low levels of education but also extended to those with parents who specialised in STEM or were highly educated. Therefore, participants in Study 2 and Study 3 were from different socioeconomic backgrounds to examine if there are differences in requirements and designs between the two groups.

The co-design workshop was conducted at the school campus with the approval, consent and collaboration of the administration of the school and the parents of the students. Participants are identified in the same manner as in Study 2. In this study, participants are identified by S3 followed by the number of the participants (e.g. S3P4). The participants are numbered based on the order of their quotes.

5.3.2 Workshop: Design the box

A two-hour co-design session was conducted at the school campus. The stated goal of the workshop was to design an e-mentoring product that realised the type of e-mentoring experience that the participants wanted.

The purpose of this activity was to translate participants' ideas and needs onto a physical object, 'the box'. The previous workshops, which allowed the participants to express their ideas physically on posters and sketches, gave a better visualisation of what ideas and thoughts they had. The box here was the medium where information was physically presented (Gray et al., 2010). The outcome of this workshop was a box which captured and reflected the needs of the participants (Wienhofen et al., 2014). Wienhofen et al. (2014) described the outcome of this activity as a product that gives prospective users the "I want this and I want it

now" sensation. The activity goes through three phases: an introduction, box design and sharing by "selling" (Gray et al., 2010).

In the introduction phase, the girls were introduced to the overall concept of mentoring through a short oral presentation on the concept of mentoring, how mentoring is applied in different contexts and the different mentoring relationships. Since the purpose of this study was to validate the findings from Study 2, these findings were not introduced to the participants of this study in order not to influence them beforehand. After that, they were divided into three groups. Each group was given time to discuss ideas about how they imagined their mentoring experience. The groups were asked to think about and discuss mentoring from (but not restricted to) different perspectives such as relationship (e.g. one-to-one or one-to-many), level of commitment and mode of communication.

In the second phase, they were set the challenge of designing the box, which depicted an e-mentoring product as if it were a tangible product sold on shelves. Each group was given a cereal box covered with white paper, markers, printed icons (Figure 5-4) and sticky notes. They were asked to start designing their e-mentoring product (Figure 5-21). The product should identify the name of the product, a tagline describing it, key features and requirements.



Figure 5-21 Participants in session designing their products

By the end of the session, each group had created its own e-mentoring product. The final phase was to 'sell' their products. Each group was given the opportunity to present the features of their product, market their product to the other girls and try to convince them to buy it (Figure 5-22).



Figure 5-22 Groups' representatives 'selling' their product

5.3.3 Data from the workshop

The girls designed their products in English and presented them in Arabic. The designs' descriptions and features listed below are extracted from each group's presentation.

Design#1

Name: world.com (Figure 5-23)

Description: A product that allows you to select the person with a major who you want and to ask whatever you want

Features:

- It is an iPhone app
- You search for mentors in different majors
- When you choose a major, the most popular person in that major will be displayed, then you select the person you are interested in
- Different communication channels: text, video, audio
- You can see their location
- You can save chat and video to benefit other people. If you ask a question that they never thought about, they will benefit from your question and answer; this could reduce the number of meetings with mentors
- You can choose whether you want your session to be public for everyone or not
- Every month, we have a group chat with one of our popular professionals

• Unlimited support; you can log in any time, you can connect with any mentor, any major, any number of times

Requirements:

- Age: above 17
- Create an account



Figure 5-23 world.com design

Design#2

Name: Mentor Me (Figure 5-24)

Tagline: Meet your future

Features:

- It is a mobile application
- Each mentor has a profile
- Communication methods: text, audio, video or face to face
- Our app is developed by an organisation that has a centre where the mentor and mentee can meet up
- You can have as many mentors as you want
- Ratings are available for both mentors and mentees; this will help mentors to accept your request or not and will help girls to choose the most helpful mentors
- Each girl has a progress page including who she has talked to and for how many sessions, what achievements she accomplished and which university or major she enrolled in; this will be public for everyone to see.
- Each mentor can set up a group meeting with many girls; live stream
- The first three months are free of charge, then a fee is set for each session.



Figure 5-24 Mentor Me design

Design#3

Name: Know Your Major (Figure 5-25)

Tagline: Brighten your future

Features:

- An app with one-time payment of \$20
- Scheduling meetings
- Communication method: chat
- No restriction on time; you can log in at any time
- All meetings are one to one
- Live sessions are one mentor with many girls
- Gain points by interacting with mentors and you will be rewarded with gift vouchers



Figure 5-25 Know Your Major design

5.3.4 Online survey

At the end of the workshop, participants were asked to complete an online survey at home (with the help and knowledge of their parents), which was the same survey as used in Study 2. It included demographic data (their ages, plus the ages, education levels and professions of their parents) and their access to female role models.

5.4 Findings and Analysis of Study 2 and Study 3

5.4.1 Online survey

Table 5-2 presents the demographics of the parents from Study 2 and Study 3. For Study 2, the table shows that 21 out of 30 mothers' highest level of education was high school, while 4 were illiterate. Even though 5 mothers had a bachelor's degree, none had proceeded further. Comparing fathers' and mothers' education, it is obvious that the fathers had higher education levels. Looking at the parents' field of study and job, of the 5 mothers who had a bachelor's degree only one had studied a STEM field (science) and she worked as a teacher. On the other hand, 11 fathers had at least a bachelor's degree and 7 had studied a STEM field: science (3), technology (2) or engineering (2), and one had a PhD in computer science. Out of 30 participants in Study 2, 13 answered "no" to a question about their siblings: "Have all your siblings who are above 18 obtained or are they currently studying a bachelor's degree?".

		Study 2		Study 3	
		Fathers	Mothers	Fathers	Mothers
Age					
	30-45	5	10	_	4
	46-60	18	19	6	6
	61 or older	4	1	4	-
	deceased	3	_	_	_
Level of e	ducation				
	Illiterate	3	4	_	-
	High school	16	21	—	1
	Bachelor's degree	9	5	5	7
	Master's degree	1	_	4	2
	Doctoral degree	1	_	1	-
Major					
	Science	3	1	_	1
	Technology	2	_	_	-
	Engineering	3	_	2	-
	Math	_	_	_	-
	Other	22	29	8	9

Table 5-2 Demographics of parents from Study 2 and Study 3

In comparison, 10 out of the 15 participants completed the survey in Study 3 (Table 5-2) and the results show that participants in this study were of a higher socioeconomic status and level of parental educational background, with almost all parents having at least a bachelor's degree (except for one who had a high school certificate). One mother had a science degree and was a teacher in a school, while 2 fathers had degrees in engineering. All participants answered "yes" to the question about their siblings: "Have all your siblings who are above 18 obtained or are they currently studying a bachelor's degree?" except for one participant.

Most of the participants (80% of Study 2 and 60% of Study 3) expressed the need for help or guidance regarding their studies and career choices (Figure 5-26) and more than half of the participants (56% of Study 2 and 60% of Study 3) indicated that they did not have a female role model in their life (Figure 5-27). More than half of the participants (57% of Study 2 and 60% of Study 3) indicated that they did not have a clear understanding of career options and opportunities related to their major (Figure 5-28).



Figure 5-26 Answers to the question: "Do you sometimes feel the need to talk to someone regarding your study or future career?"



Figure 5-27 Answers to the question: "Do you have a female role model in your life?"



Figure 5-28 Answers to the question: "Do you have a clear vision and knowledge about career options available in your field of study?"

5.4.2 Thematic analysis of discussions and presentations

The analysis of the data gathered from the discussions and presentations in Studies 2 and 3 identified similarities in the findings of the two studies. Using the inductive approach to thematic analysis (Braun and Clarke, 2006), which is a data-driven methodology where themes originate from the content of the data, I examined the data from Study 2 and identified a number of themes: *the need for mentoring, structuring e-mentoring, independence & loose connection* and *references to daily used applications*. The themes were then discussed with my supervisors.

For the data from Study 3, I applied deductive thematic analysis (Braun and Clarke, 2006), which is analyst-driven. Study 3 was conducted to validate the findings from Study 2,

so the themes emerging from Study 2 were the basis of the deductive thematic analysis. Throughout the analysis, I was open to the emergence of new themes, but none were found.

Need for mentoring

Participants from both studies who came from different socioeconomic backgrounds emphasised their need for mentoring. In Study 2, the groups in the first workshop (problem statement stories) generally agreed that the main problem faced by the girl in the video was that she needed both informational and moral support, guidance or even a means to establish whether her dream of a career in space science for a woman was possible in Saudi Arabia. The solution that most groups proposed was the connection of young women with professional women in Saudi Arabia through email, messages, audio, video or text chat. Other ideas included the provision of resources, reading materials, training, job advertisements and informational broadcasts about fields and careers by professionals.

Participants in Study 3, coming from a higher socioeconomic background, did not show less interest or need for mentoring. In Study 3, participants during their "selling" presentation were keen to exemplify how their products were important for students their age graduating from school as they would provide continuous support by having a pool of professionals who would provide mentoring to help them plan their futures.

Structuring e-mentoring

Scheduling with mentors

During the two studies, participants set guidelines and built a structure for e-mentoring. They discussed methods of interaction, duration, attachment levels and guidelines to maintain the seriousness of the system such as penalties, ratings and recognition of mentors.

In the bodystorming workshop (Study 2), participants discussed that mentors should be the ones who displayed their schedules and availability, as they would have busy schedules. In meeting invitations, mentees would have to propose the topics that they wanted to discuss, choose the mode of communication based on options the mentor specified and set the length of the session (suggestions ranged over 15–60 minutes).

In Study 3, participants described their method of interaction and how mentees and mentors could communicate in different ways. S3P1 from Design#1 described her design by saying:

When you choose a major, the popular person in that major will be displayed. Then you select the person whom you have interest in. You can also select the way you want to chat with them (text, video, audio).*

S3P2 from Design#2 explained:

All majors will be displayed and then you will choose any major you want. All professionals in that major will appear with a profile about her. Then you choose the meeting type text, audio, video, or face to face.*

Live sessions

Another feature that was common between the designs was the ability to broadcast mentoring sessions to members of the community in what they called "live sessions". In Study 2, live mentoring sessions were suggested in some form by the participants. In the bodystorming workshop, live sessions were thought of as a way to help shy attendees who would find it hard to ask questions. It would also give them the ability to easily disconnect from the session without embarrassment; as S2P9 remarked: "If I did not like what she says or the way she thinks, I can disconnect from the session without disturbing her"*. Questions and discussions that occurred in a live session might enlighten them with thoughts and questions that they had not thought of earlier. In the prototyping workshop, Group#2 and Group#3 illustrated live session icons in their sketches (Figure 5-19, Figure 5-20), which they described as a session set up by a mentor and broadcast to her "followers" (i.e. mentees). S2P10 mentioned during her prototype presentation: "The notifications section over here shows all notifications for scheduled meetings and live sessions by mentors who you follow"*.

Additionally, in Study 3 the designs also included a feature that allowed a mentor to host a group mentoring session (occasionally). S3P2 explained: "Mentors can set up a group meeting with many users like in Instagram's Live feature"* and S3P1 (from a different group) remarked: "Every month, a group chat is hosted by one of our popular professionals"*.

Sharing

Sharing on social media, sharing achievements and sharing sessions were topics raised by participants in both studies. In the first workshop of Study 2, Group#2 and Group#3 talked about how to use social media to share knowledge and gain visibility for their app. In the bodystorming workshop of Study 2, sharing achievements by mentees on social media would help in appreciating mentors and sharing the benefits of mentoring. There was a widely held belief that sharing sessions could reduce the number of meetings with mentors, because many discussions and questions would then be available for others to browse and benefit from. A

mentor could share the most frequent questions publicly on her profile page, which might answer some questions without the need to schedule a session:

I could broadcast about the mentor through my social media accounts to my friends.* – S2P11 $\,$

When I meet a friend who faces the same issues I face, I could tell her how the program helped me and that she can look for her answers there.* - S2P12

In Study 3, Design#1 (world.com) included the ability to save your chat or video session with your mentor. They believed that this would help other members and reduce the number of meetings scheduled. S3P1 explained:

You can save chat and video to benefit other people. If you ask a question that they never thought about, they will benefit from your question and answer. This could reduce the number of meetings with mentors. You can choose whether you want your session to be public for everyone or not.*

In Design#2 (Mentor Me), they presented having a progress page for each mentee which would be public to share with other users, to encourage them to schedule mentoring sessions. S3P2 elaborated on sharing:

Each girl has a progress page including who she talked to and for how many sessions, what achievements did she have, which university or major did she enrol in. It will be public for everyone to see.*

Serious engagement

Participants in both studies proposed penalties and solutions to promote "serious" engagement by mentees. In Study 2, for example, in the bodystorming workshop, one proposal that received wide support was that a mentee who did not show up for three meetings would be banned from scheduling a meeting for one month or be able to schedule on a waitlist only. Financial penalties for missing mentoring sessions (with the mentor having the power to waive the penalty) were also proposed, as was a start-up fee to use such a service:

We can charge a small amount for participation to avoid the problem of scheduling meetings and not showing up.* - S2P6

Two groups in Study 3 discussed the importance of the voluntary time contribution made by the mentors and design#2 (Mentor Me) proposed a small fee for each mentoring session:

When you register, the first 3 months are free of charge, then you pay for each session.* – S3P2

Design#3 (Know Your Major) suggested a one-time start-up fee for joining the program. The inclusion of fees was intended to ensure the "seriousness" of the mentees, to encourage them to be more committed to their meetings and to help with the administrative support of the services the girls had designed:

There is a one-time payment of \$20 so you can use the app.* - S3P3

Ratings and reviews

Moreover, there was also significant discussion about how to recognise mentors' contributions in Study 2. Some suggested that mentors could be evaluated after each session, leading to "mentor of the month" status for the highest rated. Additionally, it was suggested that mentees could write about how their mentors had contributed to their achievements and share them on their profiles or social media. Mentees could also invite other professionals to join and become mentors. S2P12 said: "Mentees could invite potential mentors to join the program"*.

Ratings and reviews were also proposed in the designs of Study 3. Design#2 (Mentor Me) included ratings and reviews after each session, with the mentee rating her mentor and vice versa. From the mentors' side, it was suggested that this would help them to decide whether to accept a request for a meeting from a mentee, and from the mentees' perspective, it would help them to choose the most helpful mentors:

After each session, you can rate your mentor and she will rate you as well. This will help other mentors if they would accept your request or not and will help girls to choose the most helpful mentors.* - S3P2

Design#3 (Know Your Major) demonstrated that a mentee got points by interacting with mentors. These points qualified her for a gift voucher:

Girls will have points for interacting with mentors, and those who have more points will get gift vouchers from our sponsor. As a way to motivate girls.* -S3P3

S3P4 from Design#2 after the presentation of the third and final product suggested combining the rating idea from their design with the voucher idea from the third design. She explained: "or when you have good ratings from your mentors, you get a gift voucher"*.

Independent and loose connection

It was common in both studies that participants wanted to initiate the mentoring relationship, that is, they would select their mentors rather than being matched up with them. Also, they were demanding flexible, uncommitted relationships. In Study 2, participants in the

bodystorming workshop stressed the importance of first-time meetings being non-committal and not exceed 15 minutes, and these would be used as the basis for a mentee to decide to enter into a mentoring relationship or not:

Sometimes you do not feel connected with someone and you can know that from the first meeting. You don't want to be committed to a person you don't feel comfortable with.* -S2P4

Where a mentee wanted to progress beyond an initial meeting, another meeting could be scheduled or a subscription established for weekly or monthly meetings. Participants were also clear that they did not want to be limited to one mentor and should be allowed to choose one or more mentors. They stressed the advantages of being in contact with more than one mentor, being connected to more people, learning about new ideas and different perspectives, and getting more than one opinion or answer to a question, and that this would increase knowledge about available careers for women in Saudi Arabia:

I can get more than one opinion/answer for my questions.* - S2P10

I want to know about more careers.* – S2P13

All designs from Study 3 shared the idea of browsing mentors and contacting as many as you wanted. Flexibility and freedom in contacting and communicating with mentors were among the significant features. One group mentioned "not being committed to one mentor"* as one of their special features. Another group described their design by saying: "You can log in any time, you can connect with any mentor, any major, any number of times"*.

References to daily used apps

Familiar concepts such as followers, following, timelines, likes, live sessions, popularity and ratings were common among the two studies. In Study 2, participants in the bodystorming session were referring to social media when talking about "live sessions" and one participant suggested "live sessions like the ones we view on Instagram"*. When the discussion about ratings was raised, one participant explained by saying: "Just like we rate Uber drivers"*. The prototyping workshop in Study 2 led to a number of designs that closely resembled social media apps and platforms, and references to these were made when participants were presenting their designs. For example, when presenting the interface elements of her group's design, one participant said: "These circles over here are for the ongoing live sessions, like the ones we see on Snapchat and Instagram"* (Figure 5-19).

Another group from Study 3 described that they would feature "popular" mentors in live sessions: "Every month, we have a group chat with one of our popular professionals"* (Figure 5-23).

5.5 Discussion

In this work, it was readily apparent from the two studies presented here and the previous pilot study that the participating young women and teenagers were seeking more flexible and less committed e-mentoring relationships than those provided by structured e-mentoring. In this section, I examine these issues further and discuss how the unique traits of the study population (Generation Z) had a major influence on the outcomes of this research. The findings suggest a need to significantly change the format of e-mentoring, by applying co-design activities as a first step in e-mentoring and considering approaches to designing e-mentoring using existing technologies and in ways better fitted to these mentees' lifestyles.

5.5.1 Cultural dimensions of the target age group

The privacy requirements of the female participants in the Saudi context, the selection of the mode of communication and the parental involvement were cultural factors that were initially thought of as requirements for designing e-mentoring for females in Saudi Arabia. However, the participants in this study did not show concern about these cultural factors. The findings in this study appear to be affected less by cultural norms and more affected by the cultural dimensions of the target age group. That is, many of the changes in the mode and format of e-mentoring that the participants identified are likely attributable to the communication preferences and online practices of the of the age group of the participants.

The target age group of this research have not known a world without the internet (Bencsik and Machova, 2016; Garrick et al., 2017), which makes them distinct, as no other generation has had access to the internet in their early years (Bencsik et al., 2016; Garrick et al., 2017). Even though the literature on their generation is relatively limited (Shatto and Erwin, 2017), their identified characteristics (compared to previous generations) include being: smarter, (more) independent, practical, unafraid of change, able to multitask, able to process information more quickly, technologically savvy, always online and dependent on the internet as their main source of information (Bencsik et al., 2016; Igel and Urquhart, 2012; Ivanova and Ivanova, 2009). However, they have also been characterised as impatient, not team players and with short attention spans (Bencsik et al., 2016; Erjongmanee, 2017; Igel and Urquhart, 2012; Ivanova and Ivanova, 2009).

The findings show some correspondence with these traits. For example, one theme that emerged from the analysis of Studies 2 and 3 was *independence*. Participants desired flexible and uncommitted relationships, the ability to connect with more than one mentor and the power to initiate mentoring relationships themselves. On a number of occasions in Study

112

2, during the bodystorming workshop, participants mentioned terms and phrases such as: "15min meetings"* and "not committed to one mentor"*, and during the session in Study 3, "you can connect with any mentor, any major, any number of times"*. In some respects, it could be considered that they did not have the patience required to develop a committed relationship or were too independent to require continuous relationships. It is also possible that participants considered mentors to be a live 'source of information' on the internet. Another possible consideration is that this younger age group of Saudi females were demanding control and power in their society. As discussed earlier in Chapter 1, while gender segregation between men and women in the Saudi culture has weakened the voice of Saudi females, it has been discussed that the online space has constructed a place for Saudi women to strengthen their visibility and make their voice heard (Al Lily, 2011; Al-Saggaf, 2011; Guta and Karolak, 2015).

As the traits of this age group are different, research studies are actively investigating how to address this new generation in different contexts such as schools (Igel and Urquhart, 2012), universities (Ivanova and Ivanova, 2009; Shatto and Erwin, 2017), workplaces (Bencsik et al., 2016; Ozkan and Solmaz, 2015) and the retail industry (Priporas et al., 2017). Likewise, there is a space for redesigning e-mentoring to better meet the needs of this new generation.

5.5.2 Co-design e-mentoring

The findings suggest a need to shift away from traditional structured e-mentoring models and platforms. Previous literature on e-mentoring has insisted that frequent regular meetings between mentors and mentees and the building of a mentoring relationship are key elements in the structure of e-mentoring programs (Bierema and Merriam, 2002; Single and Muller 2001). On the other hand, e-mentoring programs still face the problems of mentee dropout and the early termination of programs (Rhodes et al., 2006; Stoeger et al., 2013; Wallis et al., 2015), including in my own study in Chapter 4 (Study 1). These problems could be related to the lack of development of e-mentoring approaches in comparison to the development of technologies and changes in generations. Another aspect worth considering is that previous e-mentoring literature has dealt with e-mentoring as one approach for diverse populations and contexts.

I argue that the inclusion of mentees in the design process, rather than them joining a pre-structured e-mentoring program, might be one solution to this problem. Participants in this study were able to build a structure for e-mentoring by discussing regular elements in e-mentoring designs: relationship level, duration and means of communication. In addition, they

113

did not overlook elements that have been identified as challenges in maintaining the sustainability of mentoring programs, for example, recognising mentors' voluntary contributions (DuBois et al., 2011) and the recruitment of mentors by youth (Rhodes and Lowe, 2008). Designs in these studies developed a structure for sustainability by appreciating mentors (i.e. "mentor of the month"), mentee recruitment of mentors and setting penalties and solutions to maintain the seriousness of e-mentoring.

The holistic manner of thinking about e-mentoring exhibited by participants in the studies demonstrates a high level of maturity and the ability to build complete e-mentoring designs. Therefore, applying co-design activities in e-mentoring as the first phase of the process would provide those young women and teenagers with agency and power in their own e-mentoring experiences.

5.5.3 Un-platforming E-mentoring

Participant references to social media and audio/video communication technologies were made on many occasions in the studies. Indeed, it is logical and practical to consider how ementoring might reach out and connect to girls in the digital space that they already occupy, rather than imposing a new space or platform on them. Previous literature on e-mentoring programs, which provided various communication methods for contact with mentors, showed that mentees mainly used communication methods that they were familiar with and used frequently (Norodien-Fataar, 2012; Todd et al., 2016). I refer to this process of using existing technologies and social media to realise new and more accessible opportunities for ementoring as un-platforming e-mentoring. Un-platformed e-mentoring creates a space for developing frameworks on top of existing technologies to scaffold/support the recruitment of mentors and mentees. It also permits more familiar processes (for young people) to initiate and build mentoring relationships, to create and distribute content, and to control visibility. Previous work on biculturalism (Vacca, 2017) and cyberbullying (Ashktorab and Vitak, 2016) has presented designs by teenagers that mainly extended current social media platforms or were built on them. Similar to my consideration of un-platforming, Vacca (2017) suggested that the process of co-design should focus on developing plugins and extensions, rather than standalone designs. Ashktorab and Vitak (2016) also suggested that future work should take advantage of existing technologies to implement the proposed solutions designed by teenagers. The references to daily used social media such as Snapchat and Instagram in Study 2 (with young women) and Study 3 (with teenagers) provide evidence that the un-platforming of e-mentoring has wide appeal for this generation.

The notion of un-platforming raises the question of what the trade-offs are between a dedicated e-mentoring platform and the use of existing social networks. While trade-offs in the design of programs have not been discussed in the e-mentoring context (O'Neill et al., 2011), they have been discussed in other contexts such as online learning. Celina et al. (2018) discussed the trade-offs between using a platform-based course and a loosely coupled media course (equivalent to my un-platformed approach) in online learning. While the platform-based course was found to be helpful in centralising and focusing on learning objectives and tasks, the less structured learning environment (using loosely coupled media) was shown to create relationships that lasted beyond the course and promoted more independent learning and problem-solving skills. This meant that loosely coupled approaches worked better for those focusing on building social capital, while the platform-based approach worked better for those focusing on knowledge acquisition. Similarly, e-mentoring should not be thought of as a one-size-fits-all process.

Inevitably, trade-offs between platformed and un-platformed approaches to ementoring are also highly context-dependent in that they vary according to the nature of the participants, the modes and levels of engagement anticipated, and the goals of the ementoring program. For example, designing an e-mentoring program for young people with psychological or health problems who need continuous observation and follow-up meetings is very different from designing STEM e-mentoring for young women and girls. Young people with health problems are unlikely to be comfortable discussing such issues with their mentors using publicly accessible (i.e. un-platformed) social media channels and should generally be discouraged from doing so. They would be more appropriately mentored through a secure and closed digital platform. On the other hand, un-platformed e-mentoring has the potential to positively impact on mentees' levels of interaction and engagement. Given that social media is part of their everyday internet activity, it could be the means by which they build lasting relationships with mentors, as was found by Celina et al. (2018).

These aspects merit further investigation, given that most accounts of e-mentoring programs report challenges such as premature termination of relationships between mentors and mentees (Rhodes et al. 2006), program dropout (Wallis et al., 2015) and inactive mentors (Lenear, 2007). While giving up some centralised control (un-platforming) limits the scope of processes such as content moderation, the utilisation of mainstream services and technologies leverages the high standards of usability, security and privacy that come with systems and implementations which operate on a global scale. Furthermore, un-platformed e-mentoring leverages mentees' and mentors' everyday social media literacy, in terms of their ability to

115

express themselves, their understanding of who has access to messages and posts, and how data generated is stored and used.

5.6 Summary

In this chapter, I have described how I applied co-design methods to gather and capture the requirements of the targeted population. The findings of the two studies were affected less by cultural norms and more by the traits of the participants' generation. The co-design workshops pointed to a need to shift the design of e-mentoring models to incorporate co-design in order to increase engagement and give a voice to young women. The findings indicated that young women and teenagers were seeking mentoring relationships that were more flexible, with less commitment and more integrated with their everyday lifestyles and use of mainstream online technologies. As a result, I proposed the un-platforming of e-mentoring and the development of frameworks that build on existing technologies to facilitate e-mentoring.

Chapter 6 The Design of Qudwa

6.1 Introduction

This chapter introduces the design requirements and system design of *Qudwa*, an unplatformed STEM e-mentoring system designed to facilitate communication between female Saudi professional mentors and young female mentees. The findings of my previous studies indicated that a shift was needed in the approach to providing e-mentoring for young women. This design responds to the findings from Studies 1, 2 and 3. In Chapter 4, the results of Study 1 suggested further investigation into the requirements of e-mentoring for young women and teenagers in the Saudi context. In Chapter 5, the findings of Study 2 and 3 pointed out that the participants (young women and teenagers) were looking for a different form of relationship. In 2002, Bierema and Merriam (2002, p. 223) discussed the future of ementoring: "The possibilities for e-mentoring are as endless as the Internet. The extent to which this medium will be used for mentoring is unknown as are the best ways to maximize the nature of this medium for this purpose".

Qudwa has been designed based on the findings of the studies described in Chapter 4 and Chapter 5 and the lessons learned from previous work on structured e-mentoring. My results indicated that young women and teenagers preferred less committed and more flexible e-mentoring relationships. They also desired to choose their own mentors and to communicate in a way that was more integrated with their daily internet activity. Even though previous work on e-mentoring used social networking platforms as their media, they did not provide participants with control over their relationships. These programs were structured in terms of the researchers choosing the medium of communication, the frequency of contact and the matches beforehand. *Qudwa* aims to empower both mentees and mentors by giving them more freedom in choice and flexibility. This chapter begins by describing the design requirements followed in the system design of *Qudwa*.

6.2 Requirements for Design

From working on the previously described studies (Studies 1, 2 and 3) and with the lessons learned from previous work on e-mentoring, I was able to understand the needs and requirements for facilitating STEM e-mentoring relationships for young women and teenagers. Four design requirements emerged from the conducted work: 1) flexibility and control; 2) visibility; 3) integration of e-mentoring with daily internet activity; and 4) a trusted connection between mentors and mentees (Table 6-1).

Table 6-1 Summary of Qudwa's design requirements

Design requirement	nts
RQ1 Flexibility an	nd control
	RQ1.1 Mentee-initiated relationship
	RQ1.2 Flexibility and control in the e-mentoring relationship
	RQ1.3 Flexibility and control in choosing the communication channel
RQ2 Visibility	
RQ3 Integration of	f e-mentoring with daily internet activity
RQ4 A trusted cor	inection

6.2.1 RQ1 – Flexibility and control

The findings showed that the participants desired more flexibility, more control and less commitment in their relationships. This means that they wanted control in choosing who, how, how many mentors and for how long they would contact their mentors. The results of the conducted work showed that participants described an e-mentoring relationship initiated by them. The findings identified the need for a design that can support e-mentoring without requiring the girls to commit to a mentor or forcing them to contact their mentors regularly. It should allow contacting as many mentors as needed and only when needed. Another aspect that showed their desire for control is that participants requested to be able to recommend professionals to be mentors.

As mentioned earlier, e-mentoring programs have faced the problems of mentee dropout and the early termination of programs (Rhodes et al., 2006; Wallis et al., 2015), including my own study (Study 1). Previous work on e-mentoring has demonstrated how participants made changes in their e-mentoring relationships, whether a change in the means of communication or in the structure of the e-mentoring relationship.

E-mentoring programs that provided more than one communication method showed that mentees mainly used communication methods that they were acquainted with and used frequently (Norodien-Fataar, 2012; Todd et al., 2016). In the study of Norodien-Fataar (2012), the mentors, who were undergraduate students mentoring their peers, were divided into three groups and each group was asked to use a specific method of communication. One group was asked to use an LMS, the second group was asked to use Facebook and the third group was encouraged to use any means of communication. The findings showed that all three groups used Facebook and MXit (which is the largest instant messaging and social network in Africa). The reason that participants used Facebook and MXit was that these platforms were more familiar among mentees; they are widely used and the ability to be on the social network "doing your own thing" and contacting your mentor at the same time was a favourable feature. BreakThru (Todd et al., 2016) was a STEM e-mentoring program for secondary and post-secondary students with disabilities that utilised virtual worlds as a means of communication. The study concluded that the use of virtual worlds was insufficient and participants favoured the use of methods that were faster, more accessible and more familiar to them, such as audio calls, SMS text messaging and email.

In other examples, mentors and mentees changed the structure of how their ementoring relationship was set by the researchers and developed what suited them best (Stewart and Carpenter 2009). Stewart and Carpenter (2009) designed an e-mentoring program for physical therapists in rural areas. The program was set up with daily emails, weekly chat and monthly video conferencing (iChat). Because participants favoured the use of video conferencing, the form of interaction was changed by the participants after two weeks of the program to video conferencing twice weekly and email used irregularly.

Therefore, three sub-requirements were derived:

- RQ1.1 Mentee-initiated relationship
- RQ1.2 Flexibility and control in the e-mentoring relationship
- RQ1.3 Flexibility and control in choosing the communication channel

6.2.2 RQ2 – Visibility

A second design requirement is visibility. Earlier in Chapter 4 I discussed the issue I faced during the recruitment of mentors; it was not because they do not exist, but because of their lack of visibility. The lack of communities and networks for female Saudi professionals made it difficult for me to find them and reach out to them. Findings from the studies showed participants wished to learn about all available professions and opportunities for females in Saudi Arabia by connecting with more than one mentor. Their longing to be connected with many mentors was because they wanted to seize the opportunity of the availability of a group of mentors to learn more from them and get different answers or opinions to the matters they were thinking about.

6.2.3 RQ3 – Integration of e-mentoring with daily internet activity

Another design requirement is the integration of e-mentoring with daily internet activity. In

119

Chapter 5, the participants referenced social media platforms and applications that they used daily in describing how they envisioned their e-mentoring experience. From there the idea of un-platforming e-mentoring and usage of existing platforms has been derived.

Previous work in the literature showed that participants tended to use platforms that they were accustomed to or used regularly. Other articles discussed the use of SNS as a medium for structured e-mentoring (Norodien-Fataar, 2012; Rashid and Rahman, 2014; Ware and Ramos, 2013). Specifically, Facebook was used as an e-mentoring communication method. As mentioned earlier, in the study of Norodien-Fataar (2012), participants used Facebook and MXit because they could contact their mentors/mentees and interact with their social networks. In the study of Ware and Ramos (2013), Facebook was used to provide informational support for first-generation Hispanic college students. They employed Facebook to conduct structured e-mentoring where the mentor posted weekly to the group's wall. The study stated that the benefits of using social media in e-mentoring were the immediacy, constant connectivity and flexibility of such media. Similarly, the findings of Rashid and Rahman (2014) indicated that social media provided an informal, relaxed and flexible atmosphere for structured e-mentoring. All the mentees and mentors were in one Facebook group and the e-mentoring program aimed to enhance the creativity of interior design students. The findings stated the effectiveness of using Facebook as an e-mentoring medium. Mentees felt less stressed and more informal while engaging with their mentors. One mentor explained that students used Facebook constantly for their personal engagements with their friends, which increased their participation in the e-mentoring group. They also stressed the importance of learning the boundaries between personal and professional usage of social media as a medium for e-mentoring. In some cases, the mentees posted comments on their mentor's timeline which caused inconvenience for the mentors.

These examples show how e-mentoring was previously deployed using different social media platforms. While platforms such as Facebook were favoured and frequently used by the participants in these studies, this suggests the use of SNS that are highly used among young women and teenagers in Saudi Arabia.

6.2.4 RQ4 – Trusted connection

The final design requirement is a trusted connection between mentors and mentees. The fact that mentoring, in general, is an unfamiliar concept in Saudi Arabia cannot be ignored and, in some cases, participants were thinking of mentoring as a "lecture" or a "course" session. Therefore, this connection is to ensure that mentoring is conducted as it should be and to clear up any misunderstandings. Participants on different occasions in relation to their designs

referred to the presence of a centre or an organisation. This might relate to their need for safety or security when contacting someone who was a stranger to them (a mentor).

6.3 Qudwa

6.3.1 What is Qudwa

Qudwa is a STEM e-mentoring system for teenagers and young women in Saudi Arabia to be mentored and connected with female Saudi professional mentors. It is designed to give mentees and mentors more flexibility and control in their e-mentoring relationships. *Qudwa* is designed to be embedded in their daily internet activity, instead of pulling them into another space, by using existing social networking platforms. *Qudwa* aims to raise the awareness and visibility of Saudi professional women and their careers, and to encourage girls to engage in different fields of work and to introduce them to the areas of work available to them.

Qudwa is a website built on the concept of the Wizard of Oz (Figure 6-1). The home page presents the pool of mentors in one place. The mentee has the control to search, browse and choose her mentor or mentors. After choosing a mentor, she selects how she would like to be in contact with the mentor. *Qudwa* provides three means of communication: 1) an online meeting via Appear.in; 2) a WhatsApp group led by the mentor; or 3) being introduced to the mentor through social media. All requests submitted by mentees are handled manually by the admin. Then the mentor and the mentee will be in contact. Figure 6-2 describes the mentee's journey in using *Qudwa* and also presents how each step on the user journey meets the design requirements previously found.



Figure 6-1 Home page of Qudwa





Figure 6-2 User journey of Qudwa presenting each step and its design requirements

6.3.2 Meeting the requirements

RQ1 Flexibility and control

Flexibility and control are given to the mentors and mentees on how, when and for how long they want to contact each other. *Qudwa* provides three channels for communication: 1) an online meeting via Appear.in; 2) a WhatsApp group led by the mentor; or 3) being introduced to the mentor through social media.

Each mentor at the recruitment stage will have control over the methods she wants to be contacted with. All three methods are available for all mentors unless a mentor requests to eliminate a method. The mentors will provide *Qudwa* with their social media accounts of the platforms of their choice and their choice for audio calls only or video calls.

On the other hand, the e-mentoring relationship is initiated by the mentees. The mentee will browse and choose her mentor (RQ1.1). The mentee will have the choice as to which mentor to contact. *Qudwa* does not request a specific frequency of contact between the mentor and the mentee, so a mentee can have one session with her mentor or several sessions, based on her needs and preferences (RQ1.2). She also has the freedom to choose which method of communication suits her (RQ1.3). These choices are not fixed. While she might prefer to contact one mentor through an online meeting, she can communicate with another mentor through WhatsApp.

RQ2 Visibility

Qudwa presents all available mentors on the website with their information, education and previous and current workplaces and positions. The pilot study in Chapter 4 showed that displaying mentors' profiles had a positive effect on mentees' awareness of available opportunities for women in Saudi Arabia. In addition, Savoy (2013) in her study stated that the most favoured aspect for participants in her STEM e-mentoring program was knowing and reading the mentors' profiles. Participants indicated that reading the biographies was encouraging and raised their awareness of the opportunities to be a female professional in a STEM field (Savoy, 2013). Therefore, by presenting mentors' profiles *Qudwa* intends to create visibility for the mentors among their peers and among mentees, and raise awareness of current and available opportunities for females in Saudi Arabia.

RQ3 Integration with daily internet activity

According to the General Authority for Statistics (General Authority for Statistics, 2018), 72.54% of people in Saudi Arabia access the internet to participate in social media platforms. Therefore, *Qudwa* facilitates the use of existing social media platforms and the high usage of such platforms to serve e-mentoring. *Qudwa* provides three main communication methods: 1) requesting an online meeting; 2) through WhatsApp; and 3) introduction through social media. Online meetings use audio or video calls depending on the participants' choice. *Qudwa* uses Appear.in (now called Whereby). This platform does not require a sign-in process or sharing usernames, which eases the process of communicating. For each scheduled session, a new link is created and shared with the mentor and mentee. Only those who have the link can join the session. Shared links can be used on mobile or desktop/laptop devices. Because WhatsApp is the most widely used social networking platform in Saudi Arabia (Ministry of Communications and Information Technology, 2016), it was selected as the textbased communication method. WhatsApp can be used for one-to-one or one-to-many communication. The final method gives the mentee the choice of contacting the mentor through other social media platforms (provided by the mentor). If requested, the admin will introduce her to the mentor to ease the communication between them or she can contact her directly.

RQ4 Trusted connection

The trusted connection or human factor is the admin (me). All connection requests sent by mentees are received by the admin. The admin filters the requests and proceeds with them. The unautomated system is based on the Wizard of Oz method, where parts of the system behaviour are simulated by a human 'wizard' – the admin (Salber and Coutaz, 1993). In *Qudwa* the admin manages all connection requests. The Wizard of Oz is an experimental evaluation method (Salber and Coutaz, 1993).

The presence of a human in handling the system is required for evaluation and safeguarding purposes. Since e-mentoring and the system itself are novel in the context, the admin is needed for testing the initial implementation of the system, observing and collecting data. The wizard in this system is also intended to provide safety and safeguarding for the participants. The admin makes sure that connections are made, solves any problems faced by the participants and safeguards them, that is, she prevents mentors from being contacted by fake or fraudulent people who claim to be young women, for example. The admin also makes sure that mentees are connected to the right mentors and are not abused or neglected by the mentors.

6.3.3 Design of Qudwa

Home page

The home page of *Qudwa* shows all the mentors in one page (Figure 6-1). There are two language versions of *Qudwa*: English and Arabic. At the top of the page, there is a search bar and tags to ease the search for mentors. For each mentor on the home page, it shows the mentor's profile picture, brief biography, tags for related subjects and social media accounts (RQ2). Tags are used to ease the search process and to educate the mentees about different fields. At the bottom of the page there are two links, one for helping mentees if needed and one for recommending mentors (RQ1) (Figure 6-3).

other and how this conversation can go wrong in disease. science biology genetics	(technology) (developer) (web)	a current Rheumatology Fellox.	
About Oudwa facilitates the connection between young women and girls and professiona In Outwa, we win to encourage girls to engage in different fields of work and to intr areas of work available to them. About Cook	Lwomen in STEM. oduce them to the ies Ø Privacy Research Terms	Couldn't find what you are looking f	

Figure 6-3 Bottom of the home page

Mentor profile

On the mentor's profile page, a more detailed biography is presented with information about her education and work experience. In addition, each mentor shares topics that she can help with such as job applications and university applications. Mentors share their personal interests. At the top the tags are shown, while at the bottom the social media profiles of that mentor are shared. The ways to connect with a mentor are on the left of the page (RQ1.3 & RQ3) (Figure 6-4). Figure 6-5 shows the methods supported by the system for connecting with a mentor. Each method is explained through a help message (Figure 6-5).

_{فَوْس} َعْظَ Qudwa	King Saud University with bachelor's degree in information Science. I have also a master's degree from Prince Sultan University in Business Administration. I am specialized in IT projects management, robotics process automation, and Data Analytics. I am certified in PMP, TOGAF, CBAP, ITIL, MCPD & MCTS. Education and Experience MBA, Prince Sultan University, 2015
Senior Analytics Engineer, General Electric	 Bachelor Degree, Information Technology <i>King Saud University</i>, 2006 Topics I can help with university & job applications, project management, data & process analytics,
Connect with P	Personal interests sport, reading,

Figure 6-4 Profile page of a mentor



Figure 6-5 Dropdown menu shows how to connect with a mentor

Request form

When a mentee chooses to connect with a mentor, she fills in a form (RQ1.1, RQ1.2, RQ1.3 & RQ3), providing her full name, email address, mobile number, a brief introduction about herself and the topic she wants to talk about with the mentor (Figure 6-6). Each form is slightly different than the others depending on which connection method was selected. If the mentee chooses to be connected through social media, the form asks her to choose which platform she wants to be connected on and her username on that platform (Figure 6-7). With online meetings, the form asks her to specify whether she prefers audio or video calls (Figure 6-8).

	pp group lead by the mentor and get into discussion eek advice and guidance.	Ä
Full Name	Full Name	<u> </u>
Email	Email	قر _{وة} Qudwa
Mobile number	Mobile number	
Date of birth	Day \$ Month \$ Year \$	
Introduce yourself	Introduce yourself	Senior Analytics Engineer, General Electric
Why would you like to contact this mentor?	Why would you like to contact this mentor?	
I understand project and agre	this service is provided by Open Lab as a research e to the <u>Research Policy</u>	
I accept the F	rivacy Policy	
I accept the]	erms of Use	

Figure 6-6 Request form for the WhatsApp group connection method

	her, then fill the form and we will help with that.	()
Full Name	Full Name	قدوة
Email	Email	Qudwa
lobile number	Mobile number	
Date of birth	Day \$ Month \$ Year \$	inche attacce.
Connect through	Snapchat \$	Senior Analytics Engineer, General Electric
/our Snapchat profile	Your Snapchat profile	
Introduce yourself	Introduce yourself	
	10	
Why would you like to contact this mentor?	Why would you like to contact this mentor?	
I understand	this service is provided by Open Lab as a research e to the Research Policy	
	Privacy Policy	

Figure 6-7 Request form for the social network connection method

	Full Name	
Emoil	Email Output	
Email	Ethan	Qudwa
Mobile number	Mobile number	
Date of birth	Day	
Preferred communication method	 Audio coll Video coll 	Senior Analytics Engineer, General Electric
Introduce yourself	Introduce yourself	
Why would you like to contact this mentor?	Why would you like to contact this mentor?	
I understand project and gare	this service is provided by Open Lab as a research e to the <u>Research Policy</u>	

Figure 6-8 Request form for the online meeting connection method
Let us help

This page helps a mentee if she has not found a mentor in a specific field or needs guidance. A form is displayed which is similar to the request forms previously shown but has a section for the field of interest (Figure 6-9).

← Back to main page					
📾 Let us help					
If you looked for a mentor in a spec your request through this form and	:ific field and you didn't find someone that can help you, then send us d we will try to find you a mentor.				
Full Name	Full Name				
Email	Email				
Mobile number	Mobile number				
Date of birth	Day 0 Month 0 Year 0				
What field interests you?	What field interests you?				
Introduce yourself	Introduce yourself				
	li li				
Why would you like to contact a mentor?	Why would you like to contact a mentor?				
different i					
	6				
I understand project and agree	I understand this service is provided by Open Lab as a research project and agree to the <u>Research Policy</u>				
I accept the P	I accept the <u>Privacy Policy</u>				
🔲 l accept the <u>T</u>	erms of Use				
	Submit				
	John				

Figure 6-9 Let us help form

Recommend a mentor

This page is for recommending mentors. Anyone can recommend a female professional to be a mentor on *Qudwa*. The page shows a form to be filled in with the mentor's name and ways to contact her (Figure 6-10).

	nentor in any other platform or in person and you would recommend t Il the form and we will send her an invitation.
Mentor's full name	Mentor's full name
Mentor's email	Mentor's email
Mentor's mobile number	Mentor's mobile number
Mentor's social profiles	Snapchat, Twitter, Linkedin, Instagram, + Add onother profile
I understand project and agree	this service is provided by Open Lab as a research e to the <u>Research Policy</u>
 I accept the P I accept the T 	

Figure 6-10 Recommend a mentor form

Admin page

The admin page is a page for the admin to manage mentors, view submissions and observe *Qudwa*'s visitor analytics (Figure 6-11). The admin can create, edit and delete a mentor's profile. Figure 6-12 shows the fields for creating a new mentor profile.

dwa Editor			9
entors	MENTORS		
bmissions	View and edit mentors		
nalytics	CREATE MENTOR		
	COLL COMPANY	Add Post Edit	Mentor Delete
		Add Post Edit	Mentor Delete
	and musical of	Add Post Edit	Mentor Delete

Figure 6-11 Qudwa editor page

ASEEL	
SHARED CONTENT	
IMAGE URL	
	Upload Image
SOCIAL ACCOUNTS	
(Website 4) Account UR	L
+ADD SOCIAL ACCOUNT	
ENGLISH	ARABIC
NAME	NAME
JOB TITLE	JOB TITLE
WORKPLACE	WORKPLACE
TAGS	TAGS
+ADD TAG	+ADD TAG
SHORT BIO	SHORT BIO
LONG BIO	LONG BIO
EDUCATION AND EXPERIENCE	EDUCATION AND EXPERIENCE
тпе	тпье
PLACE	PLACE
DESCRIPTION	DESCRIPTION
DURATION	DURATION
+ADD EDUCATION AND EXPERIENCE	+ADD EDUCATION AND EXPERIENCE
TOPICS	TOPICS
+ADD TOPIC	+ADD TOPIC
INTERESTS	INTERESTS
+ADD INTEREST	+ADD INTEREST
	Save

Figure 6-12 Creating a mentor by filling in information in English and Arabic

All forms are received on the submissions page for the admin to review and process (RQ4) (Figure 6-13). Information about the mentee and her activity on *Qudwa* is recorded (Figure 6-14).

Qudwa Editor				🔮 Logout
Mentors	SUBMISSIONS			
Submissions	View submitted requests			
Analytics	TYPE: ALL 🔻	MENTOR NAME	FROM	TIME RECEIVED
	Recommend Mentor	100/00-001-01		04/10/2019 07:23pm
	Let Us Help		1000	01/09/2019 10:26am
	Join Whatsapp Group		(1000,000,000,000)	23/08/2019 11:34pm
	Request social media introduction	Sala Missis		24/06/2019 03:02am
	Request social media introduction	has apprecia	-	10/06/2019 06:24pm

Figure 6-13 Submissions page

Qudwa Editor			🦉 Logout
Mentors	SUBMISSION		
Submissions	METADATA		
Analytics	REQUEST TYPE		TIME RECEIVED
	JOIN WHATSAPP GROUP	Test report in the left	21/05/2019 10:41pm
	SENDER INFORMATION		
	NAME	~	
	EMAIL	tere lines and part com	
	MOBILE NUMBER	003030+007	
	DATE OF BIRTH	11/08/2001	
	INTRODUCE YOURSELF	متخرجه من الثانوية احب برمجه الحاسب ونظم المعلومات	
	WHY WOULD YOU LIKE TO CONTACT THIS MENTOR?	لكسب معلومات وخبرات	
	VISITOR INFORMATION		
	FIRST VISIT SESSION (
	10:39PM 21/05/19 1	21	2
	VISITOR PATH		
	View all activity by this visitor		
	10:41PM 21/05 Submitted Form Type: Request Whatsap	p <u>http://qudwa.me/ar/join-whatsapp?ment</u>	<u>or=</u>
	10:41PM 21/05 Visited Request Whatsapp:	http://qudwa.me/ar/join-whatsapp?ment	
	10:41PM 21/05 Visited Mentor:	http://qudwa.me/ar/scone	
	10:41PM 21/05 Submitted Form Type: Request Whatsap	p <u>http://qudwa.me/ar/join-whatsapp?n</u>	

Figure 6-14 A request submitted by one of the mentees

The analytics main page shows an overview of the activity on *Qudwa*, number of visits, number of visitors, top selected tags, top search queries, top visited mentors and top mentors who have been requested to be connected with (Figure 6-15). The analytics also records visitor paths and visitor activity. Visitor activity shows all the activities of users on *Qudwa*. Visitor paths show the individual paths of users and whether they are a returning user or not (Figure 6-16).



Figure 6-15 Analytics page showing an overview of activity on Qudwa

VISITOR	ACTIVITY			
La	st year	Last month	Last Week	<u>Today</u>
15/10/19				
05:56AM 15/10	Visited Mento	or:	http://qudwa.me/en/r	in an
05:55AM 15/10	Visited Home: Index		http://qudwa.me/en/	
14/10/19				
06:14AM 14/10	Visited Request Introduc	tion:	https://qudwa.me/ar/request-introduction	?mentor=t
06:13AM 14/10	Visited Mentor:	The second second	https://qudwa.me/ar/	
06:13AM 14/10	Visited Request Meeting	And age insertails	https://qudwa.me/ar/request-meeting?me	ntor=

VISITOR PATHS				
Last year		<u>Last month</u>	Last Week	<u>Today</u>
#1, 18 RETURNING VISITS	5, FIRST VISIT: 07/04/19			
View all activity by thi	is visitor			
05:56AM 15/10 Visit	ed Mentor:		http://qudwa.me/en/i	
05:55AM 15/10 Visit	Home: Index		http://qudwa.me/en/	
08:51AM 13/10 Visit	Home: Index		http://qudwa.me/en/	
08:06AM 13/10 Visit	Recommend Mentor		http://qudwa.me/en/recommend-mentor	

Figure 6-16 Visitor activity shows all activities of users on *Qudwa* (top) and Visitor paths showing individual paths of users and whether they are a returning user or not (bottom).

6.4 Summary

In this chapter I have introduced *Qudwa*, which was designed and implemented based on four design requirements. The design requirements are: 1) flexibility and control; 2) visibility; 3) integration of e-mentoring with daily internet activity; and 4) a trusted connection between mentors and mentees. These requirements were presented and rationalised based on the findings of the previously conducted studies. I have also presented how each design requirement was met in *Qudwa*. The next chapter will present the deployment and evaluation of *Qudwa*.

Chapter 7 Deployment and Evaluation of Qudwa

7.1 Introduction

Chapter 6 described the four design requirements: 1) flexibility and control; 2) visibility; 3) integration of e-mentoring with daily internet activity; and 4) a trusted connection between mentors and mentees, in the design of *Qudwa*. This chapter describes the fourth and final study, which shows how *Qudwa* was deployed and evaluated. The chapter begins with the study design including the recruitment and methods used to announce *Qudwa*. Data for this study was collected from submissions, *Qudwa* and social media analytics, a mentees' survey and post-study interviews with mentees and mentors. Each communication method used in *Qudwa* is reported individually. After that, the results of the survey and the thematic analysis of the interviews are presented. Finally, the chapter concludes by presenting the achieved design goals of *Qudwa* supported by the findings of the study and design implications of *Qudwa*.

7.2 Study Design

Qudwa was set to run for two and a half months (April, May and into mid-June 2019) during the end of the academic year and the beginning of the summer vacation in Saudi Arabia. *Qudwa*'s main target groups were female high school graduates and female university graduates. Public universities that are funded by the government and provide free education outnumber private universities in Saudi Arabia. All public universities in Saudi Arabia share a unified admission period, which is set during the summer vacation from mid-June until the beginning of July. Private universities open their admissions in the summer vacation as well.

Therefore, the timing was chosen to maximise the benefits of STEM e-mentoring for the mentees because it is a period that was critical for them in choosing a field to study. The time was also beneficial for university graduates who were looking for jobs, to learn about the opportunities open to them as females. *Qudwa* was not limited to the mentioned target groups but was open to any female who needed mentoring in STEM fields related to work or study. Full ethical review was conducted and approved by the Faculty of Science, Agriculture and Engineering at Newcastle University.

7.2.1 Recruitment

Mentors were recruited in the same manner applied in Study 1 (the pilot study). I contacted prospective mentors through my personal contacts and LinkedIn. Mentors who participated in Study 1 were approached as well, but none showed interest in participating further. I looked

for professional women in STEM with a minimum of five years of experience. When approaching a prospective mentor, I described *Qudwa* briefly, and if she showed interest in participating I would send her the research information sheet, the mentor's information form and the consent form. I also shared an online form for joining *Qudwa* as a mentor which was sent through WhatsApp and Twitter. People who signed up to participate in *Qudwa* were filtered based on their field of work and level of experience. Fields were limited to science, technology, engineering and maths with a minimum of three years of experience. Signed-up names who met the criteria were sent the research information sheet, the mentor's information form and the consent form.

For recruiting mentees, participation was open to all young women aged 16–20 years (but not restricted to that age range) in Saudi Arabia, as the aim was to reach mentees from all over the country. Announcements about *Qudwa* were sent through different social media channels. The information sheet for mentees was available online on the *Qudwa* website. Due to the nature of the research, the consent form for the mentees was in an online form.

7.2.2 Announcements

Announcements about *Qudwa* were conducted through different channels. To reach out to the target group, a 40-seconds animated video was designed in Arabic explaining the purpose of *Qudwa* and how to use it. Figure 7-1 shows all six scenes that the video displays.

The video was shared on WhatsApp. People were asked to share the video with their contacts. Mentors were also asked to spread the word about *Qudwa* and share the video on their social media accounts. A Twitter account was created to announce *Qudwa* and to reach a larger audience (Figure 7-2). The Twitter promotion was used to advertise *Qudwa* and promote the animation video. The number of visits was high, but the number of submissions did not exceed 10 requests. Due to low participation, an Instagram account was created (three weeks after the beginning of *Qudwa*) to promote *Qudwa* (Figure 7-2). The promotion on both platforms was set to Saudi Arabia and the target audience was females 16–24 years old. Both promotions cost \$100 each. The deployment of *Qudwa* was at the time of the summer vacation in Saudi Arabia, so visits to schools were not possible.



Figure 7-1 The first scene (top left) presents the target audience, which is high school and university female students. The second (middle left) and third (bottom left) scenes present the problem faced by the students by asking questions: "Do you need guidance from an expert?" "Do you feel confused between study fields and work fields?" The fourth scene (top right) presents the voluntary nature of *Qudwa*. The fifth scene (middle right) explains how to use *Qudwa* by choosing a mentor and how you want to connect with her. The final scene (bottom right) displays how to reach *Qudwa*.



Figure 7-2 Profiles of *Qudwa* on Instagram (left) and Twitter (right)

7.2.3 Wizard of Oz

In this study, I played the role of both 'the admin' and 'the system'. For this study, the term 'admin' refers to the tasks expected to be carried out by a person acting as the administrator and the term 'system' refers to the tasks carried out by me during the study to simulate a possible system. Table 7-1 shows each task and whether it is processed by the admin or the system. When a request is received, the system contacts the submitter through email or WhatsApp to confirm that this is the correct email address and mobile number. In the case of an audio or video meeting, the system contacts the selected mentor, provides her with the information about the mentee and asks her to suggest dates and times that suit her. After that, the system contacts the mentee and asks her to select a convenient time and date based on the mentor's availability. Then the system sends a confirmation email to the mentor and the mentee join the meeting. The admin joins the online meetings as well to listen, monitor and record.

Task	Processed by	Action
Handling Let us help request	admin	The admin contacts the submitter to help her through her problem
Handling Recommend a mentor request	system	An application is sent to the recommended mentor via email
Validating requests	system	An email is sent to the mentee to validate her contact request
Introducing through social media	admin	The admin arranges how the mentee and the mentor will contact each other through the selected social media platform
Creating a WhatsApp group	admin	The admin creates a WhatsApp group for each mentor
Adding to a WhatsApp group	system	The mentee is added to the requested mentor's WhatsApp group
Scheduling online meetings	system	Scheduling
Email confirmation of online meetings	system	An email is sent to the mentee to confirm the date and time of the online meeting
Monitoring interactions between mentors and mentees	admin	The admin monitors all interactions for undesired actions by mentors or mentees

Table 7-1 Tasks that are completed by the admin or the system

When the system receives the first request to join the WhatsApp group of a specific mentor, the admin creates the mentor's WhatsApp group. Each mentor has her own group, where her mentees will be added upon request. In the case of a *Join a WhatsApp group* request, the system adds the mentee to the group of the selected mentor and introduces her to the mentor and the members of the group, along with her purpose for contacting *Qudwa* (Figure 7-3). In the case of introduction through social media, the admin connects and introduces the mentee to the mentor through the selected platform.



Figure 7-3 Introducing two mentees in a WhatsApp group

7.2.4 Data collection

Submissions

All forms submitted by the mentees were collected for analysis.

Observations

All sessions between mentors and mentees were recorded, including WhatsApp transcripts, audio recordings of online meetings and transcripts of interactions between mentors and mentees on social media platforms. The content helped in understanding the relationships between mentors and mentees. Some of the interview questions were derived from the interactions and behaviours of the participants.

Analytics

Two types of analytics were collected, one from the social media accounts (Twitter and Instagram) used for promoting *Qudwa* and the other from *Qudwa* itself. The analytics and numbers collected from the Twitter and Instagram accounts showed user engagement with the promotions. *Qudwa* recorded the number of visits, visitors, paths and activity of users. The analytics helped in knowing whether a user was a returning one or a new visitor and also in

viewing the path of that visitor, which pages she visited and which mentors she viewed. Due to privacy, a single visitor could not be identified unless she submitted any form, whereby she could be identified. The analytics were used in the interviews to understand the patterns of mentees.

Survey

An online survey was sent to the mentees by the end of the study. It discussed three aspects: the nature of the relationship, communication channels and the impact of *Qudwa* on them. The nature of the relationship asked about the mentee's experience with mentoring, Qudwa and the mentor. It also asked how the mentee perceived her mentor and their relationship. The communication channels section asked about the available channels used in *Qudwa* and the proposed channels for communicating and promoting. It also asked for the mentee's feedback and reason for choosing these channels. The impact of *Qudwa* section asked about how comfortable the mentee was in asking questions and how their participation helped them. It asked about the role of the intermediate person as well. The survey was in Arabic.

Interviews

Interviews were conducted with mentors and mentees by phone and through WhatsApp to discuss their experiences with Qudwa. All interviews were conducted in Arabic. Phone interviews were conducted with four mentors who agreed to be interviewed. Each interview lasted for approximately 40–60 minutes. Three of the mentors had been in contact with mentees through WhatsApp groups and audio online meetings, while one mentor had only contacted her mentees through a WhatsApp group. Requests for interviews were sent to 20 mentees through WhatsApp. The mentees who had shown significant behaviours were selected, for example, a mentee who changed her requests several times, a passive mentee and a mentee who missed her meeting. As a result, a total of ten interviews were conducted. Four mentees agreed to have a phone interview, while another five mentees desired to answer the questions through WhatsApp. One participant (identified as Submitter1), who submitted a Let us help form but did not submit a contact request, was sent an interview request and she agreed to be interviewed by phone. All interviews lasted 15-20 minutes. All interviews were post-study except for one mentee, who was interviewed during the deployment of the study to understand the problems or reservations she had that might have led to the low engagement with *Qudwa* at the beginning of the study.

The interview questions were of three types: general, based on my observations and interviewee-specific. The interviews were open semi-structured and led by the interaction and

141

answers of the interviewee. The general questions were asked to all the interviewed mentors, covering their previous mentoring experience, their experience with *Qudwa* and their opinion of the channels. Some questions were a result of my observations from the audio online meetings, WhatsApp chats and mentees' interviews. These questions addressed the repetition of mentees' questions, the complete inactivity (passiveness) of some mentees, the feedback or appreciation from mentees and why the mentees did not leave the WhatsApp group. Other questions were interviewee-specific based on the activity and interaction of the mentor or mentee on *Qudwa*.

7.3 Findings and Analysis

In this section, the findings of the study are presented, starting with the results for the recruited mentors, followed by a report on the overall submissions received. Then the collected data about mentees' demographics and their reasons for joining *Qudwa* is documented and analysed. The following sections talk about the findings for each communication method: WhatsApp groups, online meetings and introduction through social media. Finally, the results of the online survey are reported, followed by analysis of the interviews conducted with mentors and mentees.

For identification, the mentors are identified by '*Qudwa*' then a number from 1 to 15. Numbers are based on the alphabetical order of the mentors' first names, for example Qudwa1. Mentees are numbered based on the order of their requests. The first to send a request is identified as Mentee1 and so on.

7.3.1 Mentors

A total of 15 mentors completed the forms they were sent and were recruited. Seven of the mentors shared their personal photos. Mentors were asked to tag themselves with science, technology, engineering or math, and they could tag more than one and use any other tags that they thought fit their specialty. Some of the mentors had overlapping tags, for example, engineering and technology or technology and science. Five mentors tagged themselves with science, 11 tagged themselves with technology and four tagged themselves with engineering (Figure 7-4). There were no mentors with maths backgrounds. Two tagged themselves with science and technology, and three tagged themselves with technology and engineering (Figure 7-5).





Figure 7-4 Examples of mentors with engineering, science or technology tags

Figure 7-5 Profiles of the three mentors who tagged themselves with technology and engineering

Six mentors shared only their Twitter accounts, two shared only their LinkedIn accounts and five shared their accounts on both platforms. One mentor shared her Twitter account and her personal webpage, and another shared her Twitter and Facebook accounts.

The tags that were created by the mentors were: developer, software engineer, genome, bioinformatics, biomedical engineering, interior designer, aircraft, biotechnology, HCI, UX, biology, genetics, health, web and medicine.

7.3.2 Submissions

A total of 82 submissions were received (Table 7-2). Some submissions were sent from the same mentees and others changed their request after the system got in contact with them. Some requests were not fulfilled because the mentee did not reply to the system. Table 7-3 shows the number of submissions for each communication method received for each mentor. Qudwa9, Qudwa10 and Qudwa11 did not receive any submissions.

Type of request	Number of submissions
Join a WhatsApp group	23
Online meeting	8
Introduction through social media	11
Let us help	23
Recommend a mentor	15
Total submissions	82

Table 7-2 Total number of submissions and number of each type of request

Table 7-3 Number of submissions received for each mentor

	Join WhatsApp group	Online meeting	Introduction through social media
Qudwa1	1	1	0
Qudwa2	3	2	0
Qudwa3	1	0	1
Qudwa4	1	0	0
Qudwa5	4	0	0
Qudwa6	3	0	1
Qudwa7	1	0	0
Qudwa8	1	1	1
Qudwa9	0	0	0
Qudwa10	0	0	0
Qudwa11	0	0	0
Qudwa12	0	1	1
Qudwa13	1	0	1
Qudwa14	2	2	3
Qudwa15	4	1	3

Table 7-4 shows the multiple submissions sent by individual mentees. Six mentees requested to connect with more than one mentor, using the same method or different methods. Two mentees asked to join the WhatsApp groups of two different mentors. One mentee sent two requests to have audio online meetings with two mentors. Another mentee requested to join a WhatsApp group with one mentor and an audio online meeting with another mentor, and one mentee had three requests, two audio online meetings with two mentors and a request to join a WhatsApp group with a third mentor. All multiple submissions were sent on the same day by the mentees except for one mentee, who submitted the second request 20 days after her first submission.

	Join a WhatsApp group	Online meeting	Introduction through social media
Mentee3	2		
Mentee6	1		1
Mentee16		2	
Mentee17	1	1	
Mentee26	2		
Mentee28	1	2	

Table 7-4 Number and type of submissions sent by the same mentee

Some mentees changed their requests after the system got in contact with them. Mentee16 originally requested two audio online meetings with two different mentors, but when the system contacted her, she changed her request to joining the WhatsApp groups of the two mentors. She explained: "WhatsApp is easier for me to explain what I want with ease"*. Mentee17 had requested an audio online meeting, but after the system contacted her, she changed her mind and asked to join the mentor's WhatsApp group. After joining the WhatsApp group, she again asked for an audio online meeting with the mentor.

All mentees requested to be connected with different mentors within the same field except for two mentees. One was a first-year undergraduate studying biomedical engineering and she was thinking of changing her major to computer science. Therefore, she asked to be connected with two mentors, one in the field of biomedical engineering and the other in the field of computer science. The other mentee had an interest in the medical field and she asked to be connected with one mentor in the field of medicine and one in the field of biomedical engineering. This behaviour can be explained due to the fact that mentees are seeking information and may contact several mentors in the same field or different fields. Mentee17 had a specific reason for this but the others were collecting knowledge and information from all available resources (mentors on *Qudwa*).

Let us help

The total of *Let us help* submissions was 23 from individual submitters. Seventeen submissions were requesting fields not supported by *Qudwa* such as finance, arts, translation, fashion design, law and business. Only six submitters were requesting help in a STEM field. Two mentees asked to be put in contact with a mentor via WhatsApp after the admin contacted them. One mentee was found to have submitted another request to join a WhatsApp group. The remaining three were contacted by the admin; two did not reply and one said that she would submit a request but she did not.

Three mentees submitted a *Let us help* form even though the fields they were requesting were available on *Qudwa*. One mentee requested help and before the admin contacted her, it was found that she had submitted a form requesting an audio online meeting with a mentor. She explained in her interview that she had submitted the help form even before she explored *Qudwa*. The admin contacted the others to help them with their needs.

One participant, identified as Submitter1, submitted a help form but did not proceed and submit a contact form after the admin contacted her. Through the interview, she explained that she had not submitted a form to contact a mentor because she wanted to be ready and have enough background about the field before she talked to a mentor.

Recommend a mentor

The system contacted all 15 recommended mentors by email and sent them an information sheet, but no response was received. It is not possible to know whether the recommendations came from the person herself who wanted to be a mentor on *Qudwa* or someone else had suggested that she could be a mentor.

7.3.3 Mentees

Qudwa had a total number of 1107 original visitors and 8288 repeat visits. The total number of participating mentees was 35. In the submission forms there were two textboxes, one asking the submitter to introduce herself and the other to specify the reason she was requesting to contact the mentor. This helped in learning about the mentees.

At the time of the study, most of the mentees (18 mentees) were at the age of 18 years, had graduated from school and were applying for university. *Qudwa* was mainly targeting

females who were graduating from high school (or younger) and undergraduates to guide them through university applications and career paths, but surprisingly the findings show the participation of older age groups (Table 7-5). Of the 35 mentees, 28 mentees (80%) were 19 years old or younger, which is the age of high school graduates, and three mentees were undergraduate students.

Older mentees used *Qudwa* to learn about other fields that they did not have a background in or were not familiar with. The eldest mentee was Mentee4 (34 years old); she wanted to learn about a field that she was not familiar with. Her background was in the Arabic language and she used *Qudwa* to increase her knowledge about a technical field. She mentioned in the submission form that she had looked at many websites, but had not found information that could help her:

I am a graduate of the Arabic language department and I hope to become knowledgeable about the technical field ... I searched on several websites, but I could not find useful information to help me.* – Mentee4

The same with Mentee2 (23 years old), who was interested in business and technology while her major was languages and translation. She joined *Qudwa* because she was passionate about learning new things and to figure out which path to take regarding her career.

On the other hand, Mentee3 (28 years old) was a postgraduate student seeking knowledge from the mentor's experience. Mentee14 (23 years old) was general in her request and purpose for joining Qudwa.

Born in	Age	Number of mentees
1985	34	1
1991	28	1
1995	24	1
1996	23	1
1997	22	1
1998	21	1
1999	20	1
2000	19	5
2001	18	18
2002	17	3
2003	16	2

Table 7-5 Age group of mentees

To understand the mentees better and have some initial background about them, they were asked to write a brief introduction about themselves and their purpose for contacting the mentors. The next sections will present the findings from their submissions.

How the mentees introduced themselves

In the textbox *Introduce yourself* most of the mentees (68.5%) described themselves briefly, mainly mentioning their name, education level and age. Some added a few words to describe themselves:

I'm ****, I'm 18 years old and I live in Jeddah.* - Mentee8

I study biomedical engineering. - Mentee18

The remaining mentees (31.5%) described themselves in more detail than the examples above. They talked about themselves, what they wished for and their hobbies and

interests in general:

I am a Computer Information Systems student at **** University, I consider myself a perfectionist, I have very high expectations of myself. I'm confident in my ability to work hard and deliver quality results, I always try to develop myself by participating in many volunteering activities and taking courses in my field of study and beyond. – Mentee5

Hi, my name is ****. I'm a student at **** university. I just finished my prep year with a GPA of 4.9, enrolling in BME. However, I have some questions regarding this major because I'm considering switching it. – Mentee17

Mentees' purpose for contacting the mentor

In the textbox *Why would you like to contact this mentor*? the answers were divided into two groups. The first group wanted to learn: learn about the experience, learn about the field, learn about job opportunities and learn about the struggles faced by the mentors. The other group was looking for guidance and advice on applying to universities, choosing a major or guidance on a specific field (Table 7-6).

Five mentees wanted to learn about the mentor's experience; for example:

To benefit from the experience and knowledge of Ms *****. - Mentee3

I want to benefit from the experience of people who have majored in this field.* – Mentee22 $\,$

Mentees' purpose for contac	cting the mentor	
Learn about		Number of mentees
	experience	5
	field	13
	job opportunities	4
	the struggles and difficulties	3
Guidance and advice on	choosing a major	9
	applying to universities	2
	career future	1
	a specific field	6

Table 7-6 Why the mentees were requesting to contact the mentors

Others wanted to learn about the field and job opportunities. Mentee2, who was a fresh graduate, wrote:

Because I want to learn more about business analysis and technology and which path I can take regarding my career.

Mentee17, who was an undergraduate, wrote:

Questions about BME, hows the job like. And opportunities. I would like to contact this mentor because I find her cv very interesting and inspiring.

Some mentees wanted to learn about the struggles and difficulties faced by the mentor

as a female in the work environment:

I would like to gain experience – to ask more about the field of this role model – maybe in the future I will become like her – I would like to ask about the universities that provide her field – to deepen the understanding of this field – and to know the difficulties encountered by this role model.* – Mentee12

To increase my knowledge about the health track in the university and to identify the difficulties in this field to overcome them in the future, God willing.* – Mentee7

Other mentees were seeking guidance and advice from mentors. Some were looking

for advice about the field from someone specialised in that field:

I have just taken my HCI course in college and it opened my eyes literally on the design of every site and app. Definitely, an area that I'm interested in and want to give more to, and I would love the opportunity to learn more about it especially in the UX field. I need some advice about how to be present in this field, what would be my next step? I'm really interested in working in projects related to this field where do you think I should go? – Mentee5 I need guidance and advice from someone specialised in the field because they are the ones who we should ask.* – Mentee6

The majority were seeking advice in choosing their majors. Some mentees showed despair and uncertainty in their submissions. Two mentees were seeking guidance and advice on applying to university:

I'm confused about the major I should pick at college and I have some interests in software engineering, but I'm looking forward to knowing more about it. – Menteel

I need advice and instructions on how to register at the university.* – Mentee19 $\,$

7.3.4 Introduction through social media

Seven of the *Introduction through social media* requests were asking to contact the mentor through a platform that was not provided by the mentor herself. Five requested to be introduced through Instagram and two through Snapchat. Four requested introductions through Twitter. All submitters who had requested to be in contact through Instagram or Snapchat were contacted by the admin, who clarified that these platforms were not provided by the mentors. The admin also suggested the use of other available methods on *Qudwa*.

There was a problem with creating a group on Twitter for a mentor, a mentee and the admin to join because the mentee had a private account and Twitter requests that members of a group should be following each other. Therefore, after the first attempt had failed, any further requests to be introduced through Twitter were responded to by the admin with the suggestion of using other methods. Nine of the submitters chose to join the WhatsApp group of their selected mentor, one did not reply to the admin and one request was overlooked by the admin.

7.3.5 Join WhatsApp group

The number of submissions to join a WhatsApp group was 23, but the total number of mentees in the groups was 34. The difference in numbers was because of the changes in the requests of the mentees. Two mentees had submitted two requests to join two different groups and nine mentees had originally requested an introduction through social networks but, as explained earlier, they joined WhatsApp groups. Two mentees asked to join their mentor's WhatsApp group after they had an online meeting with them. Two asked to join the WhatsApp group after the admin had contacted them upon their *Let us help* submissions. One mentee requested an online meeting, then changed her mind and wanted to join the mentor's WhatsApp group.

Some groups acted as a one-to-one relationship between a mentor and a mentee. Qudwa1, Qudwa3, Qudwa4, Qudwa7, Qudwa7, Qudwa8 and Qudwa12 had only one mentee in their groups. The highest number of mentees in one group was 11 in the group of Qudwa15 (Table 7-7).

Mentor	Number of mentees in group
Qudwa15	11
Qudwa2	4
Qudwa5	4
Qudwa14	4
Qudwa6	3
Qudwa13	2
Qudwa1	1
Qudwa3	1
Qudwa4	1
Qudwa7	1
Qudwa8	1
Qudwa12	1

Table 7-7 Number of mentees in each WhatsApp group

Among my observations of the WhatsApp groups, some behaviours were noted: 1) passive mentees; 2) lack of response from mentees; 3) leaving/staying in a group; and 4) lack of appreciation from mentees. Some mentees did not participate at all after they were added to a group. The admin introduced them, their questions and the reason they wanted to contact the mentor, and the mentor replied to their questions, but there was no further response from the mentee. This behaviour occurred several times in different groups. Mentee29, who asked for an interview but replied through WhatsApp, was passive in the group and then left the group. She explained that what happened was not what she had expected and that she had expected it to be a training session or a course. On the other hand, Mentee11, who was also passive, expressed that she felt weird asking the mentor when the group was sometimes inactive:

I don't know, maybe because the group was silent and there was no activity, I felt weird asking suddenly.* – Mentee11

Other mentees were interactive for some time and then suddenly inactive. This was obvious when there was a conversation going on between a mentor and a mentee in the group and the mentor was responding to the mentee, but the mentee did not respond or show any sign of receiving of the information or response the mentor had shared.

In conducting the interviews with the mentees and conducting the survey, a large amount of appreciation was shown by the mentees. This made obvious the lack of appreciation or thankfulness in the WhatsApp groups.

7.3.6 Online meetings

There were eight requests for audio *Online meetings* but the meetings actually conducted numbered six because one mentee had requested two meetings with two different mentors but, after the system contacted her, changed her mind to join one of the mentors' WhatsApp group and cancelled her request to contact the other mentor.

All the six online meetings were conducted on time except for two cases where the two mentees missed their scheduled meetings with their mentors. In the first case, the admin tried to contact Menteel via email (she did not provide her mobile number) after the meeting had started, but she did not reply. The mentor kindly agreed to wait for 20 minutes, but the mentee did not show up. The next day, the mentee apologised and explained that she had had an urgent matter and could not join the meeting, and she requested a second chance. The mentor agreed and the meeting was conducted on another date and time.

In the second case, Mentee28 had booked two meetings on different dates with two different mentors, and missed the first meeting even though the system reminded her 15 minutes beforehand via WhatsApp and she confirmed. The admin tried several times to reach her again through WhatsApp. The mentor agreed to wait for 20 minutes, but the mentee did not show up. Forty minutes later, she explained that she had thought it was a phone call. The meeting was rescheduled and conducted the next day. On her second meeting with the other mentor, she was 30 minutes late. Her excuse was that she was not at home. Mentee1 was apologetic and tried to reach and explain to the admin, while Mentee28 did not try to reach out to find why the meeting did not happen or whether the mentor was late.

7.3.7 Survey

An online survey was sent to all mentees who participated in *Qudwa*. The survey link was sent through WhatsApp. Out of the 35 mentees, 21 completed the survey. The survey discussed three aspects: the nature of the relationship, the communication channels and the impact of *Qudwa* on the mentees. This section presents some of the survey questions.

152

Nature of the relationship

Have you ever talked to a professional to help you find answers regarding your studies or career?

Seven out of 21 mentees indicated that they had contacted a professional to seek help before they participated in *Qudwa*. One mentee elaborated that she had done a test with a professional from her university to help her choose a suitable major. Another mentee mentioned that she had asked through people she knew in the community. Seven out of the 14 mentees who indicated that they had never contacted a professional before explained their answers:

I have talked with people who had experience, but not professionals.*

I tried asking professionals on social media, but I did not get any responses.*

How would you rate your overall experience with Qudwa?

The Likert scale ranged from 1 (very bad) to 7 (very good). The results show that 11 out of the 21 mentees described their experience with *Qudwa* as very good, four as good, five as somewhat good and one was neutral. There were no ratings below neutral.

How do you rate your experience with the mentor in terms of her ability to help you?

The Likert scale was from 1 (very bad) to 7 (very good). On rating the mentors that they were in contact with, 15 out of the 21 mentees rated their mentors as very good, two mentees rated their mentors as good and three mentees as somewhat good. There were no ratings below neutral (one mentee).

I felt the need for more guidance and help than Qudwa provided

The Likert scale was from 1 (strongly disagree) to 7 (strongly agree). The results show that five of the mentees strongly disagreed and three of the mentees strongly agreed that they needed more guidance and help. Five of the mentees were neutral.

The available information about each mentor was enough for you to decide which mentor to choose

The Likert scale was from 1 (strongly disagree) to 7 (strongly agree) and the results show that

the majority of the mentees agreed (six mentees strongly agreed, four mentees agreed and another four mentees somewhat agreed) and one mentee strongly disagreed, while two were neutral.

How do you feel about the short-term relationship with your mentor?

The Likert scale was from 1 (strongly dissatisfied) to 7 (strongly satisfied). The results show that the majority of mentees were satisfied with the short-term relationship (eight out of the 21 mentees were strongly satisfied, six were satisfied and three were somewhat satisfied), 13 mentees were neutral and one mentee was somewhat dissatisfied.

Could your experience with Qudwa be replaced by a Q&A website, and why?

The Likert scale was from 1 (strongly agree) to 7 (strongly disagree). The results show that eight of the 21 mentees strongly disagreed, five somewhat disagreed and four disagreed. The results also show that one mentee strongly agreed and another mentee agreed, while two mentees were natural.

One mentee who rated the replacement of *Qudwa* with a Q/A website with *somewhat agree* justified her answer:

I might find a question that formulates what I'm thinking of. Instead of being distracted by formulating the question, I might find someone who had the same questions and asked them better than me.*

The other mentees who thought that *Qudwa* could not be replaced:

Nice idea but an old one, because it will almost become the same as forums that 'ask questions and get people to answer' but on social networking sites such as WhatsApp and Twitter I see that it is better because it is the most widespread means of communication in this era and with more interaction.*

Communication channels

How did you know about Qudwa?

Most of the mentees knew about *Qudwa* from Twitter and Instagram (nine and eight mentees). The results also show that three knew about *Qudwa* from WhatsApp. One mentee mentioned that she knew about *Qudwa* from a friend.

What other channels would help promote and reach more girls?

This question had multiple options: Twitter, Instagram, WhatsApp and other [specify]. The

respondents were asked to select all that applied. Twitter was chosen the most (16 responses), followed by WhatsApp (11 responses) and then Instagram (six responses). Two social media platforms were suggested by the mentees: Telegram and Snapchat.

Why did you choose your communication channel?

The question had multiple options and the respondents were asked to select all that applied. The options were: 1) ease of use; 2) I use it a lot; 3) I feel confident using it; 4) privacy; 5) ease of expressing myself; 6) other [specify]. *Ease of use* had the majority with16 responses, followed by *ease of expressing myself* (10 responses). The results show that *I feel confident using it* got seven responses, *I use it a lot* got six responses and *privacy* got five responses. Two mentees who communicated through WhatsApp added other reasons, one that she wanted to try it out and the other that the method attracted her.

Did you face problems with the chosen communication channel?

All mentees responded with no except for one, who wrote that she faced problems with scheduling.

What channels do you prefer to use other than the one used in Qudwa?

This question was a short-answer textbox. Five mentees answered that there were no other channels that they preferred. Four mentees left the question unanswered. Two mentees mentioned WhatsApp, one of them saying: "WhatsApp is the fastest and most private method"*. One mentee suggested the use of the application Zoom because in her opinion most people used it for calls. One mentee preferred the existence of a standalone application for *Qudwa* where she did not have to share her information such as phone numbers. Another mentee recommended the use of groups to benefit from others' questions and discussions. One mentee wanted to use an application that allowed calls without specifying which application. Others suggested: Twitter, Telegram, Snapchat and Instagram.

Did you have any reservations regarding your participation in Qudwa?

The question had multiple options and the respondents were asked to select all that applied. The options were: 1) no reservations; 2) privacy; 3) communicating with strangers; 4) the nature of the communication channels; 5) other [specify]. More than half of the mentees (16 out of 21 mentees) had no reservations.

Did you visit the mentors' social media accounts provided on Qudwa?

More than half (12 mentees) answered "No". There was a follow-up question: "If yes, did you find providing the social media accounts of the mentors beneficial to you?". All who answered "Yes" answered the follow-up question with "Yes".

The impact of Qudwa on the mentees

How comfortable were you asking the mentor questions?

The Likert scale was from 1 (strongly uncomfortable) to 7 (strongly comfortable). The highest percentage of mentees felt comfortable asking questions (17 out of 21 mentees) and only two mentees felt somewhat uncomfortable.

How do you rate the effect of displaying all the mentors from different fields to you to raise your awareness about study and career options?

The Likert scale was from 1 (strongly ineffective) to 7 (strongly effective). The results show that the majority of the mentees felt that displaying the mentors' profiles on *Qudwa* was effective in terms of raising their awareness of opportunities for them.

How do you feel after your participation in Qudwa?

The Likert scale was from 1 (strongly confused) to 7 (strongly determined). The results show that 16 out of 21 mentees felt more determined than before their participation in *Qudwa* (seven were strongly determined, six determined and three somewhat determined) and only one mentee felt somewhat confused.

How helpful was the presence of an intermediate person (Aseel) for guiding and communicating in comparison to an automated system?

The Likert scale was from 1 (strongly useless) to 7 (strongly helpful). The results show that most of the mentees found the presence of an intermediate person helpful, while two mentees found it useless.

If you had the chance to make changes to Qudwa, what would you change or add?

The results showed that seven of the mentees thought that *Qudwa* did not need any changes. Some of them elaborated: No changes needed; the system is good as it is.*

Qudwa does not need to change, I hope it develops and adds more fields.*

Nothing, I wish it continues. Many thanks.*

Some mentees added their suggestions for change. One mentee wrote: "speed of response"*, another added: "an app for Qudwa"*. One mentee suggested adding activities and introductory information about the mentors. Two mentees proposed that *Qudwa* add mentors from more fields. One mentee suggested promoting *Qudwa* so that more girls benefited from it. One mentee said: "As I mentioned earlier, in groups with other people – the mentor asks questions that help the user determine their choice – the questions are not unliteral"*.

7.3.8 Interviews

Thematic analysis was applied to all interviews with mentors and mentees. The interviews were analysed using the inductive approach of Braun and Clarke (2006). Several themes were identified: the need for *Qudwa*, impact and influence, differences between mentors and mentees, cultural factors and *Qudwa* and mentors' daily activity.

The need for Qudwa

In addition to the reasons that *Qudwa* was initiated in the first place, which were to guide and help young women due to the lack of mentoring and resources in Saudi Arabia, other needs emerged too. The analysis showed that *Qudwa* was a means for the mentors to help themselves on a personal level. Despite the existence of people around them and professionals on social media platforms, the participants felt certainty and trust in receiving a response from *Qudwa*. Some interviewed mentors and mentees expressed their wish for *Qudwa* to continue and grow.

Lack of mentoring and resources

The lack of mentoring and the gap between schools and workplaces discussed in previous studies re-appeared in this study. Mentors explained their previous experiences when they were young and how they needed support and guidance:

We've been through that 'looking for help' phase and now that we've been through it, we want to help others. We didn't have this kind of support when we needed it, so why don't we help the younger generation?* – Qudwa8

Qudwa8 added later on that she had shared the concept of *Qudwa* with people she was working with in the UAE. They were working on a project to guide and mentor fresh

graduates through their career paths. Qudwa5 and Qudwa14 talked about their past experiences of choosing fields (biomedical engineering and electrical engineering) that were not widely known as options for females in Saudi Arabia:

The options are completely unclear. For example, electrical engineering was provided only at *** university several years ago. I loved physics and I thought that electronic engineering was a good option for me. But what would be my job? I didn't have any vision of what a female could be with an electronic engineering major ... If there was *Qudwa* at that time, I would definitely have used it. And especially if she's a female! I would ask her what will I study? What do you do? Where can I get a job?* – Qudwa14

Qudwa2 stressed the gap in the concept of computing between what is learned in schools and universities and what a person with a computing major actually does. Qudwa14 emphasised the lack of Arabic resources and content available online that would help students in understanding and finding answers to their needs regarding future planning. She explained that she was surprised that girls would still ask questions about the differences between computing fields even though computing had been taught in Saudi Arabia for many years. She thought that this kind of information should be well known by now.

From the mentees' side, the need for *Qudwa* was present too. They had questions that they could not find answers to, they needed to be inspired and they needed guidance from someone professional. Mentee23 expressed that *Qudwa* was needed among girls her age. She described her experience with *Qudwa*:

It solved the problem of girls my age, they need someone to guide them and advise them. They have lots of questions and they wish someone will give them answers ... There were questions on my mind without answers and I found my answers ... Lots of people will benefit and find answers to their questions. Some people are devastated, they have nothing on their minds, they don't have a goal. If they read about other people's experiences, they will be encouraged.*

Mentee13 confirmed the thoughts of Qudwa2 and Qudwa14 on the confusion in understanding the differences between fields. She reported:

I needed more information about the computing fields and the differences between them, because there were many fields.*

Among all the interviewed mentees, Mentee5 was focused and specific about what she needed. She was in the final year of her undergraduate study. She sought guidance on how to proceed and plan her career path.

The certainty of receiving a sufficient response on Qudwa

In talking about their reasons for joining *Qudwa*, mentees reported that they had previously sought help using other methods, which included their relatives and professionals on social media. They had reached out for help through their relatives and people around them, but the information they needed was not provided. They mostly did not receive a response from professionals on social media, and if they did the answers were insufficient. They believed and were certain that mentors on *Qudwa* would respond adequately. The reason behind their certainty was because these mentors had volunteered themselves with the intention of helping and guiding through *Qudwa*. The comments below present the mentees' thoughts:

I did try to contact a professional before, but it took a long time for her to respond to me, or she didn't provide a way to communicate. She just provided answers [on her website] to what she thought are the most expected questions. Most of them do not provide the opportunity for us to contact her in person.* – Submitter1

Honestly, with social media like LinkedIn and Twitter, yes there are many, but I don't have the confidence that she will reply to me like being on this website [Qudwa]. I might and it happened, honestly, that I asked people through email, but they didn't reply. But on Qudwa I was sure that someone would reply to me.* – Mentee5

The certainty of receiving a response was also noted in Qudwa14's response when she was asked if she would use *Qudwa* for a professional-to-professional connection:

Yes, I would contact her on *Qudwa* and not on her personal account on LinkedIn, for example. Because I know she is willing to help. The concept of *Qudwa* helps you to ask without feeling that you are intruding or annoying them. People honestly don't reply on LinkedIn.*

She also discussed how the girls around her did not get responses from people and if

they got responses, they were brief. I asked if those girls got answers and she responded:

No, unfortunately. People don't respond to them. They rarely get answered and if they get an answer, it would be brief and it doesn't satisfy them.*

Personal benefits for mentors

The mentors felt that *Qudwa* was beneficial for them on a personal level. It satisfied their needs to give and help, feel better about themselves and create a positive change in their lives. Qudwa8 expressed that she felt she had responsibilities towards her field, towards young females and towards her society. She reported:

First of all, I love to help in general, especially helping the new generation. Second thing, it encourages girls to be in our fields, STEM fields. It's our responsibility to encourage girls to go into these fields. And I have another responsibility, because my field is new and is not well known here in Saudi while it is wanted on the job market. It's an opportunity to make girls know about it, think about it and guide them to it ... so a responsibility towards society, my field and the girls.*

Two mentors had thought of their participation in Qudwa as a way to make positive

changes in their lives. Qudwa2 needed to enrich her life with something that she enjoyed. She

reported when asked about what had encouraged her to be a mentor on Qudwa:

Sharing knowledge is something that I enjoy. I wanted something pleasant in my life other than my work, you know. I always work on my computer and coding, I needed another aspect in my life related to my field.*

The case with Qudwa14 was different. At the time I approached her with the concept

of *Qudwa* she was not feeling content with herself. She felt unfocused and seemed to forget a lot. She explained:

There is a study of someone, I forgot his name, I think it is Jordan Hopkins, it said that mentoring helps mentors in reducing the decline in brain performance and abilities. The time when you contacted me, I was forgetting a lot and unfocused. That's why I wanted to take advantage of this opportunity of mentoring to increase my brain functionality. That's one of the reasons. It was not the best time of my life. I thought of doing something that could take my mind off it. Something that would help me feel better about myself. Wow, I did something good ... something like that, you know.*

She added that the mentees' questions reminded her of some of the knowledge that she had forgotten she had. She was thankful and appreciative of the knowledge she had; she shared her experience with her cousin, who was suffering from the same problem of feeling useless.

Continuity of Qudwa

Three mentors wished for *Qudwa* to continue and grow. Qudwa8 suggested improving *Qudwa* in ways to reduce pressure on the system, to ensure its sustainability and continuity. Qudwa5 and Mentee23 shared the thought of *Qudwa* getting larger and reaching a higher number of girls. Qudwa2 liked the idea of getting help from people from your own country and commented:

I think the effort you've put into *Qudwa* is huge and I hope the project grows bigger. The channel is amazing, to know that there are people available in one place and that you can communicate with them in any way you want. We

are used to going on the internet to seek help from people in other countries, it's nice to see people available here from our country.*

Impact and influence

Through the interviews, impact and influence were evidenced in different situations. While mentors felt the vagueness of their impact on the mentees, mentees showed different levels of how their participation in *Qudwa* had had an impact on them. In addition, peer influence was noted among participants.

Vagueness of impact

Due to the mentees' low interactivity and some lack of appreciation in the WhatsApp groups, three mentors felt the vagueness of their impact on the mentees. Qudwa5 clearly said that she did not know whether the mentees had benefited from her or not. Qudwa14 assumed that she was not helpful to her mentee because the mentee stopped responding. Qudwa2 shared the same concern when she was asked about her interaction with the mentees. It was confusing for her not to know whether the mentees were benefiting from her or not. She explained that the lack of interactivity might be because of her style in conversing and that she did not know how to approach the girls. The idea of anonymous feedback during the mentoring relationship would help her to know whether she is doing the right thing or not:

The feedback idea would allow her to talk about the problem. Maybe in the conversation, there is a problem, that's why she doesn't feel she benefited from me ... feedback is the best way to know if we are on the right track or making them even more confused ... I'm not saying that thanking me is a crucial point to me, but that's something I noticed, if they didn't interact, I don't feel that they had benefited.*

On the other hand, two mentors believed that the least they could give was sharing their experience with the mentees. Qudwa2 expressed that what she shared with her mentees was experience, something that was not taught anywhere. Qudwa8 explained that she had trouble with her mentee during a call because the mentee was not interacting with her, so she decided to talk about her life/work experience:

I believe that telling your experiences to others is beneficial. Since I couldn't advise her on what to study, why don't I tell her about my experience in life? ... I felt that I didn't give her much but somehow she benefited from what I said.*

Two mentors who had the chance to experience the mentoring through audio calls felt the appreciation of their mentees. Mentors who were in contact via WhatsApp varied in their answers. Qudwa8, who had two audio calls with two different mentees, described them as two different personalities, and felt their appreciation and gratitude. Qudwa2 was affirmative that she did not receive any kind of appreciation from the mentees in the WhatsApp group, which bothered her; on the contrary, she felt sincere appreciation from the mentees she talked to through audio. She was bothered by the sudden halt of interaction in the WhatsApp group:

There is no doubt there was no appreciation ... If you go back to the conversation, it's open without an end, my reply was the last thing. I can't tell if she's with me on the opinion or opposes me. There was not even a thank you for the opinion, the subject is still open. Ending the conversation with a thank you is enough for me.*

Qudwa5, who only experienced mentoring through WhatsApp, reported that she felt appreciation from her mentees and could not recall if they did not as it was not an important issue to her. Qudwa14 at the beginning did not remember if there was any lack of appreciation, but then she recalled that she felt appreciated from the audio call but not from the WhatsApp group. Because she volunteered to participate in *Qudwa*, the lack of appreciation or feedback did not annoy her. She reported that the reason for that was the nature of the communication methods:

It's a question and answer, it's not a continuous relationship like in an audio call.*

Different levels of impact

Even though mentors felt confused regarding their impact, interviews with mentees showed different levels of impact. Mentee17 changed her major after she was in contact with a mentor; while her decision was not only based on her talk with her mentor, it gave her reassurance that her decision was right for her. Mentee13 participated in *Qudwa* to know the differences between the computing fields. She stated that her participation had a positive outcome and that she understood the main differences between computer fields.

Some mentees through the interviews showed their appreciation and the positive impact *Qudwa* had on them:

Qudwa had a positive impact on me, I'm excited to graduate school and apply to universities. She did a great job with us; other people might say that they are busy. She gave us from her time, she answered lots of questions that were on my mind.* – Mentee23

Peer influence was another form of impact that was noted from the interviews. Mentees shared their experience on *Qudwa* with their peers and advised them to join and participate. Mentee5 shared her experience with a friend who she thought needed guidance and help. It has been found through the analytics of *Qudwa* that the search term of "biomedical engineering" was entered, which is the field of her friend.

Mentee23 told her friends that she would have a call with a mentor. She was asked about how they responded:

They told me that I should give them my feedback after the meeting. When you are done with the meeting, tell us, we might want to do the same ... After the meeting, I had a group call with them on Snapchat and I told them what I asked and that she was really nice and she gave me more than the time we scheduled and she was talking to me in a friendly way, not like she was years older than me.*

She also shared her experience with her younger sisters, who were upset that they did not know beforehand about the meeting because they wanted to join the meeting with her.

Mentee17 told her sister, who specialised in neuroscience, about *Qudwa* after she saw a mentor on *Qudwa* with the same specialty. She had shared her experience and the information she learned from her mentor with her friends too. Mentee13 also mentioned that she shared her experience with one of her friends.

Another form of indirect impact was visible from browsing the mentors. Submitter1, who did not contact any of the mentors, stated that she was amazed by the qualifications of the mentors, while Mentee17 did not expect that there were Saudi females in these positions or workplaces.

The indirect influence reached the mentors as well. Qudwa5 expressed her pride in being part of *Qudwa* with the group of mentors. Qudwa14 mentioned that she had browsed one of the mentors' profiles because she thought that it intersected with her field. She added that she had browsed other mentors' Twitter accounts.

Being in a WhatsApp group also had a peer influence on the mentees. When they were asked about their reasons for not leaving their groups or why they were inactive in the groups, some mentees mentioned that they benefited from the other mentees' questions and discussions. Mentee23, after her call with a mentor, asked to join the WhatsApp group of that mentor. She thought that she might benefit from the questions of other mentees. Although she found the call better in expressing herself clearly than the WhatsApp group, she admitted that she had benefited from the group. Submitter1, who did not participate in *Qudwa*, said that if she was to choose a way of communication, she would choose a WhatsApp group. She believed that being in a group was more beneficial because others might ask questions or open discussions that did not cross her mind. Mentee13 also agreed that being in a group was favourable because of the presence of others.

On the other hand, Mentee19 had only asked one question to the mentor in the group because most of the questions she had in mind were asked by others. Mentee17, although she was not in a group, mentioned how being in other WhatsApp groups (related to her study) with her peers was helpful for her because they shared useful things. She thought that promoting in such groups would attract other girls to *Qudwa*.

Differences between mentors and mentees

There were differences in the experiences of mentors and mentees with *Qudwa*, differences in how they perceived their mentoring experience. As mentioned earlier, there was variation in the perception of the level of impact between the mentors and the mentees. There were differences in their opinions about the communication channels, relationship levels, type of help that was expected and my role as the intermediate person.

Communication channels

All the mentors agreed that they preferred audio communication in comparison to textual communication. Qudwa14, who was in contact with mentees via both methods, mentioned that she preferred audio calls. She added that she used the voice notes feature in WhatsApp to express her thoughts easily. Qudwa2 also mentioned that she did not prefer WhatsApp communication. She believed that WhatsApp gave her flexibility in time, but would be more suitable for technical questions (coding and programming). She thought that audio calls helped in understanding the mentees better. Qudwa8 reasoned that her choice of preferring audio communication related to the age of the mentees and to the critical decision they were making. Qudwa5 shared the same concern as Qudwa8. Even though she did not experience mentoring in *Qudwa* through audio calls, she believed that an audio call would be more suitable for discussing critical decisions:

WhatsApp is good as a first communication \dots I believe that there must be vocal communication, but it depends on the girl of course. If she doesn't have big concerns, she might use WhatsApp. But if she has a lot of questions and concerns and she wants to have a one-to-one conversation, I think a call would be better.* – Qudwa5

Three of the mentees (who were interviewed by phone) agreed with these mentors and favoured audio calls. One mentee (who was interviewed by phone) preferred WhatsApp group communication. As mentioned earlier, five mentees who were asked for phone interviews preferred to communicate textually through WhatsApp.
Mentee5 preferred audio calls because what she wanted to discuss with the mentor was personal and she thought that sharing it with a group of others was not the best way. Mentee23, who experienced both methods, said that she preferred audio calls. She later joined the WhatsApp group of the mentor to gain more benefit from the discussions.

On the other hand, Mentee17 changed her request several times because she did not know what to expect. In the beginning, she requested an audio call with Qudwa12 and to join the WhatsApp group of Qudwa5. After she was added to the WhatsApp group of Qudwa5, she requested to change her request for an audio call with Qudwa12 to joining her WhatsApp group. She did that because she thought the groups would have many girls, while it had only two or three girls at that time. She also mentioned that she was not confident and thought that she "is not good with people". Later, she requested an audio call with Qudwa12. Through the interview, she mentioned that she liked audio calls more than the WhatsApp method. She added that she was not confident at the beginning of audio calls, but later on felt more confident because of the way Qudwa12 was talking to her. I asked her if she felt shy or embarrassed asking questions and she replied:

Honestly yes, at the beginning, but then I felt confident and more comfortable. Because she was comforting to talk to and I felt that she had the information that I need.*

On the contrary, other mentees preferred textual communication through WhatsApp. Mentee13 (interviewed by phone) thought that an audio call would be formal and chatting through WhatsApp gave her flexibility in answering and arranging her thoughts. Other mentees who were interviewed through WhatsApp suddenly stopped responding, which made the interview came to an end without covering the discussion points. Mentee9 explained why she chose to communicate with the mentor through WhatsApp:

It was one of the easiest applications provided and it has the advantage of sharing media like voice notes.*

When I asked her why she did not choose to have an audio call since she favoured the voice notes feature, she replied:

I did not like the idea of being committed to each other through a call when I don't know how long it will last. We can chat and whenever we have time, we respond.*

The other mentees did not provide a chance to complete the interview to ask them about the communication channels. Submitter1 through the interview said that if she had to pick a method of communication, she would pick the WhatsApp group because of the benefit she would gain from the questions and discussions by others in the group.

Level of the mentoring relationship

Mentors put more emotions into their relationships with their mentees than the mentees themselves. Mentees who said that they benefited from their mentoring relationship thought that they were satisfied and had what they needed. On the other hand, mentors felt that they needed more time to understand the mentees and help them.

Qudwa8 emphasised the importance of understanding her mentees and that was why she insisted on the use of audio communications because it showed feelings. Qudwa2 mentioned that the discussions with the mentees should be longer. She needed to understand them and understand what they were thinking about. She believed that this understanding came from conversing for a long period of time and through many sessions. She also suggested sending a set of questions to the mentee before the meeting; these questions would help her get to know the mentee better and know how to approach her and help her. Qudwa14 wished for a continuous relationship with her mentees. She believed that she was putting effort and feelings, and that she would like to keep the relationship going. Qudwa5 reported that she tried to understand her mentees by repeating their questions. She wanted them to be more precise in asking questions or describing their concerns. This way she would be able to help and guide them.

Two mentors felt a difference in the way that mentees looked at mentoring in comparison to them. They expressed their thought when the discussion was about passive mentees. They thought that the mentees were using the mentoring via WhatsApp group as a "reference" or a way to "collect opinions":

I felt like she's collecting opinions, she has the right to do that of course. I think she has addressed more than one channel to collect information from them and then she will decide, that's why there is no response from her side. I felt that they did not care.* - Qudwa2

Maybe she thought that a group is like a reference to her and that she can go back to it anytime. I don't know.* – Qudwa5

From the mentees' side, they reported that the level of the relationship was enough for them. They were asked if they needed more commitment or another session with their mentors, and they were satisfied with what they had:

Honestly, it depends on the situation. In my case, it was enough. For example, if I didn't have a plan before my use of Qudwa and I had no idea what I wanted to do, I think I would need more help. In my case, as I said, I just needed to be sure of my decision, so when I talked to the mentor, she gave me input and inspired me. I felt that this was enough for me. I just needed to learn about someone successful in that field.* – Mentee17

Type of help

The mentors were careful to help the mentees think about their options and not to give them direct answers or guide them in a specific direction. Qudwa2 noticed that some of the mentees were seeking someone to make the decision for them and give them direct answers on what major was suitable for them:

I am very careful not to guide but to help think, I help to choose the right option for that person. I know that the goal was to encourage them into STEM, but I think the goal at the end is to help that girl to choose the right place for her.* – Qudwa8

Some of them were not convinced, some of them wanted specific answers. My first call was like that, she was very confused and wanted to learn more about the fields, but she wanted a specific answer on what field suited her the most. I don't know her. I gave her all the options and what is the basis of choosing each field. Then she can make the decision, no one else.* – Qudwa2

Three mentors did not expect the kind of questions and discussions they received from their mentees. Qudwa5 was faced with some family problems from her mentee and she clearly explained to the mentee that she could not interfere in any family-related issues. Qudwa14 and Qudwa2 expected different kinds of discussions, work-related discussions. Qudwa2 wanted them to ask her about the job itself, her role as an employee and what was required from her, and she stressed that "getting a job is not the end"*. On the other hand, Qudwa14 was surprised that the mentees were asking about the future of a person with a computer science degree, as she thought that this kind of knowledge should be known by now. She thought it was unexpected that the mentees had a lot of questions about university admissions and requirements. She believed that everything was on the internet.

Two mentees who expressed that they did not benefit from *Qudwa* in the way they expected had had different expectations than what *Qudwa* provided. Mentee29, who was a passive member of a group and then left the group, was expecting that she would be provided with some training or courses. Mentee21 said she left the group because she wanted someone to tell her what field suited her the most:

No, the reason is that I asked my questions and the mentor responded. Other than that, I wanted someone to guide me in which is best for me \dots My goal with *Qudwa* was to find someone to tell me based on my interests and my abilities that this field suits me.* – Mentee21

The intermediate person

Both mentors and mentees were asked about their opinion of my role as an intermediate person between the two parties. The mentors were thinking from a professional point of view

and how the role required time and effort from my side, while the mentees loved the human touch of *Qudwa*. Qudwa8 was thinking of the continuity of *Qudwa*, so she suggested turning *Qudwa* into an automated system to ensure its sustainability:

To sustain *Qudwa*, there is pressure on you because you handle every request. If there would be a way that you supervise and the two parties communicate directly.*

Qudwa5 suggested that mentors handle the contact requests themselves. As a web developer, Qudwa14 analysed the intermediate role. She added that people in Saudi Arabia prefer WhatsApp as an easy and fast method of communication:

Whatever you do, the system can do it too. I don't want to be rude, but your role can be replaced ... I'm looking at it from the standard ways in web applications, there are no humans in the middle, but there are support systems ... For us here in Saudi, I think people use WhatsApp as the fastest way of communication ... The standard is to have a support system and you send an email, but here they won't send an email if there is a WhatsApp number.*

Mentee17 assumed that the role of *Qudwa* would end after she got connected with the mentor, but she admired that the system cared about her. She added that: "It's nice to have a person to talk to, not a website"*. Mentee23 believed that the intermediate person role was very important in her case. She reported that she had submitted the form out of curiosity, but she got excited after the system got in touch with her. In addition, she thought that people prefer talking to humans rather than systems. Mentee13 was surprised that there was a person behind *Qudwa*, she thought it was an automated system.

Cultural factors

The interviews showed that the culture had an impact on the interaction of participants in different ways. There was an emphasis on talking and getting help from Saudi females. There appeared to be a cultural problem of recommunicating or following up with someone older or of a higher position.

Female Saudi mentors

The emphasis on the need for mentors to be Saudi and to be female occurred on several occasions. While talking to female strangers is not a widespread concept in Saudi, talking to a male stranger in most Saudi societies is not acceptable. Qudwa8 talked about one of her audio online meetings with a mentee where she sensed that the mentee was uncomfortable and had reservations during the call. Qudwa8 felt that the mentee got nervous and tense when some of her family members came around while she was on the call. The mentor highlighted that

maybe talking to a male mentor would make her even more uncomfortable or cause her some problems with her family.

Mentee23 on her submission request added that she preferred her mentor to be a female. She clarified why she said that by saying that it would be more comfortable for the girls. Mentee21 in her WhatsApp interview reported that she needed to feel comfortable knowing all the participants in the group were female. When I asked her to clarify, she responded through WhatsApp:

I want to be sure that all the participants in the group are female + they all have joined the group to benefit.*

On the other hand, the fact that the mentors were preferred to be Saudi stemmed from the need to find someone who lived in the same circumstances. Qudwa14 during her interview mentioned how she had looked for a female in the electrical engineering field to help and guide her. She highlighted that the career options for men were clear and known, but what about her options as a Saudi woman?

If there was *Qudwa* at that time I would definitely have used it and especially if she's a female! I would ask her what will we study? What do you do? Where can I get a job? ... for example, here in Saudi I know what an electrical engineer does when he graduates, he gets employed in factories and other places, but females? It would be nice to have a female from the same environment, with the same options, the same challenges.*

Mentee13 showed some doubts about the mentors. From the profile pictures of some of the mentors, she thought that they were not Saudi or that they were Saudis living abroad. She clarified that their experience would be different because she intended to study and work in Saudi Arabia, which is why she thought that she might not benefit from their experience.

Other cultural factors

It appeared that the cultural background of the participants affected their participation in different cases. Qudwa8 drew attention to the differences she recognised between the two mentees she had audio meetings with. She thought that the first mentee was more interactive with her during the call, while the second was quiet during the call and cautious. She believed that the cultural differences between the two mentees had an impact on their interactivity. The first mentee was from a more developed region in Saudi Arabia and the second mentee was from a small city. She explained:

I could tell from the backgrounds of the girls that there were some cultural differences. The first girl I talked to on the internet was from the eastern region, she was aiming to study at a private university, her English language

level and knowledge were excellent, she had an internship, so her culture or society level was different. The other girl had just graduated from school and was from Taif, she had some social reservations, you know.*

The interviews showed a cultural problem or issue: fear or embarrassment at communicating or following up with older individuals or those of a higher position than themselves. Qudwa8 mentioned that she forgot about her mentee and that her mentee did not follow up with her or remind her. Qudwa8 referenced this to the cultural problem of not following up with people or organisations because they think less of themselves just because they are younger or of a lower position. This issue happened twice to Qudwa8. Through the interview Qudwa8 rementioned this problem when she asked her mentee to add her on LinkedIn to provide her with courses, articles and organisations regarding the field she was interested in. She explained that she was willing to help and give her mentee the resources she needed if the mentee showed interest by taking one step towards her. Up to the time of the interview, she had not received any connection request from the mentee on LinkedIn.

Mentee19 was asked about how to develop *Qudwa* and she wanted the mentor to talk about her struggles and the difficulties she faced, give them advice and guidance. I told her that she had the chance to ask the mentor herself in the group and she said that she was busy with university interviews and applications and would ask her after she was done with them. Submitter1 did not submit a form to contact any mentor because she wanted to be ready before she talked to the mentor. She gave a sense of exaggerating the status of a professional. She described her previous attempts to get answers from professionals on social media platforms as failed attempts. Although she wanted to learn more about the field that she was interested in and she believed that *Qudwa* could help her, she thought that she needed to be prepared beforehand.

On the other hand, Mentee17 and Mentee23 were astonished by the way the mentors were talking and addressing them. Mentee17 had some reservations about talking to her mentor using audio because she thought that she was not good with people. After the call, she described her mentor as "very friendly and very nice"*. Mentee 23 was surprised that her mentor was talking to her as if they were of the same age.

Qudwa and mentors' daily activity

Three mentors who had repetition of questions in their groups were asked about how they felt regarding this situation. The three mentors reported that they were not interrupted or distracted by the use of WhatsApp during their day. They all stated that they answered questions and communicated with their mentees during their free time:

Honestly, it wasn't a big effort, it was just a period of time, that didn't affect my work time, even on WhatsApp. Whenever I felt I had free time, I replied to them.* – Qudwa2

Qudwa5 and Qudwa14 were not bothered by the repetition, but Qudwa2 was annoyed about the repetition because the mentees suddenly stopped interacting. She expressed that she would not mind repeating her answers if she had seen responses from the mentees in the group:

Yes, it bothered me, and what bothered me the most is that the interaction stopped. If there was any kind of interaction, then I wouldn't have a problem with repeating my answers, information or discussions, but they stopped the interaction and it felt like they reached a point where they wanted someone to decide for them. They suddenly stopped. I don't know what happened to them. Honestly, this is the thing that bothered me most.* – Qudwa2

Qudwa2 and Qudwa14 suggested sharing the answers to common questions asked by the mentees. Qudwa14 suggested that the answers would be written as a collaboration between *Qudwa*'s mentors. Qudwa2 proposed that any liked or favourited answer by the mentees could be shared on *Qudwa*.

7.3.9 Social media announcements

This section presents the analytics collected from the promotions on Twitter and Instagram, followed by an analysis and reflection.

Twitter

The Twitter account had a total of 352 followers. The promoted tweet which presented the animated video about *Qudwa* had a total of 51,074 impressions with 82 clicks on the website link (Figure 7-6). The video had 8,900 views, 6 retweets and 48 likes (Figure 7-7).

× Tweet Analytics	× Tweet Analytics
موقع @Qudwa_ مالية ثانوي او جامعية؟ بتلك ثانوية او جامعية؟ تمتاع مساعدة و توجيه من مختصات بميالات مختلفة؟ 	Detail expands 467 times people viewed the details about this Tweet
قدرة هي منصة تطوعية لساعدة الفتيات من جميع الاعمار على الاجابة على استقساراتهم عن التخصصات و مجالات العمل http://www.qudwa.me #STEM pic.twitter.com/Mt2ajtfihs جهامعة #STEM pic.twitter.com/Mt2ajtfihs	111 organic 356 promoted Media engagements number of clicks on your media counted across videos, vines, gifs, and images
Impressions 51,0 times people saw this Tweet on Twitter 2,780 organic 48,294 promoted	74 § 53 organic § 219 promoted Profile clicks number of clicks on your name, @handle, or profile photo 231 Profile clicks on your name, @handle, or profile photo Profile clicks on your name, @handle, or profile photo Profile clicks on your name, @handle, or profile photo Profile clicks on your name, @handle, or profile photo Profile clicks on your name, @handle, or profile photo Profile clicks on your name, @handle, or profile photo Profile clicks on your name, @handle, or profile photo
Media views all views (autoplay and click) of your media are counted across videos, vines, gifs, and images	4 52 organic 179 promoted Link clicks clicks on a URL or Card in this Tweet
532 organic 8,432 promoted Total engagements 1,12 times people interacted with this Tweet	Image: Solution of the second state
259 organic 891 promoted	0 organic 48 promoted

Figure 7-6 Twitter analytics



Figure 7-7 Number of views of the promoted tweet

Instagram

The Instagram account had 319 followers. The promotion on Instagram resulted in 1,828 profile visits and 1,554 website visits (Figure 7-8, left). The post was bookmarked 82 times and got 51 likes, and the video on the promoted post was viewed 9,153 views (Figure 7-8, right). Insights from Instagram provided details on the location of the audience, their age range and gender (Figure 7-9). The insights showed that most were from Riyadh (29%), followed by Jeddah (20%). Most were female (94%) with 45% of them in the age range 18–24 years (Figure 7-9).



Figure 7-8 Insights from Instagram on the number of visits and views



Figure 7-9 Insights about the audience on Instagram

Promotion on social media

The number of followers was almost the same on both platforms, the Twitter account with 352 followers and the Instagram account with 319 followers. The views of the promoted animation video on Twitter numbered 8,900, which is close to the number of views on Instagram which reached 9,153. The insights from both promotions showed that audiences on Instagram were more interactive with the promotion than on Twitter. The audience on Instagram who visited the account profile numbered 1,828, in comparison to the 231 profile clicks on Twitter. The number who visited the website from Instagram was 1,554 in comparison to 82 clicks on the link to the website on Twitter.

The post-study survey showed that most mentees who knew about *Qudwa* came from Twitter (43%) in comparison to 38% who knew about *Qudwa* from Instagram and 14% from WhatsApp. Most of them also suggested Twitter as a helpful method for promoting *Qudwa* (44%), while Instagram received 17% and WhatsApp received 31%.

Mentee5 in her interview, when asked about which channel was best for promoting *Qudwa*, mentioned that Instagram was not the right place to look for help regarding her studies:

I think Twitter because it is more suitable for *Qudwa*. I don't think I would go for Instagram to search for things like *Qudwa*, if I wanted something like this I would go to Twitter.*

The insights from the social media contradict the results of the survey. The insights showed more interaction from Instagram than from Twitter. Participants might have found Twitter a more suitable platform for a service like *Qudwa*, but their online availability or online presence on Instagram was greater than on Twitter, which may explain the higher interaction from the Instagram promotion than from the Twitter promotion.

7.4 Discussion

In this section, I present the evidence from the findings on how *Qudwa* accomplished the design requirements presented earlier in Chapter 6. I also discuss the automation and improvement of *Qudwa*. A more detailed and overall discussion will be presented in the next chapter.

7.4.1 Design goals

The findings have shown that *Qudwa* achieved the design goals. From the interviews and survey, it was noted that the flexibility and control features in *Qudwa* were highly appreciated and favoured by the mentees. Some mentees needed to be prepared before contacting a mentor, while others contacted more than one mentor. *Qudwa* being mentee-initiated helped in coping with the different personalities and needs of mentees. Giving the mentees the flexibility and control in their mentoring relationships showed that the majority of mentees (81%) were satisfied with the level of their relationship. In addition, the interviews showed that mentors felt the need for a stronger relationship, while mentees were contented and more comfortable with what they had experienced on *Qudwa*. On the other hand, the variety of personalities and needs of mentees also influenced in the decision on a communication channel. Personal preference was the basis of the selection of communication channel for some mentees. Some preferred textual communication for ease of use, due to their shyness or

lack of self-confidence and to feel less formal, while others showed that the selection of the communication channel depended on their needs. For specific and more personal help and guidance, they would choose a one-to-one audio call. For more general help, they would choose to be in a WhatsApp group. The findings showed that some mentees communicated with the same mentor through different channels.

The second design requirement was visibility. Visibility was evidenced in different aspects. The survey results showed that the majority of mentees (81%) stated that *Qudwa* had raised their awareness of female STEM professionals in Saudi Arabia and that 43% of the mentees visited the mentor's social media accounts and found them beneficial. The findings showed that mentees had shared their experience on Qudwa and their surprise at such mentors with their peers in general. They had also recommended *Qudwa* to specific individuals they knew, either because they believed that *Qudwa* would be beneficial for that specific person or they found a mentor who was specialised in that individual's field or a field that she wanted to study. This gives proof that the mentees had widened their knowledge and awareness of different mentors in different fields. While Qudwa intended to create visibility of the mentors among the mentees, the findings showed that mentees were pleased with the visibility of knowing that others of their age shared their concerns. The results showed that being in a group with others was beneficial for them because they believed that others might discuss things that did not cross their minds. Visibility also reached the mentors. Some browsed other mentors' profiles and visited their social media accounts. Others showed pride in being among such a group of mentors and would get in contact with other mentors.

Qudwa was designed to fit within the participants' daily internet activities. The survey results showed that the mentees did not have any problems with the communication channels. Ease of use was the most common reason why mentees chose their communication channel. Through the interviews, mentees and mentors described WhatsApp as an "easy" and "fast" method of communication. Mentors agreed that *Qudwa* was not a burden and that it fit well with their daily activity and work schedule.

The final design requirement was the trusted connection represented by the human presence in applying the Wizard of Oz concept. The reason behind this design requirement was to ensure that mentoring was conducted as it should be and to clear up any misunderstandings. The findings of the survey reported that almost all the mentees (90%) found the presence of an intermediate person helpful. The findings showed that the intermediate person in *Qudwa* was helpful and needed in different situations. As expected, some mentees needed to understand the concept of *Qudwa* and others needed direction in which mentors to choose, while others had different expectations. Some mentees liked the

idea of a caring person and indicated that the presence of a human who was in contact with them had encouraged them to continue their participation in *Qudwa* compared to a fully automated system.

7.4.2 Improvement of Qudwa

The design of *Qudwa* was dependent on a human who acted as the system. While the idea of a human presence was favoured by the mentees and fulfilled the design requirement of a trusted connection, the findings showed that some of the tasks were preferred to be automated or delegated to mentors. To ensure the sustainability of Qudwa, the mentors suggested some modifications including automated scheduling, transferring the process of handling contact requests to the mentors and collaboration between mentors to provide answers to commonly asked questions. These suggestions stemmed from the mentors putting themselves in my place and their desire to lower the burden on them and me (who acted as the system). On the other hand, 90% of the mentees were satisfied with the human presence and thought of it positively. It is reasonable to think that the lack of understanding by the mentees of the professionals' workload did not widen their thinking about the burden caused by handling some of the tasks manually. Manual scheduling and handling of each contact request were thought of by the mentors as a burden and time-consuming to me. If mentors handled their own requests, it would help them to get to know their mentees better by asking them questions before the e-mentoring session. Even though mentors were not bothered by the repetition of questions, some suggested that collaboration between *Oudwa* mentors could result in the answering of common questions. Others suggested that answers that were *liked* by the mentees could be collected on Qudwa. Collecting the commonly asked questions or favoured answers publicly would provide more time for mentees to schedule mentoring sessions with different issues to discuss than those provided in the Qudwa repository. In addition, it would reduce the pressure on the mentor to answer the repeated questions, which might result in boredom or a lack of enthusiasm in answering and interacting with their mentees. On the contrary, the mentees disagreed with a question-and-answer format of Qudwa. The findings of the survey reported that 81% showed disagreement with turning Qudwa into a Q&A website. This could be a reaction to their difficulty in finding someone who answered their questions adequately instead of reading blogs and informational websites or asking professionals who answered briefly or not at all.

An integration between what the mentors suggested and what the mentees wanted would result in keeping *Qudwa* as is in terms of the e-mentoring relationship and communication methods, along with adding a section for commonly asked general questions

to lessen the burden on mentors. Regarding the automation of *Qudwa*, some tasks will be automated in an improved version of *Qudwa*. These tasks include handling *Recommend a mentor requests*, scheduling meetings and emailing confirmations for the meetings. Going back to Table 7-1, this means that all the tasks that were designed to be completed by the system will be automated except for one task, which is handling mentees' requests. Mentors suggested that they could handle the requests of mentees themselves. In their opinion, it would help them learn more about their mentees. The human side of *Qudwa* represented by the admin will still be available since the mentees favoured the existence of a human on a website. The admin will help through the website and monitor interactions to make sure everything goes well.

7.5 Summary

In this chapter, the deployment and findings of *Qudwa* have been presented. The deployment, which lasted for two and a half months, showed variation in the findings in different aspects. There was variation between mentors and mentees regarding the preferred communication channel and the level of the mentoring relationship. There was variation as well between the mentees, which was affected by their personalities, their needs and their cultural background. The chapter concluded with how *Qudwa* will be improved. Some changes are clear and obvious, such as the automation of scheduling meetings and sending confirmation emails to participants. Other changes were suggested by the mentors, such as building a repository for frequently asked questions and moving the task of handling requests from the admin's responsibilities to the mentors' responsibilities. The main design of *Qudwa* will remain the same, based on un-platforming and short-term relationships between mentors and mentees.

Chapter 8 Discussion and Conclusion

8.1 Introduction and Overview

This thesis has presented how I explored e-mentoring for female teenagers and young women in Saudi Arabia with the purpose of encouraging them to seek careers in STEM fields and increase their awareness of Saudi female professionals in STEM. This research had three research questions: first, what is the potential for applying a STEM e-mentoring program for teenagers and young women in the Saudi context; second, what are the factors that contribute to designing a STEM e-mentoring system for teenagers and young women in the Saudi context; and, finally, how to design a STEM e-mentoring system for teenagers and young women in the Saudi context.

The absence of mentoring in the Saudi context and similar contexts directed the exploratory nature of this research. Four studies were conducted to answer the questions of the research. Chapter 4 described the pilot study (Study 1), which was a deployment of a structured STEM e-mentoring design. The cultural factors of Saudi Arabia were the main factors considered in this study. The pilot study aimed to explore the potential for applying a STEM e-mentoring program for teenagers and young women in the Saudi context. The results of the pilot study led to carrying out two co-design studies (Studies 2 & 3) presented in Chapter 5. These co-design studies aimed to identify the factors for designing a STEM ementoring for teenagers and young women in the Saudi context. Chapter 6 presented four design requirements: 1) flexibility and control; 2) visibility; 3) integration of e-mentoring with daily internet activity; and 4) a trusted connection between mentors and mentees. It described the design of *Qudwa*. *Qudwa* was designed based on the findings of the studies conducted in Chapter 4 and Chapter 5 and the lessons learned from previous work on structured ementoring. It was designed to give mentees and mentors more flexibility and control in their own e-mentoring relationships. Qudwa was designed to be embedded in their daily internet activity, instead of pulling them into another space, by using existing social networking platforms. Chapter 7 presented the deployment and evaluation of *Oudwa* and concluded with improvements to the design.

In this chapter, I revisit the research questions and reflect on the work conducted. Several discussion points have emerged from this research that future research should take into consideration. These discussion points reflect the Saudi context and the e-mentoring design. This research was context-specific, so findings related to Saudi Arabia and similar contexts were developed. The first section reflects on the first research question of this research, to examine the potential for e-mentoring in Saudi Arabia. The second section

revisits the following two research questions related to identifying the factors and designing of e-mentoring. It explains how the inclusion of co-design activities shaped the redesign of e-mentoring. It further discusses the idea of unplatforming and augmenting existing social media platforms. I talk about the evaluation and measurement of e-mentoring and report on the absence of mentors in the design process. Finally, I conclude with design implications, future work and limitations of this research.

8.2 The Saudi Context

One main aspect of this research was the context of Saudi Arabia. The first research question of this work was to explore the potential for applying a STEM e-mentoring program for teenagers and young women in the Saudi context. While the findings of this research showed that the design requirements were less affected by the cultural factors and more by the age group, several discussion points have emerged that address the potential and need for e-mentoring in Saudi Arabia and are worth further looking into.

8.2.1 The demand for mentoring

As discussed earlier, the lack of mentoring in general in Saudi Arabia is an issue that has been discussed rarely in the literature. In comparison to the developed work around mentoring and e-mentoring in other contexts such as the USA and Europe, few articles were found around traditional mentoring in Saudi Arabia (Abalkhail and Allan, 2015; Al Qahtani, 2015; Fallatah et al., 2018; Ghawji et al., 2017; Sattar et al., 2017; Shukri and Leil, 2017) and none were found around e-mentoring in Saudi Arabia. These articles pointed to the need to initiate traditional mentoring programs in Saudi Arabia in different fields. They argued that the success of mentoring programs and initiatives in other contexts provided a motive for replicating and developing mentoring programs and initiatives in Saudi Arabia. One main contribution of this work is exploring the requirements for and consequently conducting the first documented e-mentoring program in Saudi Arabia. Even though the work conducted in this research was on a small scale and for a limited period of time, the results of all four studies show potential and demand for the concept of mentoring.

It is important to highlight that the participants in Chapter 4 and Chapter 5 came from diverse backgrounds and yet they showed interest in and demand for e-mentoring. In Chapter 4, the findings showed that the need for mentoring was not limited to girls with parents who had low levels of education, but was also found in those with parents who specialised in STEM or were highly educated. In Chapter 5, the co-design workshops were conducted with two groups. The first group (Study 2) were undergraduates studying at community colleges

who came from low socioeconomic backgrounds and the second group (Study 3) were high school students from higher socioeconomic backgrounds as compared to the participants of the first group. The themes that emerged from the thematic analysis of data from Study 2 were not different from the themes that emerged from the data from Study 3. Participants from both studies, who came from different socioeconomic backgrounds, emphasised their need for mentoring. The holistic manner of thinking about e-mentoring exhibited by participants in Chapter 5 demonstrated high levels of maturity and interest in e-mentoring. This demand came also from visitors with different backgrounds requesting non-STEM fields and mentees of older age groups.

In Chapter 7, while the backgrounds of the participants were not collected, it was interesting that the visitors to *Qudwa* were looking for mentors in fields other than STEM fields. The findings showed that 17 *Let us help* submissions were requesting fields not supported by *Qudwa* such as finance, arts, translation, fashion design, law and business. Although *Qudwa* was designed to support and guide females who are graduating from high school (or younger) and undergraduates, to guide them through university applications and career paths, the findings showed that there were participants who were older. Older mentees sought knowledge about fields that they were not familiar with. This is consistent with the findings of previous work that reported that participants in Saudi Arabia had positive thoughts about mentoring and believed in the values and benefits of mentoring (Al Qahtani, 2015; Fallatah et al., 2018; Sattar et al., 2017).

The findings of this research reported different levels of impact on the mentees. In Chapter 4 and Chapter 7, after their participation in the studies, mentees showed higher levels of awareness of available STEM careers and professional Saudi women in STEM, and some mentees took action after their participation. Three years after Study 1, I followed up with some of the mentees and one mentee said that after she participated in the pilot study, she was reassured about the field she wanted to study and that the e-mentoring experience had widened her perception of career options. Another mentee who joined *Qudwa* needed to make a decision about changing her study major. While her decision was not only based on her talk with her mentor, it gave her reassurance that her decision was right for her. Therefore, it is fair to say that *Qudwa* (as discussed in Chapter 7) was a successful implementation of ementoring in Saudi Arabia in terms of increasing awareness and leading to action, even though this was not measured in terms of effectiveness and there was no long-term assessment.

Qudwa broadly confirms Sulphey and Allam's (2017) argument that the success of traditional mentoring programs in other countries will have the same impact and success in

Saudi Arabia. The lack of mentoring and e-mentoring in Saudi Arabia contributed to the demand and success of *Qudwa*. Literature around mentoring and e-mentoring in Saudi Arabia and similar context is scarce, and further explorations and investigations on these topics are needed to develop a full understanding of the area. Although this research aimed to tackle the issue of low participation of females in STEM in the context of Saudi Arabia, future research in this context should not be limited to STEM or to teenagers and young women, as the findings showed demand from different fields, different age groups and different purposes.

8.2.2 Lack of Arabic content on the internet

Instrumental/vocational support, psychological support, informational support and role modelling are the defined functions of a mentor (Ensher et al., 2003; Single and Single, 2005). The mentors in *Qudwa* showed their support through the different functions of a mentor. Mentors provided advice and guided the mentees through which study and career paths they could take. They also encouraged the mentees who found problems in making a choice. Mentors were role models to the mentees; they inspired and shared their previous experiences and challenges with them. What was surprising to the mentors was the kind of information needed by the mentees, which the mentors considered common knowledge and basic information. Mentors attributed this behaviour to the lack of Arabic content on the internet.

In a recent study examining information-seeking using the Saudi Digital Library (SDL), which is a multilingual digital library, it was reported that finding Arabic resources for academic purposes was difficult (Alsalmi, 2019). The participants were asked to perform Arabic and English search tasks, and the results showed that the participants expressed concerns and lack of trust and confidence in the Arabic results in comparison to the English results. They reported that the English search tasks provided more accurate resources, where in the Arabic search task they found challenges in finding relevant resources in Arabic. Due to frustration, some participants had ended their Arabic search task, which is a concerning result because these search tasks might be important for their academic purposes. The study concluded that the lack of Arabic content was one of the challenges experienced by the participants and suggested focusing on and paying more attention to Arabic content (Alsalmi, 2019). Another study in the UAE explored the barriers to using educational videos in the classroom in UAE schools. The study indicated that the lack of Arabic content on the web was one of the barriers (Alawani et al., 2016). Even though the kind of digital Arabic content in the studies of Alsalmi (2019) and Alawani et al. (2016) were different to the kind of

information that the mentees in this research were looking for, they all shared the same obstacle, which is the lack of Arabic content.

Arab internet users constitute 5% of the world's internet users (Abubaker et al., 2015), yet the Arabic content represents less than 1% of web content (Al-Kabi et al., 2018). It is considered to be low-quality content because its main source is inaccurate translations from other languages (Albalooshi et al., 2011) and around 33% of it is generated by social media users (Al-Kabi et al., 2018). In Chapter 7 (Study 4), the mentors suggested collaboration between *Qudwa*'s mentors to build a repository on *Qudwa* to provide answers by professionals to commonly asked questions. This suggestion is similar to the one proposed by Alawani et al. (2016), who suggested progressively creating an Arabic video repository for teachers and learners in the UAE and similar contexts that fit the culture and the requirements of the Ministry of Education.

This finding stressed the need for the creation of a variety of Arabic content on the web. Arab professionals and specialised individuals in all sectors and fields are encouraged to participate in the enrichment of trusted Arabic content on the internet. Another important reason to increase Arabic content on the internet is that online content is used to promote countries economically and socially (Abubaker et al., 2015). Abubaker et al. (2015) reported that one of the challenges that face Arab countries in creating Arabic content is the lack of specialised professionals, while the mentors in this research showed interest in voluntarily collaborating to create a professional repository. This presents a gap between Arab governments and regional initiatives and these professionals who are willing to help. Future research should be undertaken to investigate the scaffolding for creating and increasing the Arabic content on the web, which will be beneficial in mentoring, learning and promoting Arab countries.

8.2.3 Technology and culture

Technology use is dependent on culture because culture determines how individuals in one culture communicate with each other using technology (Pookulangara and Koesler, 2011). When designing for a specific context or culture, the norms and culture imply the factors and design implications for that context. Culture in HCI has gained its fair share of studies in the literature (Kyriakoullis and Zaphiris, 2016), with previous studies examining how culture influences the use and design of technology. Saudi Arabia is considered a collectivist country that is mainly affected by culture and religion (Al Lily, 2011; Al-Saggaf, 2011). At the beginning of this research, the cultural factors were the main concern in the design. In Chapter 4, the use of textual communication and the platform Edmodo were design choices to

maintain the privacy of the female participants. The findings showed that mentors and mentees agreed that the platform and the nature of the communication were the major barriers to their experience. The results also showed mothers' approval of the use of other communication methods, including audio and video communication methods, which are not commonly used among strangers in Saudi Arabia. Because of the unfamiliarity of mentoring in Saudi Arabia, parents might have concerns about their daughters talking to professionals they do not know and the involvement of parents was a design choice based on the nature of the culture. Although parents had the opportunity to participate, the results showed that none of the parents did or felt the need to do so. They believed that the rapid changes in technology make it hard for them to control what their children watch or use, so they rely on raising them well and trusting them. This change in the cultural norms of older generations (the mothers) accepting the use of new technologies is likely to be related to the high rate of internet and social media usage by the younger generation (the daughters) in Saudi Arabia, which has led to the rapid adoption of such technologies (Bafakih et al., 2016; Winder, 2014).

In Chapter 7, audio calls were used but none of the participants requested video calls, even though this was one of the available options, because in Saudi Arabia it is considered an uncommon method of communication between female strangers. In addition, Saudi females tend to not share or reveal their personal photos online (Al-Saggaf, 2011; Binsahl and Chang, 2012). WhatsApp was highly used on *Qudwa* and one of the mentors said that the reason is that Saudis use WhatsApp heavily even for professional communications and they find it the easiest way of communication. She also compared her work with non-Saudis and their use of email for professional communication. Technologies that were used on *Qudwa* were designed and used globally, and were not specifically designed for the Saudi culture. In *Qudwa* the cultural factors were relegated to the norms already established for using existing social media technologies. In other words, participants used social media technologies, in a way that was appropriate to their culture.

The design requirements in this research (which are discussed in more detail below) shifted from being culture-specific to being generation-specific. While culture in HCI literature is usually considered a barrier (Kyriakoullis and Zaphiris, 2016), an interesting area of research could look into how people in different contexts use globally designed technology in ways that do not conflict with their culture.

8.2.4 Why Saudi female mentors

The nature of Saudi Arabia has guided this research to develop same-gender e-mentoring. As mentioned earlier, segregation between men and women publicly and privately is one characteristic of Saudi society (Al Alhareth, 2013; Al Lily, 2011). Social interaction between men and women is inappropriate, with the exception of professional interaction (Al-Saggaf, 2016). Therefore, it would be unacceptable and a subject for criticism and rejection to create a space where teenagers and young women communicated with male strangers. The difficulties that Saudi women face are different than in other cultures, so presenting female Saudi professionals who have lived in the same circumstances and succeeded was essential to raise hopes and ambitions of the mentees. The literature has shown that same-gender mentoring for women is more successful (Leck et al., 2012). Packard mentioned that mentors who share characteristics with their mentees can be seen by their mentees as "possible selves" (Packard et al., 2004). In Chapter 4, the mentees were amazed by the achievements of their mentors and were also relieved that such fields and careers were available to Saudi women. One mentor mentioned the importance of presenting Saudis as mentors and increasing their visibility because, from her experience, she found that young Saudis had a conception that accomplishments and successes in STEM cannot be achieved in Saudi Arabia or by Saudis. All the mentors who participated in this research (Chapter 4 and Chapter 7) understood the struggles the mentees faced. Mentors explained their previous experiences when they were young and how they needed support and guidance. Mentees in Chapter 7 insisted on the importance of having a female Saudi mentor. They thought that they would feel more comfortable talking to a female and they wanted someone who lived in the same culture and circumstances. They believed that if the mentor had not been in the same circumstances, then they would not benefit from the relationship. One mentor experienced discomfort from her mentee when some member of her family entered the room while they were talking. The mentor reported that the mentee might get into trouble if she was talking to a male mentor.

Another aspect is that one main goal of this research was to create visibility for professional Saudi women in STEM. For women in Saudi Arabia and similar contexts, welldesigned and well-utilised e-mentoring has the potential to promote the social capital of both girls and female professionals. Lack of professional support for women in Saudi Arabia is recognised as a major barrier to the professional development of women and contributes to the underrepresentation of women in the workforce (Al-Ahmadi, 2011; Alomair, 2015). During the recruitment of mentors for this research, I faced difficulties in finding female Saudi STEM professionals, not because of a lack of such professionals, but because of the

lack of communities and networks of female Saudi professionals. Existing platforms such as LinkedIn helped me find professional women, but I still faced the problem of their lack of visibility. Mentors in Chapter 7 expressed their intention to contact other mentors through *Qudwa*, which may lead to professional mentoring relationships. Previous literature has stressed the need to understand the development of professional support networks and interventions for women in Saudi Arabia such as mentoring and networking (Alsubaie and Jones, 2017). Indeed, given the lack of female professional networks in Saudi Arabia, participation in appropriately configured e-mentoring could be a pillar on which such networks might be constructed.

Due to the lack of documented work on e-mentoring in Saudi Arabia, the first research question was to explore the potential for applying e-mentoring in Saudi Arabia. This research showed evidence of the potential e-mentoring has in Saudi Arabia for advocating STEM and showed the need for e-mentoring in other fields as well. It discussed the problem of lacking reliable Arabic online resources, which may be one additional factor in the need for ementoring in Saudi Arabia. While examining the potential for e-mentoring, the cultural norms of the Saudi context were not as important as expected, but the fact that the mentors were female and Saudi was important to increase their visibility and promote the social capital of both girls and female professionals.

8.3 The E-mentoring Design

In this section, I revisit the following two research questions. This section reflects on the factors that contributed to the design of a STEM e-mentoring system and how the design was created. It discusses the changes in the form and design requirements of e-mentoring that resulted from this research. It also reports on the enhancement that the central website (*Qudwa*) had over existing social media technologies. Furthermore, it discusses how the e-mentoring design can be evaluated and how the absence of mentors' input in the design process affected their e-mentoring experience.

8.3.1 Redesigning e-mentoring

This exploratory research started with expectations and assumptions that have proven to be of less importance. In the beginning, the cultural factors of Saudi Arabia were the main driver for the design of the pilot study (Study 1 in Chapter 4). As mentioned earlier, in Chapter 4 the privacy requirements of the female participants in the Saudi context, the selection of the mode of communication and parental involvement were the cultural factors that were thought of as requirements in designing e-mentoring for females in Saudi Arabia. The findings of Study 1

showed that these factors were not as important as expected. The participants preferred other means of communications and none of the parents participated in the study. In fact, the interviews and focus group that were conducted at the end of Study 1 presented different requirements than expected. This highlighted that the mentees desired a different style of e-mentoring that had more to do with their generation than Saudi culture. Co-design studies in Chapter 5 with the target group emphasised that the cultural norms of the Saudi context were less important in the design than the generation traits of the participants. The participants in Chapter 5 did not show concern about the cultural factors used in Study 1 or any other cultural factors. The format of e-mentoring that the participants desired is possibly linked to the communication preferences and online practices of their generation. This finding has shifted the design of e-mentoring from culture-specific to generation-specific.

Rowland (2012) in her work discussed the effect of generational differences on ementoring. She addressed the generational differences between mentors and mentees from a technology-use perspective. She argued that older generations (Generation X, born 1966– 1976) might not accommodate e-mentoring because they think online communications are impersonal and they are not considered to be technology savvy in comparison to Generations Y (born 1977–1994) and Z (born after 1994). Generations Y and Z are considered to have more knowledge about technology and to use it more than the previous generation. It is possible that challenges might occur when mentors from Generational differences existed but not in the technology use but in the design and structure of e-mentoring. The findings of Chapter 7 showed that the mentees were happy with the nature of the relationship; on the other hand, the mentors showed different views. The mentors wanted stronger and longer relationships with their mentees. They believed that they put their emotions and feelings into the relationships and wanted them to grow and last. This finding presents another generational difference between the mentees and the mentors.

Previous e-mentoring frameworks and models (Bierema and Merriam, 2002; Single and Muller, 2001) were built earlier in time for an older generation and utilising technologies that are less developed than current technologies. They attributed the success of the ementoring relationship to frequent, continuous contact between mentors and mentees. In Chapter 5, the design requirements extracted from the co-design workshops did not show any relation to regular, frequent meetings with mentors; in contrast, they expressed less committed relationships. In this research, a new form of e-mentoring has emerged that is different to the known structured form of e-mentoring. Possible explanations for this change might be the change in generations and the advancement of technology. These explanations are consistent

with what Williams et al. (2012) reported. They discussed that it is hard to construct a fully developed e-mentoring model due to the rapid advancement of technology. They suggested that practitioners and researchers in the future should be flexible in designing e-mentoring based on the development of all aspects involved in the process. Further studies on the topic of e-mentoring with other populations and different contexts are recommended. Co-design methods in this research have helped in understanding the reason behind the low participation of the mentees (in Chapter 4). They also helped in changing the form and requirements of e-mentoring from a structured e-mentoring form to a less structured form and from a design based on culture to a design based on age group traits.

While the design requirements were derived from the age group traits, it is possible that the context of Saudi Arabia had some effect on which of the traits were dominant. While the culture did not affect the technology design, its effects were more prominent in the lack of Arabic content, the lack of awareness of female Saudi professionals in STEM and the lack of professional networks for females in Saudi Arabia. As discussed earlier, professional women in Saudi Arabia have low visibility and lack professional female networks. This factor may have contributed to the need for the mentees to be connected with as many mentors as they wanted. It is possible that when conducting co-designing methods with the same age group and with the same purpose of advocating STEM but in different regions that have more visibility of professional females, other generation traits might appear and dominate the design requirements. I argue that e-mentoring is not a one-size-fits-all process and that further investigations of the inclusion of co-design methods in other settings of e-mentoring are recommended. The insights from the co-designing sessions (Chapter 5) have changed my perspective and enlightened me about the participants and e-mentoring in Saudi Arabia. My expectations about e-mentoring have been influenced by my experience of being a Saudi STEM professional who lived, studied and worked in Saudi Arabia. While I assumed that I have a good understanding of the culture, my assumptions and expectations have proven to be less important than other factors. The user-centred design approach adopted in this research gave me a new perspective about my culture and how Saudis interact with new experiences and technologies.

8.3.2 Supporting social media technologies

The increasing use of social media around the world and in Saudi Arabia specifically has brought people closer and eased communication. Logically, this means that youth should easily be able to find professionals to help and guide them. The findings presented in this research showed the contrary. In Chapter 7, the mentees expressed their struggle in finding

someone who would professionally answer their questions or guide them. They tried to reach out to professionals (on different social media platforms) who they thought they would be able to help them, but they were either ignored or only briefly answered by the professionals. While *Qudwa* was built on the concept of unplatforming, which means taking advantage of and utilising existing technologies, the findings showed that the sole existence of these technologies was not enough to provide what the individuals needed. The mentees reported that what made *Qudwa* special was not only the communication methods, but that they were confident they would find the answers they needed.

Oudwa was a central website that acted as a meeting point for mentors and mentees. It created a space for mentors who were willing to volunteer to help and guide, and created a trusted space for mentees to find those mentors through technologies they were already using. In the recruitment process for Study 1 and Study 4, I faced problems in finding those mentors due to the lack of professional networks for women in Saudi Arabia. I relied mainly on LinkedIn and my personal contacts. Even as a professional, many of my contact attempts on LinkedIn were ignored. The mentors who agreed to participate showed enthusiasm to help and participate. They valued the opportunity given to them. They had previously provided help and guidance to young people, but their help was limited to their surroundings and people they knew. They also showed their support to people who reached them through social media platforms. The chance for young women who are seeking advice or guidance to know which of the randomly available professionals on social media platforms are willing to help is very low. Therefore, Qudwa as a central website had a major role in providing a pool of professionals who were willing to mentor and to give the opportunity to those professionals to widen their help to a wider group of teenagers and young women. Other studies reported that mentees described having their mentors in one place as "directly accessible" (Savoy, 2013).

Lambton-Howard et al. (2020) referred to the use of external technologies (e.g. central websites) beyond the boundaries of the employed social media technologies as "hard augmentation". Hard augmentation is used to overcome the limitations of social media technologies. Lambton-Howard et al. (2020) described two case studies where hard augmentation was applied. In WhatFutures (Lambton-Howard et al., 2019), an external website was set up to overcome the limitation of WhatsApp groups not being able to view each other's activity. The administrators of WhatFutures uploaded the activity of each WhatsApp group on the leader board of the website to increase motivation and engagement between groups. In Online UWC (Celina et al., 2018, 2016), which used Facebook, Google Hangouts and Google+ for messaging and communication, a central website was designed to facilitate scheduling between participants. Similarly, in *Qudwa* the hard augmentation or the

use of a central website enabled finding mentors who were willing to help, created visibility of female STEM professionals and was used for scheduling. *Qudwa* further supports the idea proposed in Lambton-Howard et al. (2020) that future research should explore how hard augmentation can be used as an enhancement and improvement that supports existing social media technologies.

8.3.3 Effectiveness of e-mentoring

One of the goals of this research was to raise awareness of existing female Saudi professionals in STEM and guide the mentees in their choice of STEM field. The pilot study in Chapter 4, although it was on a small scale and had low interaction levels, showed positive outcomes for both the mentors and the mentees. Chapter 4 reported that low participation numbers did not translate to lesser effectiveness or impact. The quantitative data for interaction frequency and duration did not meet the expectations. The chapter also reported that high levels of participation do not inevitably mean that there is interaction between the mentors and the mentees, as some mentors were writing several posts without receiving any feedback or interaction from their mentees. On the other hand, mentees and mentors showed some level of impact. The mentees reported that their level of awareness of female Saudi professional in STEM had risen and some mentees reported that they did not know that such careers existed for women in Saudi Arabia. Both mentors and mentees shared their positive experiences with the people around them. One mentee reported that her e-mentoring experience and connection with the mentor had increased her love of science. Another expressed that her participation made her more reassured about her choice and gave her more options about her future.

Chapter 7 showed the same results of a higher level of awareness, positive feedback and actions taken by the mentees after their participation. Even though mentors felt the vagueness of their impact, the interviews with mentees showed a different level of impact. For example, one mentee changed her study major after being in contact with two mentors. Other mentees mentioned that they were seeking certain information and their participation resulted in increasing their knowledge about it. Peer influence was highly noted; mentees shared their experience with their peers and also recommended *Qudwa* to them. The impact reached visitors to *Qudwa* as well. One visitor reported that she did not submit a contact request, but browsed the mentors and was surprised by their majors and qualifications. The mentors had their share of positive impact. Some mentors showed their pride in the range of mentors in different fields. They also browsed the social media accounts of those mentors. Other mentors

expressed the personal benefits *Qudwa* provided for them. Their participation as mentors satisfied their need to give and added more value to their lives.

Previous research reported that the number of sessions between a mentor and a mentee is important for effective e-mentoring (Bierema and Merriam, 2002; Single and Muller, 2001). Some programs faced problems in maintaining e-mentoring relationships and suffered from low interaction, but these studies showed positive outcomes and appreciation from the mentees (Lee and Noh, 2003; Savoy, 2013; Stoeger et al., 2013). The aim of this research was to establish further understanding of e-mentoring in Saudi Arabia due to the lack of previous work on e-mentoring in the context, so this research did not aim to measure the effectiveness of e-mentoring specifically. Although the numbers of visits to websites, interactions and sessions held can be tangible measurements of the effectiveness of e-mentoring, other different forms of intangible impact are also worth considering to measure the effectiveness of e-mentoring including peer sharing, recommendations and personal and professional impacts on mentors.

8.3.4 Mentors' perspective

This study has been limited by the inclusion of mentees only in the design process and therefore eliciting the design requirements from the mentees' side. Although mentors' experiences with e-mentoring in Chapter 4 (the pilot study) and Chapter 7 (Qudwa) were taken into consideration and used for evaluation purposes, mentors were not included in the design process. In Chapter 4, the mentors acknowledged the low interaction between them and the mentees, yet some mentors tried to establish a personal connection and others tried to engage them through activities. In Chapter 5, the participants recognised the mentors' efforts and appreciated their valuable time. Several suggestions were made in the co-design workshops that addressed the mentors. Some suggested giving feedback or evaluations, others gave control to mentors in the case of mentees missing their scheduled sessions and some shared the accomplishments a mentee had made with the help of her mentor. The findings in Chapter 7 (Qudwa) showed that the mentors desired a more connected relationship, as opposed to the less committed relationship wanted by the mentees. They also provided valuable input on how to improve Qudwa and reduce pressure and repetition for the mentors. The findings of *Oudwa* showed that mentors gained personal benefits from their participation as mentors. It satisfied their needs to give and help, made them feel better about themselves and created a positive change in their lives. Although mentoring in general is considered a mutually beneficial relationship, previous literature has focused more on the outcomes for the mentees and less on the outcomes for the mentors (Beltman and Schaeben, 2012; Williams et

al., 2012). Consistent with the literature, this research suggests that more effort is needed to address and enhance the outcomes of e-mentoring for mentors. Therefore, there is a need to include mentors as key stakeholders in co-design workshops for e-mentoring to develop a design that satisfies both parties.

Two research questions of this research were to identify the factors that would contribute to an e-mentoring design and the creation of a design based on these identified factors. The co-design methods helped in understanding the requirements of the mentees, who were from a different generation. The redesign of e-mentoring based on the identified factors resulted in *Qudwa*. The un-platforming design of *Qudwa* shows the potential of existing technologies and social media platforms to facilitate a new form of e-mentoring. This research shows how to evaluate e-mentoring from different perspectives. It also shows the importance of mentors' input in the evaluation and design processes.

8.4 Design Implications

Channel and medium of communication: Textual communications were used in Chapter 4 and Chapter 7 but on different platforms; in Chapter 4 Edmodo was used and in Chapter 7 the textual method was WhatsApp, a channel that mentors and mentees were already using. The findings of Chapter 4 reported that the communication method (i.e. text-only) was a barrier, as participants said that they wanted a method that would help them to easily express themselves, such as audio and video, while in Chapter 7 that same textual communication method was favoured and highly requested. As both methods are textual in nature, this implies that, while having flexibility in the type of media used for communication is important (see below), the findings from Chapter 7 show that the communication channels (i.e. the tools used), and their familiarity and level of use, are even more important than the nature of the communication media.

Flexibility: Another design implication related to communication channels is providing flexible e-mentoring relationships (one-to-one e-mentoring vs. group e-mentoring, textual vs. audio and on-when-needed vs. scheduled) to accommodate mentees' different personalities and preferences. In Chapter 4 only one method of communication was available and the mentees were grouped together, which might have increased the sense of the communication method as a barrier to their relationships. In Chapter 7, some mentees joined the WhatsApp group of a mentor and requested a one-to-one audio call with the same mentor. They reported that the audio calls were needed to discuss personal matters that they preferred not to share with others, while the WhatsApp groups were for more general matters and to gain benefits

from the discussions with others. It is not clear which factor was more important to them, the method of communication being audio or that they preferred a one-to-one session. Although it was not an available option, the findings did not show any mentee who requested a one-to-one conversation with a mentor through WhatsApp or mentees who contacted their mentors privately through WhatsApp. The findings showed mentees with different personalities and preferences. The findings also presented how some mentees changed their requests several times; some changed the medium of communication and others who sent multiple requests to contact more than one mentor cancelled some of their requests. Some mentees preferred WhatsApp and others preferred audio calls. Some mentees were shy and lacked confidence, or just did not know what to expect. One mentee said that she thought she "is not good with people" and she was hesitant to have an audio call, but she felt more confident by the end of the audio call. Therefore, providing such flexibility is an important design recommendation to help mentees initiate e-mentoring relationships that fit their needs.

Raising awareness: Another design implication is to create more visibility of female STEM professionals and raise awareness of career options by highlighting mentors' qualifications and work experience. The findings of Chapter 4 and Chapter 7 showed that the mentees did not expect females to work in such fields or careers. In Chapter 4, mentees and mentors shared their experiences with e-mentoring and their knowledge about female Saudi STEM professionals with people around them. In Chapter 7, some mentors visited other mentors' pages and showed an intention to be in contact if their career paths crossed. This visibility was not limited to the mentees and mentors, but also reached the visitors to *Qudwa*. In Chapter 7, one submitter (who only submitted a form but did not get in contact with any of the mentors) reported that she was amazed by the qualifications and achievements of the mentors. By publicising the group of participating mentors, their qualifications and experiences, the mentees were exposed to the variety of STEM career options available to them.

Use of mainstream communication tools augmented with a central website: This research provides evidence that to support e-mentoring, it is not required or even desirable to develop a standalone or bespoke system. Utilising existing technologies and social media platforms is a key factor in the success of such systems, as demonstrated by the co-design workshops, deployment and evaluation of *Qudwa*. Utilising social media platforms that were embedded in the daily internet activity of the mentors and mentees eased the process of e-mentoring. A significant factor was that e-mentoring was initiated in a space that mentors and mentees already occupied. These existing platforms were augmented by a central website to achieve

the goals of this research. The idea of un-platforming is supported by the work of Lambton-Howard et al. (2020). The findings of the two separate research projects in Open Lab, despite their different aims, led to the same conclusion of augmenting existing social media platforms with a central website.

Adopting a co-design approach: Co-design workshops with the target group had a significant effect on the design. In Chapter 5, the co-design methods used were chosen to give the participants power and agency to design their own e-mentoring experience. The findings reported the importance of including the mentors as well, as they are considered a key factor in the success of e-mentoring. E-mentoring is a mutually beneficial relationship between a mentor and a mentee; therefore, it is important to address the needs and requirements of both parties to balance the design for a more effective e-mentoring relationship. This research suggests the inclusion of all stakeholders in an e-mentoring relationship in the design process by using appropriate co-design methods for each party.

Recruitment: The diffusion of social media in everyday life, especially with the youth population, presents opportunities for influencing, recruiting and targeting such populations. In this research, Facebook was eliminated as an option for recruitment due to its low use in Saudi Arabia and Twitter was thought of as an appropriate platform for promoting an educational/career-related topic. In Chapter 7, due to the low recruitment numbers from Twitter at the beginning of Study 4, Instagram was used as an additional method for recruitment. The insights from the ads showed that Instagram had more impressions, clicks and interactions than Twitter. The number of visits to *Qudwa* from Instagram was 1,554 in comparison to 82 clicks from Twitter. The post-study survey showed that 44% of the mentees thought that Twitter would be a suitable platform for promoting *Qudwa* and only 17% thought that Instagram was a suitable platform for promotion. The insights provided by the social media contradict the results of the survey. The insights showed more interaction from Instagram than from Twitter. These contradictions may be a result of the mentees perceiving Twitter as a more appropriate platform for a service like *Qudwa*, but the online availability of youth on other social media platforms such as Instagram made Qudwa more visible and accessible to them. The mentees reported that 43% of them knew about *Qudwa* from Twitter and 38% from Instagram; while there is no significant difference between these numbers, the Instagram ad insights showed higher numbers of visits and interactions, which may have contributed to raising the awareness of the visitors about female Saudi professionals in STEM just by visiting Qudwa (as discussed earlier). It is recommended to use social media platforms for youth research recruitment and examine which platforms are more appealing and would

result in more recruitment.

8.5 Limitations and Future Work

As mentioned earlier, an issue that was not addressed in this research was the measurement of effectiveness in e-mentoring. This research has presented intangible positive effects of e-mentoring on both mentors and mentees that are worth considering in future research when measuring the effectiveness of an e-mentoring design. Another limitation was that mentors were not included in the design process. The inclusion of mentors in the design process would help in designing e-mentoring that is balanced between the needs of the mentors and the needs of the mentees.

It was noted from Chapter 4 and Chapter 7 that there was a misunderstanding on the mentees' side about what mentoring means, due to the absence of the concept of mentoring in Saudi culture. Although the research purposes and concepts were described for the mentees in Chapter 4 and an additional animated video was used in Chapter 7 to promote *Qudwa*, the findings showed that some mentees had different expectations. For example, some mentees thought that *Qudwa* would provide courses or explicit answers on what major to study or suit them. This research acknowledges the absence of mentoring in Saudi Arabia, yet introducing the concept of mentoring to the participants was not addressed. Future research should look into how to introduce a new concept to such a culture to increase its effectiveness and use.

In Chapter 5, two different groups from different socio-economic backgrounds were included in the design process, and both groups showed similar requirements for an e-mentoring design, yet this research is considered of a small scale and did not cover the different backgrounds of the Saudi society. Conducting larger-scale research would result in gaining more understanding about the requirements and design of an e-mentoring system in the context of Saudi Arabia.

In Chapter 7, I faced the problem of mentees who did not answer the survey, ignored my interview requests or suddenly stopped responding to interview questions. This limitation lessened my understanding of the mentees' experience with *Qudwa*. Future work should look into how to engage mentees in the evaluation process considering their age group. Furthermore, *Qudwa* presented mentees with different personalities, different understanding of e-mentoring and different interactions with their mentors, conducting larger-scale research would result in developing a better understanding of how different participants with different backgrounds and personalities would interact in *Qudwa*.

Additional future design work on *Qudwa* was discussed in Chapter 7. This future work included the automation of some processes on *Qudwa*, such as scheduling, handling of

contact requests by the mentors and collaboration between mentors to provide answers to commonly asked questions. These to-be-automated design recommendations do not imply the elimination of the human presence, which was a design requirement that was favoured by the mentees in creating a trusted connection between mentors and mentees. The *Qudwa* feature *Introduction through social media* did not work as expected due to privacy controls on the social media platforms. In future work, I would test the need for this feature from the mentees' perspective and try to overcome the barriers faced. Mentees made several suggestions of other social media platforms such as Snapchat and Telegram, which suggests examining the appropriateness of using these platforms on *Qudwa*, especially to conduct audio calls, as some mentees thought that the used platform (appear.in) was not a platform that they commonly used.

8.6 Conclusion

This research has presented investigations that addressed young women, the Saudi context and the design requirements of e-mentoring in such a context through a number of engagements with Saudi women (mentees and mentors). This work succeeded in addressing the three research questions. Four studies were conducted to address the research questions. The first study showed the applicability and need for e-mentoring in Saudi Arabia which was the concern of the first research question. The second and third studies addressed the second research question. They helped in gaining insights and collecting requirements to design an ementoring system. The final research question was fulfilled through the design and deployment of *Qudwa*. This research has concluded with several recommendations for future research and, despite its small scale, has demonstrated the need for mentoring and ementoring in STEM and other fields, and resulted in a successful deployment in the Saudi context, the last phase of which was the *Qudwa* e-mentoring system, which was developed based on the design requirements identified through earlier studies.

This research has also brought into focus the problem of the lack of reliable Arabic online resources, which may have contributed further to the need for mentors. It highlights the importance of having female Saudi mentors to mentor female Saudi teenagers and young women, because they are similar in the sense of living in the same culture and facing the same challenges. One of the most significant findings of this research is the role that the co-design workshops played in redesigning e-mentoring based on the traits of the generation, rather than the cultural factors. This research has identified the need for co-design as a method to support the design of e-mentoring in Saudi Arabia and particularly for this generation of future women professionals. The two co-design studies concluded with factors and design

recommendations to implement and evaluate a design concept driven by these co-design studies. The suggestion to include mentors in the design process has been presented to develop a design more suited to both mentors and mentees.

Following the research questions, three contributions are recognised. First, this is the first documented systematic investigation of e-mentoring in Saudi Arabia. The lack of previous examples applied in the same context pointed to the need to understand the cultural norms that would affect the application of e-mentoring in Saudi Arabia. While I aimed to understand the role of cultural norms, the studies demonstrated that the findings were affected less by cultural norms and more by the traits of the participants' generation. This research proves the applicability and success of e-mentoring in Saudi Arabia for teenagers and young women, and suggests the application of e-mentoring in other fields and settings in Saudi Arabia and the Arab region.

Second, the application of co-design methods helped in addressing the gap in understanding the needs and opportunities for designing e-mentoring. This research presents new insights into how co-design methods are beneficial in designing e-mentoring for young people. The results point to the need for a redesign of e-mentoring models for younger generations with less committed and more flexible relationships. Other forms of e-mentoring could emerge from applying co-design methods with different purposes and different populations.

The final contribution is the design, deployment and evaluation of a new form of ementoring. *Qudwa* is an e-mentoring design that is based on facilitating e-mentoring through existing social media technologies and practices (un-platforming). This contribution shows the potential of existing technologies and social media platforms to facilitate a new form of ementoring. This research directs future research to further investigate e-mentoring in the context of Saudi Arabia and similar contexts, to further explore the possibilities of other designs of e-mentoring in other contexts by applying co-design methods and to examine different applications of un-platforming.

References

- Abalkhail, J.M., Allan, B., 2015. Women's career advancement: Mentoring and networking in Saudi Arabia and the UK. Human Resource Development International 18, 153–168. https://doi.org/10.1080/13678868.2015.1026548
- Abras, C., Maloney-Krichmar, D., Preece, J. and Bainbridge, W., 2004. Encyclopedia of human-computer interaction. Thousand Oaks: Sage Publications, 37, pp.445-456.
- Abubaker, H., Salah, K., Al-Muhairi, H., Bentiba, A., 2015. Digital Arabic content: Challenges and opportunities, in: 2015 International Conference on Information and Communication Technology Research (ICTRC). Presented at the 2015 International Conference on Information and Communication Technology Research (ICTRC), pp. 330–333. https://doi.org/10.1109/ICTRC.2015.7156489
- Adams, A., Lunt, P., Cairns, P., 2008. A qualititative approach to HCI research, in: Cairns, P., Cox, A. (Eds.), Research Methods for Human–Computer Interaction. Cambridge University Press, Cambridge, UK, pp. 138–157.
- Adams, C.T., Hemingway, C.A., 2014. What does online mentorship of secondary science students look like? BioScience biu147. https://doi.org/10.1093/biosci/biu147
- Akram, M.S., Albalawi, W., 2016. Youths' social media adoption: Theoretical model and empirical evidence. International Journal of Business and Management 11, 22. https://doi.org/10.5539/ijbm.v11n2p22
- Al Alhareth, Y., 2013. E-learning contribution to the enhancement of higher education opportunities for women in Saudi Arabia (Pilot study). US–China Education Review A 3, 637–648.
- Al Lily, A.E.A., 2011. On line and under veil: Technology-facilitated communication and Saudi female experience within academia. Technology in Society 33, 119–127. https://doi.org/10.1016/j.techsoc.2011.03.005
- Al Qahtani, S., 2015. Students' knowledge of, and attitudes toward, mentoring: A case study at the Master's Program in Health and Hospital Administration. Adv Med Educ Pract 6, 149–152. https://doi.org/10.2147/AMEP.S67804
- Al-Ahmadi, H., 2011. Challenges facing women leaders in Saudi Arabia. Human Resource Development International 14, 149–166. https://doi.org/10.1080/13678868.2011.558311

- Al-Asfour, A., Tlaiss, H.A., Khan, S.A., Rajasekar, J., 2017. Saudi women's work challenges and barriers to career advancement. Career Dev Int 22, 184–199. https://doi.org/10.1108/CDI-11-2016-0200
- Alawani, A., Senteni, A., Singh, A.D., 2016. A video platform to produce, support, and share educational Arabic content (Working paper). Commonwealth of Learning (COL) and Open University Malaysia (OUM).
- Albalooshi, N., Mohamed, N., Al-Jaroodi, J., 2011. The challenges of Arabic language use on the Internet, in: 2011 International Conference for Internet Technology and Secured Transactions. Presented at the 2011 International Conference for Internet Technology and Secured Transactions, pp. 378–382.
- Al-Ghaith, W., 2015. Applying the technology acceptance model to understand social networking sites (SNS) usage: Impact of perceived social capital. International Journal of Computer Science and Information Technology 7, 105–117.
- Al-Jabri, I.M., Sohail, M.S., Ndubisi, N.O., 2015. Understanding the usage of global social networking sites by Arabs through the lens of uses and gratifications theory. Journal of Service Management 26, 662–680. https://doi.org/10.1108/JOSM-01-2015-0037
- Al-Kabi, M., Alsmadi, I., Khasawneh, R., Wahsheh, H., 2018. Evaluating social context in Arabic. Opinion Mining 15, 9.
- Al-Kathiri, F., 2015. Beyond the classroom walls: Edmodo in Saudi secondary school EFL instruction, attitudes and challenges. English Language Teaching 8, 189–204.
- Al-Mutairi, S.H., Al-Jummah, R.J., Bou-Mikael, S., Al-Aruri, A.D., Raju, V.V.S., Izzett, P., 2015. Developing people through mentoring to transfer knowledge and technology.
 Presented at the SPE Middle East Oil & Gas Show and Conference, Society of Petroleum Engineers. https://doi.org/10.2118/172664-MS
- Alomair, M.O., 2015. Female leadership capacity and effectiveness: A critical analysis of the literature on higher education in Saudi Arabia. International Journal of Higher Education 4, 81–93.
- Al-Saggaf, Y., 2016. An exploratory study of attitudes towards privacy in social media and the threat of blackmail: The views of a group of Saudi women. The Electronic Journal of Information Systems in Developing Countries 75, 1–16. https://doi.org/10.1002/j.1681-4835.2016.tb00549.x

- Al-Saggaf, Y., 2011. Saudi females on Facebook: An ethnographic study. International Journal of Emerging Technologies and Society; Hawthorn 9, 1–19.
- Al-Said, K.M., 2015. Students' perceptions of Edmodo and mobile learning and their real barriers towards them. Turkish Online Journal of Educational Technology – TOJET 14, 167–180.
- Alsalmi, H.M., 2019. Information-seeking in multilingual digital libraries: Comparative case studies of five university students. Library Hi Tech ahead-of-print. https://doi.org/10.1108/LHT-06-2019-0119
- Alshahrani, H.A., 2016. A brief history of the internet in Saudi Arabia. TechTrends 60, 19– 20. https://doi.org/10.1007/s11528-015-0012-5
- Alsubaie, A., Jones, K., 2017. An overview of the current state of women's leadership in higher education in Saudi Arabia and a proposal for future research directions. Administrative Sciences 7, 36. https://doi.org/10.3390/admsci7040036
- Alwagait, E., Shahzad, B., Alim, S., 2015. Impact of social media usage on students' academic performance in Saudi Arabia. Computers in Human Behavior, Computing for Human Learning, Behaviour and Collaboration in the Social and Mobile Networks Era 51, 1092–1097. https://doi.org/10.1016/j.chb.2014.09.028
- Alwazzan, L., Rees, C.E., 2016. Women in medical education: Views and experiences from the Kingdom of Saudi Arabia. Medical Education 50, 852–865. https://doi.org/10.1111/medu.12988
- Ashktorab, Z., Vitak, J., 2016. Designing cyberbullying mitigation and prevention solutions through participatory design with teenagers, in: Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems, CHI '16. ACM, New York, NY, USA, pp. 3895–3905. https://doi.org/10.1145/2858036.2858548
- Bacikova, M., 2014. Using social networks for supervising and mentoring, in: 2014 IEEE
 12th International Conference on Emerging eLearning Technologies and Applications (ICETA). Presented at the 2014 IEEE 12th International Conference on Emerging
 eLearning Technologies and Applications (ICETA), pp. 17–22.
 https://doi.org/10.1109/ICETA.2014.7107542
- Bafakih, A., Othman, M., Dhillon, J.S., 2016. A survey on internet use by middle and high school students in Kingdom of Saudi Arabia.

- Beede, D.N., Julian, T.A., Langdon, D., McKittrick, G., Khan, B., Doms, M.E., 2011. Women in STEM: A gender gap to innovation (SSRN Scholarly Paper No. ID 1964782).Social Science Research Network, Rochester, NY.
- Beltman, S., Schaeben, M., 2012. Institution-wide peer mentoring: Benefits for mentors. The International Journal of the First Year in Higher Education 3, 33–44. http://hdl.handle.net/20.500.11937/13607
- Bencsik, A., Horváth-Csikós, G., Juhász, T., 2016. Y and Z Generations at workplaces. Journal of Competitiveness 8. https://doi.org/10.7441/joc.2016.03.06
- Bencsik, A., Machova, R., 2016. Knowledge sharing problems from the viewpoint of intergeneration management, in: ICMLG2016-4th International Conference on Management, Leadership and Governance: ICMLG2016. Academic Conferences and publishing limited, p. 42.
- Bernstein, B., 2003. Class, codes and control: The structuring of pedagogic discourse. Psychology Press.
- Bers, M.U., Cantrell, K.A., 2012. Virtual worlds for children with medical conditions:
 Experiences for promoting positive youth development. E-Health Communities and
 Online Self-Help Groups: Applications and Usage 1–23. https://doi.org/10.4018/9781-60960-866-8.ch001
- Bierema, L.L., Merriam, S.B., 2002. E-mentoring: Using computer mediated communication to enhance the mentoring process. Innovative Higher Education 26, 211–227. https://doi.org/10.1023/A:1017921023103
- Binsahl, H., Chang, S., 2012. International Saudi Female students in Australia and social networking sites: What are the motivations and barriers to communication, in: ISANA International Academy Association Conference.
- Blackwell, J.E., 1989. Mentoring: An action strategy for increasing minority faculty. Academe 75, 8–14.
- Blandford, A., Furniss, D., Makri, S., 2016. Qualitative HCI research: Going behind the scenes. Synthesis Lectures on Human-Centered Informatics 9, 1–115. https://doi.org/10.2200/S00706ED1V01Y201602HCI034
- Braun, V., Clarke, V., 2006. Using thematic analysis in psychology. Qualitative Research in Psychology 3, 77–101. https://doi.org/10.1191/1478088706qp063oa
- Cambridge Dictionary, n.d. . Cambridge Dictionary | English Dictionary, Translations & Thesaurus. URL https://dictionary.cambridge.org/ (accessed 1.12.20).
- Cantrell, K., Fischer, A., Bouzaher, A., Bers, M., 2010. The role of e-mentorship in a virtual world for youth transplant recipients. Journal of Pediatric Oncology Nursing 27, 344– 355. https://doi.org/10.1177/1043454210372617
- Celina, H., Kharrufa, A., Preston, A., Comber, R., Olivier, P., 2016. SOLE meets MOOC: Designing infrastructure for online self-organised learning with a social mission, in: Proceedings of the 2016 ACM Conference on Designing Interactive Systems, DIS '16. ACM, New York, NY, USA, pp. 484–496. https://doi.org/10.1145/2901790.2901848
- Celina, H., Lee, C., Olivier, P., Kharrufa, A., 2018. LearningCircle.io: Lessons learned from organising courses with and without a dedicated platform. Presented at the EdMedia + Innovate Learning, Association for the Advancement of Computing in Education (AACE), pp. 76–81.
- Crisp, G., Cruz, I., 2009. Mentoring college students: A critical review of the literature between 1990 and 2007. Res High Educ 50, 525–545. https://doi.org/10.1007/s11162-009-9130-2
- Dahalan, N., Omar, H.H., Atan, H., 2011. Mentor online: Students' social ability and their openness to e-mentoring. Malaysian Journal of Distance Education 13, 1–20.
- Danish, A.Y., Smith, H.L., 2012. Female entrepreneurship in Saudi Arabia: Opportunities and challenges. International Journal of Gender and Entrepreneurship 4, 216–235. https://doi.org/10.1108/17566261211264136
- Dawson, P., 2014. Beyond a definition: Toward a framework for designing and specifying mentoring models. Educational Researcher 43, 137–145. https://doi.org/10.3102/0013189X14528751
- DiRenzo, M.S., Weer, C.H., Linnehan, F., 2013. Protégé career aspirations: The influence of formal e-mentor networks and family-based role models. Journal of Vocational Behavior 83, 41–50. https://doi.org/10.1016/j.jvb.2013.02.007
- Dorner, H., 2012. Effects of online mentoring in computer-supported collaborative learning environments: Mentor presence and cognitive engagement. American Journal of Distance Education 26, 157–171. https://doi.org/10.1080/08923647.2012.692265

- Dorner, H., Kumar, S., 2016. Online collaborative mentoring for technology integration in pre-service teacher education. TechTrends 60, 48–55. https://doi.org/10.1007/s11528-015-0016-1
- DuBois, D.L., Holloway, B.E., Valentine, J.C., Cooper, H., 2002. Effectiveness of mentoring programs for youth: A meta-analytic review. Am J Community Psychol 30, 157–197. https://doi.org/10.1023/A:1014628810714
- DuBois, D.L., Portillo, N., Rhodes, J.E., Silverthorn, N., Valentine, J.C., 2011. How effective are mentoring programs for youth? A systematic assessment of the evidence. Psychol Sci Public Interest 12, 57–91. https://doi.org/10.1177/1529100611414806
- Ehrich, L.C., Hansford, B., Tennent, L., 2004. Formal mentoring programs in education and other professions: A review of the literature. Educational Administration Quarterly 40, 518–540. https://doi.org/10.1177/0013161X04267118
- Ensher, E.A., Heun, C., Blanchard, A., 2003. Online mentoring and computer-mediated communication: New directions in research. Journal of Vocational Behavior, Special Issue on Technology and Careers 63, 264–288. https://doi.org/10.1016/S0001-8791(03)00044-7
- Erjongmanee, S., 2017. Assessment of how Thai generation-Z students gain understanding in engineering courses: Case study, in: 2017 IEEE Global Engineering Education
 Conference (EDUCON). Presented at the 2017 IEEE Global Engineering Education
 Conference (EDUCON), pp. 1237–1242.
 https://doi.org/10.1109/EDUCON.2017.7943006
- Fallatah, H., Park, Y.S., Farsi, J., Tekian, A., 2018. Mentoring clinical-year medical students: Factors contributing to effective mentoring. Journal of Medical Education and Curricular Development. https://doi.org/doi:10.1177/2382120518757717.
- Fitton, D., Bell, B., 2014. Working with teenagers within HCI research: Understanding teencomputer interaction, in: Proceedings of the 28th International BCS Human Computer Interaction Conference on HCI 2014 – Sand, Sea and Sky – Holiday HCI, BCS-HCI '14. BCS, UK, pp. 201–206. https://doi.org/10.14236/ewic/hci2014.23
- Fitton, D., Read, J.C.C., Horton, M., 2013. The challenge of working with teens as participants in interaction design, in: CHI '13 Extended Abstracts on Human Factors in Computing Systems, CHI EA '13. ACM, New York, NY, USA, pp. 205–210. https://doi.org/10.1145/2468356.2468394

- Fong, N.S., Mansor, W.F.A.W., Zakaria, M.H., Sharif, N.H.M., Nordin, N.A., 2012. The roles of mentors in a Collaborative Virtual Learning Environment (CVLE) Project.
 Procedia Social and Behavioral Sciences, The 8th International Language for Specific Purposes (LSP) Seminar Aligning Theoretical Knowledge with Professional Practice 66, 302–311. https://doi.org/10.1016/j.sbspro.2012.11.272
- Garner, R., 2016. Summer Institute in Science, Technology, Engineering and Research SISTER Program. NASA. URL http://www.nasa.gov/content/summer-institute-inscience-technology-engineering-and-research-sister-program (accessed 7.4.17).
- Garrick, B., Pendergast, D., Geelan, D., 2017. Through the lens of generational theory, in: Theorising Personalised Education. Springer, Singapore, pp. 47–60. https://doi.org/10.1007/978-981-10-2700-0_4
- Gelinas, L., Pierce, R., Winkler, S., Cohen, I.G., Lynch, H.F., Bierer, B.E., 2017. Using social media as a research recruitment tool: Ethical issues and recommendations. The American Journal of Bioethics 17, 3–14. https://doi.org/10.1080/15265161.2016.1276644
- General Authority for Statistics, 2018. ICT Access and Usage by Households and Individuals Survey. General Authority for Statistics. URL https://www.stats.gov.sa/en/952 (accessed 10.8.19).
- General Authority for Statistics, 2016. Labor Force Survey. General Authority for Statistics. URL https://www.stats.gov.sa/en/873 (accessed 9.11.16).
- General Authority for Statistics, 2015. Labor Force Survey. General Authority for Statistics. URL https://www.stats.gov.sa/en/403 (accessed 9.11.16).
- Ghawji, M., Sajid, M.R., Shaikh, A.A., Sheriff, R., Cahusac, P., Alkattan, K., 2017.
 Perspectives of students and mentors on a formal mentorship program in Saudi
 Arabia. Int J Med Educ 8, 25–27. https://doi.org/10.5116/ijme.587a.06f8
- Glasemann, M., Kanstrup, A.M., 2011. Emotions on diabetes: A design case of user mockups by young people living with diabetes. CoDesign 7, 123–130. https://doi.org/10.1080/15710882.2011.609894
- Glasemann, M., Kanstrup, A.M., Ryberg, T., 2010. Making chocolate-covered broccoli:
 Designing a mobile learning game about food for young people with diabetes, in:
 Proceedings of the 8th ACM Conference on Designing Interactive Systems, DIS '10.
 ACM, New York, NY, USA, pp. 262–271. https://doi.org/10.1145/1858171.1858219

- Gray, D., Brown, S., Macanufo, J., 2010. Gamestorming: A playbook for innovators, rulebreakers, and changemakers. O'Reilly Media, Inc.
- Greenhow, C., Askari, E., 2017. Learning and teaching with social network sites: A decade of research in K–12 related education. Educ Inf Technol 22, 623–645. https://doi.org/10.1007/s10639-015-9446-9
- Gruen, D., Rauch, T., Redpath, S., Ruettinger, S., 2002. The use of stories in user experience design. International Journal of Human–Computer Interaction 14, 503–534. https://doi.org/10.1080/10447318.2002.9669132
- Guillory, J., Wiant, K.F., Farrelly, M., Fiacco, L., Alam, I., Hoffman, L., Crankshaw, E.,
 Delahanty, J., Alexander, T.N., 2018. Recruiting hard-to-reach populations for survey research: Using Facebook and Instagram advertisements and in-person intercept in LGBT bars and nightclubs to recruit LGBT young adults. Journal of Medical Internet Research 20, e197. https://doi.org/10.2196/jmir.9461
- Gunawardena, C.N., Jayatilleke, B.G., Fernando, S., Kulasekere, C., Lamontagne, M.D.,
 Ekanayake, M.B., Thaiyamuthu, T., 2012. Developing online tutors and mentors in Sri
 Lanka through a community building model: Predictors of satisfaction, in: 2012
 International Conference on Advances in ICT for Emerging Regions (ICTer).
 Presented at the 2012 International Conference on Advances in ICT for Emerging
 Regions (ICTer), pp. 145–155. https://doi.org/10.1109/ICTer.2012.6421413
- Guta, H., Karolak, M., 2015. Veiling and blogging: Social media as sites of identity negotiation and expression among Saudi women. ProQuest. Journal of International Women's Studies 16.
- Haggard, D.L., Dougherty, T.W., Turban, D.B., Wilbanks, J.E., 2011. Who is a mentor? A review of evolving definitions and implications for research. Journal of Management 37, 280–304. https://doi.org/10.1177/0149206310386227
- Hanington, B.M., 2010. Relevant and rigorous: Human-centered research and design education. Design Issues 26, 18–26. https://doi.org/10.1162/DESI_a_00026
- Headlam-Wells, J., 2004. E-mentoring for aspiring women managers. Women in Management Review 19, 212–218. https://doi.org/10.1108/09649420410541281
- Headlam-Wells, J., Gosland, J., Craig, J., 2006. Beyond the organisation: The design and management of E-mentoring systems. International Journal of Information Management 26, 372–385. https://doi.org/10.1016/j.ijinfomgt.2006.04.001

- Hicks, A., 2012. Analyzing La Cuna : New approaches for mentoring in professional associations. Collaborative Librarianship 2–13.
- Hixenbaugh, P., Dewart, H., Drees, D., Williams, D., 2006. Peer e-mentoring: Enhancement of the first year experience. Psychology Learning & Teaching 5, 8–14. https://doi.org/10.2304/plat.2005.5.1.8
- Igel, C., Urquhart, V., 2012. Generation Z, meet cooperative learning. Middle School Journal 43, 16–21. https://doi.org/10.1080/00940771.2012.11461816
- Ivanova, A., Ivanova, G., 2009. Net-generation learning style: A challenge for higher education, in: Proceedings of the International Conference on Computer Systems and Technologies and Workshop for PhD Students in Computing, CompSysTech '09. ACM, New York, NY, USA, pp. 72:1–72:6. https://doi.org/10.1145/1731740.1731818
- Jacobi, M., 1991. Mentoring and undergraduate academic success: A literature review. Review of Educational Research 61, 505–532. https://doi.org/10.3102/00346543061004505
- Janasz, S.C. de, Godshalk, V.M., 2013. The role of e-mentoring in protégés' learning and satisfaction. Group & Organization Management 38, 743–774. https://doi.org/10.1177/1059601113511296
- Kalisch, B.J., Falzetta, L., Cooke, J., 2005. Group e-mentoring: A new approach to recruitment into nursing. Nurs Outlook 53, 199–205. https://doi.org/10.1016/j.outlook.2004.12.005
- Khan, R., Gogos, A., 2013. Online mentoring for biotechnology graduate students: An industry–academia partnership. Journal of Asynchronous Learning Networks 17, 89– 107.
- Khare, R., Sahai, E., Pramanick, I., 2013. Remote mentoring young females in STEM through MAGIC. arXiv:1304.7547.
- Klasen, N., Clutterbuck, D., 2012. Implementing mentoring schemes. Routledge.
- Kramsch, C., Widdowson, H.G., 1998. Language and culture. OUP Oxford.
- Kuppuswamy, P., Rekha, S.N., 2015. The effect of social networks media on the quality of changes in the people life style: A study report on Kingdom of Saudi Arabia and other Middle East. IOSR Journal of Engineering 05.

- Kyriakoullis, L., Zaphiris, P., 2016. Culture and HCI: A review of recent cultural studies in HCI and social networks. Univ Access Inf Soc 15, 629–642. https://doi.org/10.1007/s10209-015-0445-9
- Lamb, P., Aldous, D., 2014. The role of e-mentoring in distinguishing pedagogic experiences of gifted and talented pupils in physical education. Physical Education and Sport Pedagogy 19, 301–319. https://doi.org/10.1080/17408989.2012.761682
- Lambton-Howard, D., Anderson, R., Montague, K., Garbett, A., Hazeldine, S., Alvarez, C., Sweeney, J.A., Olivier, P., Kharrufa, A., Nappey, T., 2019. WhatFutures: Designing large-scale engagements on WhatsApp, in: Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems, CHI '19. Association for Computing Machinery, Glasgow, Scotland Uk, pp. 1–14. https://doi.org/10.1145/3290605.3300389
- Lambton-Howard, D., Olivier, P., Vlachokyriakos, V., Celina, H., Kharrufa, A., 2020.
 Unplatformed Design: A Model for Appropriating Social Media Technologies for Coordinated Participation, in: Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems, CHI '20. Association for Computing Machinery, Honolulu, HI, USA, pp. 1–13. https://doi.org/10.1145/3313831.3376179
- Lasater, K., Young, P.K., Mitchell, C.G., Delahoyde, T.M., Nick, J.M., Siktberg, L., 2014. Connecting in distance mentoring: Communication practices that work. Nurse Education Today 34, 501–506. https://doi.org/10.1016/j.nedt.2013.07.009
- Lazar, J., Feng, J.H., Hochheiser, H., 2010. Research methods in human-computer interaction. Wiley Publishing.
- Leck, J.D., Elliott, C., Rockwell, B., 2012. E-mentoring women: Lessons learned from a pilot program. Journal of Diversity Management (JDM) 7, 83–96. https://doi.org/10.19030/jdm.v7i2.7476
- Lee, H., Noh, S., 2003. Educational use of e-mentoring to encourage women into science and engineering, in: Web and Communication Technologies and Internet-Related Social Issues HSI 2003. Presented at the International Conference Human Society@
 Internet, Springer Berlin Heidelberg, pp. 75–84. https://doi.org/10.1007/3-540-45036-X_8
- Lenear, P.E., 2007. E-mentoring interaction models., in: International Research Conference in the Americas of the Academy of Human Resource Development.

- Letourneau, N., Stewart, M., Masuda, J.R., Anderson, S., Cicutto, L., McGhan, S., Watt, S., 2012. Impact of online support for youth with asthma and allergies: Pilot study. Journal of Pediatric Nursing 27, 65–73. https://doi.org/10.1016/j.pedn.2010.07.007
- Li, Q., Moorman, L., Dyjur, P., 2010. Inquiry-based learning and e-mentoring via videoconference: A study of mathematics and science learning of Canadian rural students. Education Tech Research Dev 58, 729–753. https://doi.org/10.1007/s11423-010-9156-3
- Ligadu, C., Anthony, P., 2015. E-mentoring 'MentorTokou': Support for mentors and mentees during the practicum. Procedia – Social and Behavioral Sciences, The Proceedings of 5th World Conference on Learning, Teaching and Educational Leadership 186, 410–415. https://doi.org/10.1016/j.sbspro.2015.04.144
- Macafee, D., 2012. E-mentoring in surgical training. Health Services Insights 9. https://doi.org/10.4137/HSI.S9613
- Mason, L.D., Duin, A.H., Lammers, E., 1994. Linking learners: Structuring a mentoring via telecommunications course. Computers and Composition 11, 123–135. https://doi.org/10.1016/S8755-4615(06)80005-8
- Mazzone, E., Read, J.C., Beale, R., 2008. Design with and for disaffected teenagers, in:
 Proceedings of the 5th Nordic Conference on Human–Computer Interaction: Building Bridges, NordiCHI '08. ACM, New York, NY, USA, pp. 290–297.
 https://doi.org/10.1145/1463160.1463192
- McDaniel, S., Yarbrough, A.-M., 2016. A literature review of afterschool mentoring programs for children at risk. JARI 1.
- MentorNet Program Account, 2016. How MentorNet works. URL https://www.youtube.com/watch?v=9UPntkr3pYY (accessed 24.3.20).
- Ministry of Communications and Information Technology, 2016. Over 18 million users of social media programs and applications in Saudi Arabia. URL https://www.mcit.gov.sa/en/media-center/news/89698 (accessed 10.8.19).
- Ministry of Education, n.d. Higher Education Statistics. البوابة الإلكترونية لوزارة التعليم. URL https://www.moe.gov.sa/en/HigherEducation/governmenthighereducation/Pages/HES. aspx (accessed 9.11.16).
- Mizukami, M. da G.N., Reali, A.M. de M.R., Tancredi, R.M.S., 2015. Construction of professional knowledge of teaching: Collaboration between experienced primary

school teachers and university teachers through an online mentoring programme. Journal of Education for Teaching 41, 493–513. https://doi.org/10.1080/02607476.2015.1108626

- Mollica, M., Mitchell, A., 2013. Increasing retention and student satisfaction utilizing an online peer mentoring program: Preliminary results. Procedia – Social and Behavioral Sciences, 4th International Conference on New Horizons in Education 106, 1455– 1461. https://doi.org/10.1016/j.sbspro.2013.12.163
- Muller, C.B., Barsion, S.J., 2003. Assessment of a large-scale e-mentoring network for women in engineering and science: Just how good is MentorNet? Women in Engineering ProActive Network.
- Nor, A.S.M., Kasim, N.A.A., 2015. Distance mentoring and usage of technology: What can be learned from the success story? Procedia Economics and Finance, International Accounting and Business Conference 2015, IABC 2015 31, 303–312. https://doi.org/10.1016/S2212-5671(15)01198-3
- Norodien-Fataar, N., 2012. An account of student mentors' "modes of reflexivity" during an e-mentoring programme at a university. South African Journal of Higher Education 26, 546–563.
- O'Neill, D.K., 2004. Building social capital in a knowledge-building community: Telementoring as a catalyst. Interactive Learning Environments 12, 179–208. https://doi.org/10.1080/10494820512331383419
- O'Neill, D.K., Asgari, M., Dong, Y.R., 2011. Trade-offs between perceptions of success and planned outcomes in an online mentoring program. Mentoring & Tutoring: Partnership in Learning 19, 45–63.
- Oulasvirta, A., Kurvinen, E., Kankainen, T., 2003. Understanding contexts by being there: Case studies in bodystorming. Pers Ubiquit Comput 7, 125–134. https://doi.org/10.1007/s00779-003-0238-7
- Oxford Learner's Dictionaries , n.d. . Oxford Learner's Dictionaries | Find definitions, translations, and grammar explanations at Oxford Learner's Dictionaries. URL https://www.oxfordlearnersdictionaries.com/ (accessed 1.12.20).
- Ozkan, M., Solmaz, B., 2015. The changing face of the employees Generation Z and their perceptions of work (a study applied to university students). Procedia Economics and

Finance, 4th World Conference on Business, Economics and Management (WCBEM-2015) 26, 476–483. https://doi.org/10.1016/S2212-5671(15)00876-X

- Packard, B.W.-L., Ellison, K.L., Sequenzia, M.R., 2004. Show and tell: Photo-interviews with urban adolescent girls. International Journal of Education & the Arts 5, 1–20.
- Petridou, E., 2009. E-mentoring women entrepreneurs: Discussing participants' reactions. Gender in Mgmt: Int J 24, 523–542. https://doi.org/10.1108/17542410910991809
- Piliouras, T., Yu, P.L., Tian, X., Yu, S., Sultana, N., Lauer, J., Berry, L., 2013. Anytime, anywhere, anyone instruction: A tool for life-long learning, in: 2013 IEEE Integrated STEM Education Conference (ISEC). Presented at the 2013 IEEE Integrated STEM Education Conference (ISEC), pp. 1–5. https://doi.org/10.1109/ISECon.2013.6525225
- Plummer, C.A., Nyang'au, T.O., 2009. Reciprocal e-mentoring: Accessible international exchanges. International Social Work 52, 811–822. https://doi.org/10.1177/0020872809342652
- Pookulangara, S., Koesler, K., 2011. Cultural influence on consumers' usage of social networks and its impact on online purchase intentions. Journal of Retailing and Consumer Services 18, 348–354. https://doi.org/10.1016/j.jretconser.2011.03.003
- Poole, E.S., Peyton, T., 2013. Interaction design research with adolescents: Methodological challenges and best practices, in: Proceedings of the 12th International Conference on Interaction Design and Children, IDC '13. ACM, New York, NY, USA, pp. 211–217. https://doi.org/10.1145/2485760.2485766
- Priporas, C.-V., Stylos, N., Fotiadis, A.K., 2017. Generation Z consumers' expectations of interactions in smart retailing: A future agenda. Computers in Human Behavior. https://doi.org/10.1016/j.chb.2017.01.058
- Quintana, M.G.B., Zambrano, E.P., 2014. E-mentoring: The effects on pedagogical training of rural teachers with complex geographical accesses. Computers in Human Behavior 30, 629–636. https://doi.org/10.1016/j.chb.2013.07.042
- Radcliffe, D., Bruni, P., 2019. State of social media, Middle East: 2018 (SSRN Scholarly Paper No. ID 3334515). Social Science Research Network, Rochester, NY.
- Ramirez, F.O., Kwak, N., 2015. Women's enrollments in STEM in higher education: Crossnational trends, 1970–2010, in: Jr, W.P., Frehill, L.M., McNeely, C.L. (Eds.), Advancing women in science. Springer International Publishing, pp. 9–49. https://doi.org/10.1007/978-3-319-08629-3_2

- Rashid, R.A., Rahman, M.F.A., 2014. Social networking sites for online mentoring and creativity enhancement. International Journal of Technology Enhanced Learning 6, 34–45. https://doi.org/10.1504/IJTEL.2014.060024
- Read, J., Horton, M., 2016. Future directions for quality teenCI Research, in: Perspectives on HCI Research with Teenagers, Human–Computer Interaction Series. Springer, Cham, pp. 269–278. https://doi.org/10.1007/978-3-319-33450-9_11
- Read, J., Horton, M., Iversen, O., Fitton, D., Little, L., 2013. Methods of working with teenagers in interaction design, in: CHI '13 Extended Abstracts on Human Factors in Computing Systems, CHI EA '13. ACM, New York, NY, USA, pp. 3243–3246. https://doi.org/10.1145/2468356.2479657
- Redmond, P., 2015. Discipline specific online mentoring for secondary pre-service teachers. Computers & Education 90, 95–104. https://doi.org/10.1016/j.compedu.2015.08.018
- Reid, J., Smith, E., Iamsuk, N., Miller, J., 2017. Balancing the equation: Mentoring first-year female STEM students at a regional university. International Journal of Innovation in Science and Mathematics Education (formerly CAL-laborate International) 24.
- Reyaee, S., Ahmed, A., 2015. Growth pattern of social media usage in Arab Gulf states: An analytical study. Social Networking 04, 23. https://doi.org/10.4236/sn.2015.42003
- Rhodes, J., Lowe, S.R., 2008. Youth mentoring and resilience: Implications for practice. Child Care in Practice 14, 9–17. https://doi.org/10.1080/13575270701733666
- Rhodes, J., Spencer, R., Saito, R.N., Sipe, C.L., 2006. Online mentoring: The promise and challenges of an emerging approach to youth development. J Primary Prevent 27, 497–513. https://doi.org/10.1007/s10935-006-0051-y
- Rickard, K., 2004. E-mentoring and pedagogy: A useful nexus for evaluating online mentoring programs for small business? Mentoring & Tutoring: Partnership in Learning 12, 383–401. https://doi.org/10.1080/030910042000275972
- Risquez, A., 2008. E-mentoring: An extended practice, an emerging discipline.
- Risquez, A., Sanchez-Garcia, M., 2012. The jury is still out: Psychoemotional support in peer e-mentoring for transition to university. The Internet and Higher Education, Emotions in online learning environments 15, 213–221. https://doi.org/10.1016/j.iheduc.2011.11.003

- Roberts, A., 2000. Mentoring revisited: A phenomenological reading of the literature. Mentoring & Tutoring: Partnership in Learning 8, 145–170. https://doi.org/10.1080/713685524
- Rowland, K.N., 2012. E-mentoring: An innovative twist to traditional mentoring. J.Technol. Manag. Innov. 7, 228–237. https://doi.org/10.4067/S0718-27242012000100015
- Salber, D., Coutaz, J., 1993. Applying the Wizard of Oz technique to the study of multimodal systems, in: Bass, L.J., Gornostaev, J., Unger, C. (Eds.), Human-Computer Interaction, Lecture Notes in Computer Science. Springer Berlin Heidelberg, pp. 219– 230.
- Salem, F., 2015. The Arab social media report 2017: Social media and the Internet of Things: Towards data-driven policymaking in the Arab world. MBR School of Government.
- Salimi, G., Mohammadi, M., Hosseini, N., 2017. Presenting a framework for effective ementoring relationships in medical sciences: A meta-synthesis method. Interdisciplinary Journal of Virtual Learning in Medical Sciences 8.
- Sanchez, B., Harris, J., 1996. Online mentoring A success story. Learning & Leading with Technology 23, 57–60.
- Sanders, E.B.-N., Stappers, P.J., 2008. Co-creation and the new landscapes of design. CoDesign 4, 5–18. https://doi.org/10.1080/15710880701875068
- Sattar, K., Abdulghani, H.M., Ahmad, T., John, J., Dabeeb, D.A., Meo, S.A., 2017. Principled physicians are not born – they are mentored: Medical students' perception of the values and need of mentors at the College of Medicine, Saudi Arabia. Journal of Pakistan Medical Association 67, 7.
- Savoy, J.N., 2013. Women in Technology Sharing Online (WitsOn): Assessing usage, satisfaction, and outcomes from an e-mentoring course, in: Engage Everyone:
 Building an Inclusive Climate for Diverse Communities of Women in STEM.
 Presented at the Engage Everyone: Building an Inclusive Climate for Diverse Communities of Women in STEM.
- Schimke, D., 2007. Design, implementation and evaluation of an online community to foster girls' interest and participation in STEM, in: GROUP '07 Doctoral Consortium Papers, GROUP '07. ACM, New York, NY, USA, pp. 12:1–12:2. https://doi.org/10.1145/1329112.1329124

- Schleicher, D., Jones, P., Kachur, O., 2010. Bodystorming as embodied designing. Interactions 17, 47–51. https://doi.org/10.1145/1865245.1865256
- Scott, J.R., 2013. Einstein girls: Exploring STEM careers, interest, and identity in an online mentoring community. University of Florida, Ann Arbor.
- Sears, A., Jacko, J.A., Jacko, J.A., Jacko, J.A., Jacko, J.A., 2007. The human–computer interaction handbook : Fundamentals, evolving technologies and emerging applications (second edition). CRC Press. https://doi.org/10.1201/9781410615862
- Shatto, B., Erwin, K., 2017. Teaching Millennials and Generation Z: Bridging the generational divide. Creative Nursing; Minneapolis 23, 24–28. http://dx.doi.org/10.1891/1078-4535.23.1.24
- Shukri, N., Leil, A.A., 2017. Female mentees' perspectives of a mentoring program in the Saudi context. The International Journal of Social Sciences and Humanities Invention 4, 3888–3899. https://doi.org/10.18535/ijsshi/v4i9.01
- Sinclair, C., 2003. Mentoring online about mentoring: Possibilities and practice. Mentoring & Tutoring: Partnership in Learning 11, 79–94. https://doi.org/10.1080/1361126032000054826
- Single, P.B., Muller, C.B., 2005. Electronic mentoring programs: A model to guide practice and research.
- Single, P.B., Muller, C.B., 2001. When email and mentoring unite: The implementation of a nationwide electronic mentoring program – MentorNet, the National Electronic Industrial Mentoring Network for Women in Engineering and Science.
- Single, P.B., Single, R.M., 2005. E-mentoring for social equity: Review of research to inform program development. Mentoring & Tutoring: Partnership in Learning 13, 301–320. https://doi.org/10.1080/13611260500107481
- Smith-Jentsch, K.A., Scielzo, S.A., Yarbrough, C.S., Rosopa, P.J., 2008. A comparison of face-to-face and electronic peer-mentoring: Interactions with mentor gender. Journal of Vocational Behavior, Mentoring 72, 193–206. https://doi.org/10.1016/j.jvb.2007.11.004
- Stanger, N., Alnaghaimshi, N., Pearson, E., 2017. How do Saudi youth engage with social media? First Monday 22. http://dx.doi.org/10.5210/fm.v22i5.7102

- Stebbins, R., 2001. Exploratory research in the social sciences. SAGE Publications, 2455 Teller Road, Thousand Oaks California 91320 United States of America. https://doi.org/10.4135/9781412984249
- Stewart, S., Carpenter, C., 2009. Electronic mentoring: An innovative approach to providing clinical support. International Journal of Therapy and Rehabilitation 16, 199.
- Stoeger, H., Duan, X., Schirner, S., Greindl, T., Ziegler, A., 2013. The effectiveness of a oneyear online mentoring program for girls in STEM. Computers & Education 69, 408– 418. https://doi.org/10.1016/j.compedu.2013.07.032
- Sulphey, M.M., Allam, Z., 2017. Efficacy of mentoring in enhancing academic outcome of Business Students in KSA. The Social Sciences 12, 1384–1388.
- Thongmak, M., 2013. Social network system in classroom: Antecedents of Edmodo © adoption. Journal of e-Learning and Higher Education 1–15. https://doi.org/10.5171/2013.657749
- Todd, R. I., Moon, N. w., Langston, C., 2016. E-mentoring and its relevance for competencybased education for students with disabilities: Research from the GSAA BreakThru model. Competency-Based Education 1, 17–30. https://doi.org/10.1002/cbe2.1009
- Vacca, R., 2017. Bicultural: Examining teenage latinas' perspectives on technologies for emotional support, in: Proceedings of the 2017 Conference on Interaction Design and Children, IDC '17. ACM, New York, NY, USA, pp. 117–126. https://doi.org/10.1145/3078072.3079742
- Vines, J., Clarke, R., Wright, P., McCarthy, J., Olivier, P., 2013. Configuring participation: On how we involve people in design, in: Proceedings of the SIGCHI Conference on Human Factors in Computing Systems, CHI '13. ACM, New York, NY, USA, pp. 429–438. https://doi.org/10.1145/2470654.2470716
- Vredenburg, K., Mao, J.-Y., Smith, P.W., Carey, T., 2002. A survey of user-centered design practice, in: Proceedings of the SIGCHI Conference on Human Factors in Computing Systems, CHI '02. Association for Computing Machinery, New York, NY, USA, pp. 471–478. https://doi.org/10.1145/503376.503460
- Wallis, J.A.M., Riddell, J.K., Smith, C., Silvertown, J., Pepler, D.J., 2015. Investigating patterns of participation and conversation content in an online mentoring program for northern Canadian Youth. Mentoring & Tutoring: Partnership in Learning 23, 228– 247. https://doi.org/10.1080/13611267.2015.1072395

- Ware, P., Ramos, J., 2013. First-generation college students: Mentoring through social media. Intl Jnl of Ment & Coach in Ed 2, 149–162. https://doi.org/10.1108/IJMCE-02-2013-0009
- Wienhofen, L.W.M., Lindkvist, C., Noebels, M., 2014. User-centered design for smart solarpowered micro-grid communities, in: 2014 14th International Conference on Innovations for Community Services (I4CS). Presented at the 2014 14th International Conference on Innovations for Community Services (I4CS), pp. 39–46. https://doi.org/10.1109/I4CS.2014.6860551
- Wilburn, N.L., Amer, T.S., Kilpatrick, B.G., 2009. Establishing an eMentor program: Increasing the interaction between accounting majors and professionals, in: Advances in Accounting Education, Advances in Accounting Education. Emerald Group Publishing Limited, pp. 27–59.
- Williams, S., Sunderman, J., Kim, J., 2012. E-mentoring in an online course: Benefits and challenges to e-mentors 10, 15.
- Winder, B.P.C., 2014. The hashtag generation: The Twitter phenomenon in Saudi society. Journal of Georgetown University-Qatar Middle Eastern Studies Student Association 6. https://doi.org/10.5339/messa.2014.6
- Wolfe, G., Gregg, N., 2015. E-mentoring supports for improving the persistence of underrepresented students in on-line and traditional courses, in: Antona, M., Stephanidis, C. (Eds.), Universal access in human-computer interaction. Access to Learning, Health and Well-Being, Lecture Notes in Computer Science. Springer International Publishing, pp. 243–251. https://doi.org/10.1007/978-3-319-20684-4_24
- Woodley, C., Stephen Burgess, Rafael Paguio, Scott Bingley, 2015. Technology mentors: enablers of ICT uptake in Australian small business. Education + Training 57, 658– 672. https://doi.org/10.1108/ET-08-2014-0095
- Wu, Y., Zheng, F., Cai, S., Xiang, N., Zhong, Z., He, J., Xu, F., 2012. Effective collaborative learning in biomedical education using a web-based infrastructure, in: 2012 Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC). Presented at the 2012 Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC), pp. 5070–5073. https://doi.org/10.1109/EMBC.2012.6347133
- Yalof, B., Chametzky, B., 2016. Mentoring online doctoral students through a community of practice model. The Online Journal of Distance Education and e-Learning 4, 39.

- Yamin, M., Aljehani, S.A., 2016. E-learning and women in Saudi Arabia: An empirical study. BVICA M's International Journal of Information Technology 8, 950.
- Yang, S.C., Liu, S.F., 2002. A qualitative study of mentors' scaffolding in a teacher professional development online workshop, in: Proceedings of the International Conference on Computers in Education, ICCE '02. IEEE Computer Society, Washington, DC, USA, pp. 1342–.
- Yarosh, S., Radu, I., Hunter, S., Rosenbaum, E., 2011. Examining values: An analysis of nine years of IDC research, in: Proceedings of the 10th International Conference on Interaction Design and Children, IDC '11. ACM, New York, NY, USA, pp. 136–144. https://doi.org/10.1145/1999030.1999046
- Zey, M.G., 1984. The mentor connection: Strategic alliances in corporate life. Transaction Publishers.

Appendix A Pilot Study

A.1 Mentors Profiles



Group E	·
Interactive media design	
Princess Norah University	12
About me:	
Hi, my name is *****. I work as Lecturer in Princess Norah University. in Graphic design and	
digital media department. During our journey together we will explore and create magical	
interactive project in fun and easy way.	
Group F	
Engineering / Architecture	
Dar Al Hekma University	KINA
About me:	
A young architect who appreciates innovative experiments and aims to fill the gap between the	
academic field and professional practice.	
Group G	
Game Development	
Semanoor & ALFAC	
About me:	
I design games, characters, battles and write the code to see them come to life.	
Group H	
Computer Science – Space Technology	
KACST - National Satellite Technology Center	
About me:	
I am working in Imaging Payload for satellites. I started with CAD Designer - systems engineer	
then as PM for the project. Working in engineering areas is really interesting and exciting, the	
way of solving the problem, the way you can see your designs in real life, the way you learn	
each day something new are all joyful and interesting. Knowing that you are working in a	
unique field, in a space related tasks, and facing challenges that are not common in the world	
are really challenge and way to prove that ladies are always capable to work and contribute	
significantly.	

A.2 Pre-Study Survey for the Mentees

This survey is intended to collect data about you and your daughter as part of your participation in the e-mentoring program.

Please compete the following about you and your daughter who is participating in the study. The first section to be completed by the you, the second section is to be completed the mentee (your daughter) with your help.

First section: This section to be completed by the parent/guardian/carer.

Mentee's Name:

Date of birth:

Grade at school:

- 1) The child lives with her:
 - a) Mother
 - b) Father
 - c) Both parents
 - d) Other (Please state relationship to child)
- 2) How old is the father?
 - a. 25-30
 - b. 30-45
 - c. 45-60
 - d. 61 or older

3) How old is the mother?

- a. 25-30
- b. 30-45
- c. 45-60
- d. 61 or older

4) If the child is living with someone other than her parents, how old is he/she:

- a. 25-30
- b. 30-45
- c. 45-60
- d. 61 or older
- 5) Which province do you live in:
 - a. Central province
 - b. Eastern province
 - c. Western Province
 - d. Northern Province

e. Southern Province

City:

- 6) What is the highest level of education the father has completed?
 - a. High school degree
 - b. Bachelor's degree
 - c. Master's degree
 - d. Doctoral degree
 - e. Professional degree (MD)
 - f. Other
- 7) What is the highest level of education the mother has completed?
 - a. High school degree
 - b. Bachelor's degree
 - c. Master's degree
 - d. Doctoral degree
 - e. Professional degree (MD)
 - f. Other
- 8) If the child is living with someone other than her parents, what is the highest level of education he/she has completed?
 - a. High school degree
 - b. Bachelor's degree
 - c. Master's degree
 - d. Doctoral degree
 - e. Professional degree (MD)
 - f. Other
- 9) What is the major of the father?
 - a. Science
 - b. Engineering
 - c. Technology
 - d. Medicine
 - e. Business
 - f. Other, specify
- 10) What is the occupation of the father? Where does he work?
- 11) What is the major of the mother?
 - a. Science
 - b. Engineering
 - c. Technology
 - d. Medicine
 - e. Business
 - f. Other, specify

- 12) What is the occupation of the mother? Where does she work?
- 13) If the child is living with someone other than her parents, what is the major of him/her?
 - a. Science
 - b. Engineering
 - c. Technology
 - d. Medicine
 - e. Business
 - f. Other, specify
- 14) If the child is living with someone other than her parents, what is his/her occupation? Where does he/she work?
- 15) Has anyone in the family been in a mentoring program? Who?
- 16) How did you know about this program?
 - i. School
 - ii. Event
 - iii. Friends
 - iv. Social network, specify
 - v. Other, specify

Second section:

This section is to be completed the mentee (your daughter) with your help.

- Do you know any adult who works in science, engineering or technology careers? Yes | No | Not sure
- 2) Do you know any woman who works in science, engineering or technology careers? Yes | No | Not sure
- 3) Are you excited to be in the mentoring program? Yes | No | Not sure
- 4) Do you think that you will benefit from the mentoring program? Yes | No | Not sure
- 5) Did anyone influence you into joining the e-mentoring program? Yes | No | Not sure

If yes, who?

- 6) Why did you join the e-mentoring program?
- 7) Have you ever been in a mentoring/e-mentoring Program? Yes | No | Not sure
- 8) Has anyone in the family been in a mentoring program? Who?
- 9) How did you know about this program?
 - i. School
 - ii. Event
 - iii. Friends
 - iv. Social network, specify
 - v. Other, specify

A.3 Post-Study Survey for the Mentees

Dear Mentee,

This survey is intended to collect data about your experience with the e-mentoring program. It is composed of two parts. The first part contains open ended questions please feel free to elaborate your answers. The second part has statements that we would like to know your view about them regarding the e-mentoring experience.

Part 1: Please answer the following questions regarding your experience with the e-mentoring program. Feel free to elaborate:

- Would you describe your e-mentoring experience as a positive one?
- Did you find the 8-week program an appropriate duration?
- If no, would you prefer the program to be shorter or longer? What duration would you see as ideal?
- How would you describe your relationship with your mentor? Please elaborate
- Please describe an example of a recommendation/idea/advice that your mentor suggested, and you benefited from it.
- What do you think is the best thing about e-mentoring?
- What do you think is the biggest problem with e-mentoring?
- Which means of communication would you like to use on the e-mentoring program, select all that apply:
 - a) Audio calls
 - b) Video calls
 - c) Video snaps and clips
 - d) Synchronous communication such as chat
 - e) Asynchronous communication such as discussion boards and forums.
 - f) Other,
- Describe the benefits of the program to you
- What would you suggest to improve the program?

Part 2: Indicate your view of the following aspects of your e-mentoring experience. (For each of the items please indicate your agreement or disagreement with the following statements. Please tick \sqrt{y} your response.)

		Strongly agree	agree	Neither Agree nor Disagree	disagree	Strongly disagree
1)	My mentor is always interested in what I want to talk about.					
2)	My mentor thinks of fun and interesting things to talk about.					
3)	I feel bored when I participate in the e-mentoring program.					
4)	I enjoyed participating in the e-mentoring program.					
5)	I benefited from participating in the e-mentoring program					
6)	After the e-mentoring program, my awareness of STEM careers has increased.					
7)	After the e-mentoring program, my confidence in myself has increased.					
8)	I spent most of time on my group reading posts					
9)	I spent most of time on my group writing posts					
10)	The e-mentoring program consumed a lot of my time.					
11)	I was confident enough to write and express my opinions to my mentor.					
12)	Being in a group with mentees is distracting.					
13)	Being in a group with mentees showed me their different perspectives.					
14)	In the future, I will participate in a mentoring program.					
15)	Using online communications make it hard for me to understand my mentor.					
16)	I want to meet my mentor physically.					
17)	I am satisfied with the advice, assistance and support provided by my mentor					
18)	I am satisfied with the frequency of contact with my Mentor					
						L

A.4 Semi-structured Interview Questions for the Mentors

- 1. Did you ever have a mentor, and do you have any mentors in your life right now?
- 2. If yes, what impact did being mentored affect your life?
- 3. Why do you want to be a mentor?
- 4. What expectations did you have about the e-mentorship program?
- 5. Did you have any doubts or concerns about being a mentor?
- 6. Would you describe your e-mentoring experience as a positive one?
- 7. In your opinion what are the strengths of the program?
- 8. What aspects of the program would you like to improve?
- 9. How would you describe your relationship with your mentees? Please elaborate
- 10. Did you feel overwhelmed or burdened by participating in the e-mentoring program?
- 11. Please describe an example of a recommendation/idea/advice that you suggested to your mentee and you feel that she benefited from it.
- 12. Would you volunteer to serve as a mentor again in the future? Why or why not?
- 13. Which means of communication would you like to use on the e-mentoring program?
- 14. What would you suggest to improve the program?

A.5 Semi-structured Interview Questions for the Mentees

- 1. Did you ever have a mentor, and do you have any mentors in your life right now?
- 2. Would you describe your e-mentoring experience as a positive one?
- 3. In your opinion what are the strengths of the program?
- 4. What expectations did you have about the e-mentoring system?
- 5. Why did you join the program?
- 6. Why did you choose your mentor?
- 7. What barriers or problems did you face during your participation?
- 8. Did you share your experience or what you have learned with others?
- 9. Would you join an e-mentoring program again?
- 10. Which means of communication would you like to use on the e-mentoring program?
- 11. What would you suggest to improve the program?

A.6 Semi-structured Interview Questions for the Parents

- 1. What is your highest level of education? Your major?
- 2. Have you ever been mentored? by whom?
- 3. What do you think mentoring is?
- 4. If your daughter asked you to join an e-mentoring program that uses video communications, would you agree? Why or why not?
- 5. What are your concerns regarding e-mentoring?
- 6. How do you think these concerns would be eliminated?
- 7. What do you think the benefit of mentoring is to your daughter?

Appendix B Co-design Workshops

B.1 Survey

Dear Participant:

This survey is intended to collect data about you as part of your participation in the design session of an e-mentoring platform.

- 1) How old are you?
- 2) How old is your father?
 - a. 30-45
 - b. 45-60
 - c. 61 or older
- 3) How old is your mother?
 - a. 30-45
 - b. 45-60
 - c. 61 or older
- 4) What is the highest level of education your father has completed?
 - a. Illiterate/ not educated
 - b. High school degree
 - c. Bachelor's degree
 - d. Master's degree
 - e. Doctoral degree
 - f. Professional degree (MD)
 - g. Other
- 5) What is the highest level of education your mother has completed?
 - a. Illiterate/ not educated
 - b. High school degree
 - c. Bachelor's degree
 - d. Master's degree
 - e. Doctoral degree
 - f. Professional degree (MD)
 - g. Other
- 6) What is the major of your father?
 - a. Science
 - b. Engineering
 - c. Technology
 - d. Medicine
 - e. Business
 - f. Other, specify
- 7) What is the occupation of your father? Where does he work?

- 8) What is the major of your mother?
 - a. Science
 - b. Engineering
 - c. Technology
 - d. Medicine
 - e. Business
 - f. Other, specify
- 9) What is the occupation of your mother? Where does she work?

10) Number of family members (father, mother and siblings)

- 11) Did all you siblings who are older than 17 have been enrolled in higher education?
 - a. Yes
 - b. No

12) Do you have a female role model in your life?

- a. Yes
- b. No
- c. I don't know
- 13) If you answered Q.13 with yes, do you communicate with her?
 - a. Yes
 - b. No
- 14) Do you feel sometimes the need to talk to someone regarding your study or future career?
 - a. Yes
 - b. No
 - c. Maybe
- 15) Do you have a clear vision and knowledge about career options available for your field of studying?

Appendix C Qudwa

C.1 System Architecture

The system of Qudwa consists of three main components: front-end, backend, and database. The front-end which is a static website is built using Jekyll (https://jekyllrb.com/) and hosted on GitHub pages. Jekyll was chosen because it reduces the complexity and need for deploying a dynamic content management system (CMS) by generating a static website from markdown files with GitHub repositories. The front end connects to the application programming interface (API) using JavaScript to send form submissions and analytics data. The backend is built with NodeJS as an Express server (https://expressjs.com/) and hosted on Open Lab servers. The backend helps in two parts. For the first part, it exposes a number of API endpoints that can be used to collect analytics data and form submissions from the website, storing them in Qudwa's database and displaying them on the admin panel. The second part offers an interface for the admin to edit the content of *Oudwa* without directly interacting with the GitHub repository. It achieves that by utilizing GitHub API to create a new commit to *Oudwa*'s repository every time the content is edited, which triggers GitHub to rebuild the front-end and serve the updated Qudwa website. The third component of Qudwa is a MySQL database, which is hosted on Open Lab servers. It is used to store analytics data and form submissions.

C.2 Post-Study Survey

Dear Mentee,

This survey is intended to collect data about your experience with Qudwa. It is composed of three parts: the nature of the relationship, communication channels and the impact of *Qudwa* on you.

Nature of Relationship

- 1. Have you ever talked to a professional to help you find answers regarding your studies or career?
 - Yes
 - No If yes, please explain
- 2. How would you rate your overall experience with Qudwa?
 - Very good
 - Good
 - Somewhat good

- Neutral
- Somewhat bad
- Bad
- Very bad

Please explain

- 3. How do you rate your experience with the mentor in terms of her ability to help you?.
 - Very good
 - Good
 - Somewhat good
 - Neutral
 - Somewhat bad
 - Bad
 - Very bad

Please explain

- 4. I felt the need for more guidance and help than what *Qudwa* provided
 - Strongly agree
 - Agree
 - Somewhat agree
 - Neutral
 - Somewhat disagree
 - Disagree
 - Strongly disagree

Please explain

- 5. Why did you participate in Qudwa?
 - Finding answers
 - The need to talk to someone with experience
 - increasing my knowledge
 - Curiosity/new experience
 - Others, please specify
- 6. The available information about each mentor was enough for you to decide which mentor to choose.
 - Strongly agree
 - Agree
 - Somewhat agree
 - Neutral
 - Somewhat disagree
 - Disagree
 - Strongly disagree

Please explain

7. Did you communicate with more than one mentor? Why?

Please explain

- 8. How do you feel about the short-term relationship with your mentor?
 - Strongly satisfied
 - Satisfied
 - Somewhat satisfied
 - Neutral
 - Somewhat dissatisfied
 - Dissatisfied
 - Strongly dissatisfied

Please explain

- 9. Your experience with Qudwa could be replaced by a Q&A website, and why?
 - Strongly agree
 - Agree
 - Somewhat agree
 - Neutral
 - Somewhat disagree
 - Disagree
 - Strongly disagree

Please explain

Communication channels

- 1. How did you know about *Qudwa*?
 - Twitter
 - Instagram
 - WhatsApp
 - Other, specify
- 2. What other channels would be helpful for promoting and reaching more girls?
 - Twitter
 - Instagram
 - WhatsApp
 - Other, specify
- 3. Why did you choose your communication channel?
 - Ease of use
 - I use it a lot
 - Confident using it
 - Privacy
 - Ease of expressing myself,

- Other, specify
- 4. Did you face any problems with the channel you chose?
 - Yes
 - No

If yes, please explain

- 5. What other channels do you prefer to use other than the one used in *Qudwa*?
- 6. Did you have any reservations regarding your participation with Qudwa?
 - Privacy
 - Connecting with strangers
 - The nature of communication channels
 - No reservations
 - Others, please specify
- 7. Did you visit the mentors' social media accounts provided on the website?
 - Yes
 - No

If yes, please explain

Impact of Qudwa

- 1. How comfortable were you asking the mentor questions?
 - Strongly uncomfortable
 - Uncomfortable
 - Somewhat uncomfortable
 - Neutral
 - Somewhat comfortable
 - Comfortable
 - Strongly comfortable
- 2. Did you recommend Qudwa to others?
 - Yes
 - No
- 3. Did you share your experience with *Qudwa* with others?
 - Yes
 - No
- 4. Did you browse other mentors' profiles to increase your knowledge about STEM jobs?
 - Yes
 - No

- 5. How do you rate the effect of displaying all the mentors from different fields on you to raise your awareness about study and career options?
 - Strongly effective
 - Effective
 - Somewhat effective
 - Neutral
 - Somewhat ineffective
 - Ineffective
 - Strongly ineffective
- 6. How do you feel after your participation in Qudwa?
 - Strongly determined
 - Determined
 - Somewhat determined
 - Neutral
 - Somewhat confused
 - Confused
 - Strongly confused
- 7. How helpful was the presence of an intermediate person (Aseel) for guiding and communicating in comparison to an automated system?
 - Strongly helpful
 - Helpful
 - Somewhat helpful
 - Neutral
 - Somewhat useless
 - Useless
 - Strongly useless
- 8. If you had the chance to make changes on *Qudwa* what would you change or add?

C.3 Semi-Structured Interview Questions for the mentors and the mentees

- 1. Did you ever have a mentor, and do you have any mentors right now?
- 2. If yes, what impact did being mentored have on your life?
- 3. How did you get matched with your mentor?
- 4. What expectations do you have about *Qudwa*?
- 5. Would you consider your experience with Qudwa as successful? Why or Why not?
- 6. What barriers did you face during your participation?
- 7. In your opinion what were the elements in the Qudwa that were most helpful?
- 8. Did you have any reservations about the channels of communications?
- 9. Which channel of communication did you prefer? And why ?
- 10. How did *Qudwa* fit into your daily activities?

- 11. How do you feel about your participation in Qudwa?
- 12. What do you think about my role in *Qudwa*?